

Attachment I of RNTPC Paper No. 6/21

圖 例 NOTATION			
ZONES		地帶	
COMMERCIAL	с	商業	
COMPREHENSIVE DEVELOPMENT AREA	CDA	綜合 發展 區	
RESIDENTIAL (GROUP B)	R(B)	住宅(乙類)	
RESIDENTIAL (GROUP C)	R(C)	住宅(丙類)	
RESIDENTIAL (GROUP D)	R(D)	住宅(丁類)	
RESIDENTIAL (GROUP E)	R(E)	住宅(戊類)	
VILLAGE TYPE DEVELOPMENT	v	鄕 村 式 潑 展	
GOVERNMENT, INSTITUTION OR COMMUNITY	G/IC	政府、機構或社區	
OPEN SPACE	0	休憩用地	
OTHER SPECIFIED USES	OU	其他指定用途	
GREEN BELT	GB	緣 化 地 帶	
CONSERVATION AREA	CA	自然保育區	
COMMUNICATIONS		交通	
RAILWAY AND STATION (ELEVATED)	#34 STATION	鐵路及車站(高架)	
LIGHT RAIL	<u>→→→→</u> L R →→→→	輕鎡	
MAJOR ROAD AND JUNCTION		主要道路及路口	
ELEVATED ROAD		高架道路	
MISCELLANEOUS		其他	
BOUNDARY OF PLANNING SCHEME	<u> </u>	規 劃 範 園 界 線	
PETROL FILLING STATION	PFS	加油站	

土地用途及面積一覽表 SCHEDULE OF USES AND AREAS

	大約面積及百分率 APPROXIMATE AREA & %		田法
0323	公頃 HECTARES	% 百分率	用逐
COMMERCIAL	0.69	0.15	商業
COMPREHENSIVE DEVELOPMENT AREA	6.99	1.47	綜合發展區
RESIDENTIAL (GROUP B)	9.54	2.01	住宅(乙類)
RESIDENTIAL (GROUP C)	2.52	0.53	住宅(丙類)
RESIDENTIAL (GROUP D)	20.36	4.29	住宅(丁類)
RESIDENTIAL (GROUP E)	11.56	2.43	住宅(戊類)
VILLAGE TYPE DEVELOPMENT	102.78	21.64	鄉村式發展
GOVERNMENT, INSTITUTION OR COMMUNITY	5.78	1.22	政府、機構或社區
OPEN SPACE	1.93	0.41	休憩用地
OTHER SPECIFIED USES	29.66	6.25	其他指定用途
GREEN BELT	110.68	23.30	綠 化 地 帶
CONSERVATION AREA	152.43	32.10	自然保育區
NULLAH	0.99	0.21	明渠
MAJOR ROAD ETC.	19.02	3.99	主要道路等
TOTAL PLANNING SCHEME AREA	474.93	100.00	規劃範圍總面積

夾附的《註釋》屬這份圖則的一部分 THE ATTACHED NOTES ALSO FORM PART OF THIS PLAN

> 乱割 署 遵 照 城 市 規 割 委 員 會 指 示 擬



圖則編號 PLAN No.

S/TM-LTYY/10



圖例		
NOTATION		
ZONES		地 帶
COMMERCIAL	С	商業
COMPREHENSIVE DEVELOPMENT AREA	CDA	綜合發展區
RESIDENTIAL (GROUP A)	R(A)	住宅(甲類)
RESIDENTIAL (GROUP B)	R(B)	住宅(乙類)
RESIDENTIAL (GROUP C)	R(C)	住宅(丙類)
RESIDENTIAL (GROUP D)	R(D)	住宅(丁類)
RESIDENTIAL (GROUP E)	R(E)	住宅(戊類)
VILLAGE TYPE DEVELOPMENT	V	鄉村式發展
GOVERNMENT, INSTITUTION OR COMMUNITY	G/IC	政 府 、 機 構 或 社 區
OPEN SPACE	0	休憩用地
OTHER SPECIFIED USES	OU	其他指定用途
GREEN BELT	GB	綠化地帶
CONSERVATION AREA	CA	自然保育區
COMMUNICATIONS		交通
RAILWAY AND STATION (ELEVATED)	STATION	鐵路及車站(高架)
LIGHT RAIL	-++L R-+++	輕鐵
MAJOR ROAD AND JUNCTION	l L	主要道路及路口
ELEVATED ROAD		高架道路
MISCELLANEOUS		其他
BOUNDARY OF PLANNING SCHEME		規 劃 範 圍 界 線
PETROL FILLING STATION	PFS	加油站
	ŝ.	

土地用途及面積一覽表 SCHEDULE OF USES AND AREAS

	大約面積 APPROXIMA	及百分率 TE AREA & %	
0525	公頃 HECTARES	% 百分率	田述
COMMERCIAL	0.69	0.15	商 業
COMPREHENSIVE DEVELOPMENT AREA	6.99	1.47	() () () () () () () () () () () () () (
RESIDENTIAL (GROUP A)	21.52	4.53	住宅(甲類)
RESIDENTIAL (GROUP B)	9.54	2.01	住宅(乙類)
RESIDENTIAL (GROUP C)	2.52	0.53	住宅(丙類)
RESIDENTIAL (GROUP D)	20.36	4.29	住宅(丁類)
RESIDENTIAL (GROUP E)	2.00	0.42	住宅(戊類)
VILLAGE TYPE DEVELOPMENT	102.78	21.64	鄉村式發展
GOVERNMENT, INSTITUTION OR COMMUNITY	6.31	1.33	政 府 、 機 構 或 社 區
OPEN SPACE	1.93	0.41	休憩用地
OTHER SPECIFIED USES	29.66	6.25	其他指定用途
GREEN BELT	98.19	20.67	緣 化 地 帶
CONSERVATION AREA	152.43	32.10	自然保育區
NULLAH	0.99	0.21	明渠
MAJOR ROAD ETC.	19.02	3.99	主要道路等
TOTAL PLANNING SCHEME AREA	474.93	100.00	規 劃 範 圍 總 面 積

夾附的《註釋》屬這份圖則的一部分[,] 現經修訂並按照城市規劃條例第5條展示。 THE ATTACHED NOTES ALSO FORM PART OF THIS PLAN AND HAVE BEEN AMENDED FOR EXHIBITION UNDER SECTION 5 OF THE TOWN PLANNING ORDINANCE

核准圖編號 S/T M-L T Y Y/1 0 的修訂 AMENDMENTS TO APPROVED PLAN No. S/TM-LTYY/10

AMENDMENTS EXHIBITED UNDER SECTION 5 OF THE TOWN PLANNING ORDINANCE

按照城市規劃條例第 5 條 展示的修訂

AMENDMENT ITEM A AMENDMENT ITEM B



修訂項目A項 修訂項目B項

(參看附表) (SEE ATTACHED SCHEDULE)

PREPARED BY THE PLANNING DEPARTMENT UNDER THE DIRECTION OF THE TOWN PLANNING BOARD

圖則編號 PLAN No.

S/TM-LTYY/10A

APPROVED DRAFT LAM TEI AND YICK YUEN OUTLINE ZONING PLAN NO. S/TM-LTYY/10A

(Being an Approved *a Draft* Plan for the Purposes of the Town Planning Ordinance)

NOTES

(N.B. These form part of the Plan)

- (1) These Notes show the uses or developments on land falling within the boundaries of the Plan which are always permitted and which may be permitted by the Town Planning Board, with or without conditions, on application. Where permission from the Town Planning Board for a use or development is required, the application for such permission should be made in a prescribed form. The application shall be addressed to the Secretary of the Town Planning Board, from whom the prescribed application form may be obtained.
- (2) Any use or development which is always permitted or may be permitted in accordance with these Notes must also conform to any other relevant legislation, the conditions of the Government lease concerned, and any other Government requirements, as may be applicable.
- (3) No action is required to make the use of any land or building which was in existence immediately before the first publication in the Gazette of the notice of the draft development permission area plan conform to this Plan, provided such use has continued since it came into existence. Any material change of such use or any other development (except minor alteration and/or modification to the development of the land or building in respect of such use which is always permitted) must be always permitted in terms of the Plan or in accordance with a permission granted by the Town Planning Board.
- (4) A use or development of any land or building permitted under an earlier draft or approved plan including development permission area plan for the area and effected or undertaken during the effective period of that plan is always permitted under this Plan. Any material change of such use or any other development (except minor alteration and/or modification to the completed development of the land or building which is always permitted) must be always permitted in terms of the Plan or in accordance with a permission granted by the Town Planning Board.
- (5) Except to the extent that paragraph (3) or (4) applies, any use or development falling within the boundaries of the Plan and also within the boundaries of the draft development permission area plan, unless always permitted in terms of the Plan, shall not be undertaken or continued on or after the date of the first publication in the Gazette of the notice of the draft development permission area plan without permission from the Town Planning Board.
- (6) Except as otherwise specified by the Town Planning Board, when a use or material change of use is effected or a development or redevelopment is undertaken, as always permitted in terms of the Plan or in accordance with a permission granted by the Town Planning Board, all permissions granted by the Town Planning Board in respect of the site of the use or material change of use or development or redevelopment shall lapse.

- (7) Road junctions, alignment of roads and railway tracks, and boundaries between zones may be subject to minor adjustments as detailed planning proceeds.
- (8) The following uses or developments are always permitted on land falling within the boundaries of the Plan except (a) where the uses or developments are specified in Column 2 of the Notes of individual zones or (b) as provided in paragraph (9) in relation to areas zoned "Conservation Area":
 - (a) maintenance, repair or demolition of a building;
 - (b) provision, maintenance or repair of plant nursery, amenity planting, open space, rain shelter, refreshment kiosk, footpath, bus/public light bus/light rail stop or lay-by, cycle track, taxi rank, public utility pipeline, electricity mast, lamp pole, telephone booth, telecommunications radio base station, automatic teller machine and shrine;
 - (c) maintenance or repair of road, railway track, watercourse, nullah, sewer and drain;
 - (d) geotechnical works, local public works, road works, sewerage works, drainage works, environmental improvement works, marine related facilities and waterworks (excluding works on service reservoir) and such other public works co-ordinated or implemented by Government;
 - (e) rebuilding of New Territories Exempted House;
 - (f) replacement of an existing domestic building, i.e. a domestic building which was in existence on the date of the first publication in the Gazette of the notice of the draft development permission area plan, by a New Territories Exempted House; and
 - (g) provision, maintenance or repair of a grave of an indigenous New Territories villager or a locally based fisherman and his family members for which permission has been obtained from Government.
- (9) In areas zoned "Conservation Area",
 - (a) the following uses or developments are always permitted:
 - (i) maintenance or repair of plant nursery, amenity planting, sitting out area, rain shelter, refreshment kiosk, road, watercourse, nullah, public utility pipeline, electricity mast, lamp pole, telephone booth, shrine and grave;
 - geotechnical works, local public works, road works, sewerage works, drainage works, environmental improvement works, marine related facilities, waterworks (excluding works on service reservoir) and such other public works co-ordinated or implemented by Government; and
 - (iii) provision of amenity planting by Government; and
 - (b) the following uses or developments require permission from the Town Planning Board:

provision of plant nursery, amenity planting (other than by Government), sitting out area, rain shelter, refreshment kiosk, footpath, public utility pipeline, electricity mast, lamp pole, telephone booth and shrine.

(10) In any area shown as 'Road', all uses or developments except those specified in paragraphs (8)(a) to (8)(d) and (8)(g) above and those specified below require permission from the Town Planning Board:

road, toll plaza, on-street vehicle park, railway station and railway track.

(11) (a) Except in areas zoned "Conservation Area", temporary use or development of any land or building not exceeding a period of two months is always permitted provided that no site formation (filling or excavation) is carried out and that the use or development is a use or development specified below :

structures for carnivals, fairs, film shooting on locations, festival celebrations, religious functions or sports events.

- (b) Except as otherwise provided in paragraph (11)(a), and subject to temporary uses for open storage and port back-up purposes which are prohibited in areas zoned "Conservation Area", temporary use or development of any land or building not exceeding a period of three years requires permission from the Town Planning Board. Notwithstanding that the use or development is not provided for in terms of the Plan, the Town Planning Board may grant permission, with or without conditions, for a maximum period of three years, or refuse to grant permission.
- (c) Temporary use or development of land or building exceeding three years requires permission from the Town Planning Board in accordance with the terms of the Plan.
- (12) Unless otherwise specified, all building, engineering and other operations incidental to and all uses directly related and ancillary to the permitted uses and developments within the same zone are always permitted and no separate permission is required.
- (13) In these Notes, unless the context otherwise requires or unless as expressly provided below, terms used in the Notes shall have the meanings as assigned under section 1A of the Town Planning Ordinance.

"Existing building" means a building, including a structure, which is physically existing and is in compliance with any relevant legislation and the conditions of the Government lease concerned.

"New Territories Exempted House" means a domestic building other than a guesthouse or a hotel; or a building primarily used for habitation, other than a guesthouse or a hotel, the ground floor of which may be used as 'Shop and Services' or 'Eating Place', the building works in respect of which are exempted by a certificate of exemption under Part III of the Buildings Ordinance (Application to the New Territories) Ordinance (Cap. 121).

APPROVED DRAFT LAM TEI AND YICK YUEN OUTLINE ZONING PLAN NO. <u>S/TM-LTYY/10A</u>

SCHEDULE OF USES

	Page
COMMERCIAL	1
COMPREHENSIVE DEVELOPMENT AREA	3
RESIDENTIAL (GROUP A)	6
RESIDENTIAL (GROUP B)	6 9
RESIDENTIAL (GROUP C)	8 11
RESIDENTIAL (GROUP D)	10 13
RESIDENTIAL (GROUP E)	12 15
VILLAGE TYPE DEVELOPMENT	14 17
GOVERNMENT, INSTITUTION OR COMMUNITY	16 19
OPEN SPACE	18 21
OTHER SPECIFIED USES	19 22
GREEN BELT	20 23
CONSERVATION AREA	22 25

Column 1	Column 2
Uses always permitted	Uses that may be permitted with or
	without conditions on application
	to the Town Planning Board
Ambulance Depot	Broadcasting, Television and/or Film Studio
Commercial Bathhouse/	Flat
Massage Establishment	Government Refuse Collection Point
Eating Place	Hospital
Educational Institution	House (other than rebuilding of New
Exhibition or Convention Hall	Territories Exempted House or
Government Use (not elsewhere specified)	replacement of existing domestic
Hotel	building by New Territories Exempted
Information Technology and	House permitted under the covering Notes)
Telecommunications Industries	Petrol Filling Station
Institutional Use (not elsewhere specified)	Residential Institution
Library	
Market	
Off-course Betting Centre	
Office	
Place of Entertainment	
Place of Recreation, Sports or Culture	
Private Club	
Public Clinic	
Public Convenience	
Public Transport Terminus or Station	
Public Utility Installation	
Public Vehicle Park	
(excluding container vehicle)	
Recyclable Collection Centre	
Religious Institution	
Rural Committee/Village Office	
School	
Shop and Services	
Social Welfare Facility	
Training Centre	
Utility Installation for Private Project	
Wholesale Trade	

COMMERCIAL

Planning Intention

This zone is intended primarily for commercial developments, which may include shop, services, place of entertainment and eating place, functioning mainly as local shopping centre(s) serving the immediate neighbourhood.

COMMERCIAL (Cont'd)

<u>Remarks</u>

- (a) No new development, or addition, alteration and/or modification to or redevelopment of an existing building shall result in a total development and/or redevelopment in excess of a maximum plot ratio of 3.6 and a maximum building height of 12 storeys including car park (36m), or the plot ratio and height of the building which was in existence on the date of the first publication in the Gazette of the notice of the draft development permission area plan, whichever is the greater.
- (b) In determining the maximum plot ratio for the purposes of paragraph (a) above, any floor space that is constructed or intended for use solely as car park, loading/unloading bay, plant room and caretaker's office, provided such uses and facilities are ancillary and directly related to the development or redevelopment, may be disregarded.
- (c) Based on the individual merits of a development or redevelopment proposal, minor relaxation of the plot ratio and building height restrictions stated in paragraph (a) above may be considered by the Town Planning Board on application under section 16 of the Town Planning Ordinance.

Column 1	Column 2
Uses always permitted	Uses that may be permitted with or
juli i juli juli juli juli juli juli jul	without conditions on application
	to the Town Planning Board
	U
	Eating Place
	Educational Institution
	Flat
	Government Refuse Collection Point
	Government Use (not elsewhere specified)
	House (other than rebuilding of New
	Territories Exempted House or
	replacement of existing domestic
	building by New Territories
	Exempted House permitted
	under the covering Notes)
	Institutional Use (not elsewhere specified)
	Library
	Market
	Off-course Betting Centre
	Office
	Place of Entertainment
	Place of Recreation, Sports or Culture
	Private Club
	Public Clinic
	Public Convenience
	Public Transport Terminus or Station
	Public Utility Installation
	Public Venicle Park
	(excluding container venicle)
	Recyclable Collection Centre Deligious Institution
	Religious Institution
	Residential Institution
	School
	Schop and Services
	Shop and Scivics Social Walfare Eacility
	Utility Installation for Drivate Droject
	Othery instantion for Flivate Flojeet

COMPREHENSIVE DEVELOPMENT AREA

Planning Intention

This zone is intended for comprehensive development/redevelopment of the area for residential *and/or commercial* uses with the provision of commercial, open space and other supporting facilities, if any. The zoning is to facilitate appropriate planning control over the development mix, scale, design and layout of development, taking account of various environmental, traffic, infrastructure and other constraints.

COMPREHENSIVE DEVELOPMENT AREA (Cont'd)

<u>Remarks</u>

- (a) Pursuant to section 4A(2) of the Town Planning Ordinance, and except as otherwise expressly provided that it is not required by the Town Planning Board, an applicant for permission for development on land designated "Comprehensive Development Area" shall prepare a Master Layout Plan for the approval of the Town Planning Board and include therein the following information:
 - (i) the area of the proposed land uses, the nature, position, dimensions, and heights of all buildings to be erected in the area;
 - (ii) the proposed total site area and gross floor area for various uses, total number of flats and flat size, where applicable;
 - (iii) the details and extent of Government, institution or community (GIC) and recreational facilities, public transport and parking facilities, and open space to be provided within the area;
 - (iv) the alignment, widths and levels of any roads proposed to be constructed within the area;
 - (v) the landscape and urban design proposals within the area;
 - (vi) programmes of development in detail;
 - (vii) an environmental assessment report to examine any possible environmental problems that may be caused to or by the proposed development during and after construction and the proposed mitigation measures to tackle them;
 - (viii) a drainage and sewerage impact assessment report to examine any possible drainage and sewerage problems that may be caused by the proposed development and the proposed mitigation measures to tackle them;
 - (ix) a traffic impact assessment report to examine any possible traffic problems that may be caused by the proposed development and the proposed mitigation measures to tackle them; and
 - (x) such other information as may be required by the Town Planning Board.
- (b) The Master Layout Plan should be supported by an explanatory statement which contains an adequate explanation of the development proposal, including such information as land tenure, relevant lease conditions, existing conditions of the site, the character of the site in relation to the surrounding areas, principles of layout design, major development parameters, design population, types of GIC facilities, and recreational and open space facilities.

COMPREHENSIVE DEVELOPMENT AREA (Cont'd)

Remarks (Cont'd)

- (c) On land designated "Comprehensive Development Area" to the west of Fuk Hang Tsuen Road near Lam Tei Tsuen, no new development, or addition, alteration and/or modification to or redevelopment of an existing building shall result in a total development and/or redevelopment in excess of a maximum domestic plot ratio of 2.0, a maximum non-domestic plot ratio of 0.11 and a maximum building height of 15 storeys excluding car park (45m).
- (d) On land designated "Comprehensive Development Area" to the north of Yuen Long Highway near Fuk Hang Tsuen, no new development, or addition, alteration and/or modification to or redevelopment of an existing building shall result in a total development and/or redevelopment in excess of a maximum plot ratio of 0.54 and a maximum building height of 6 storeys over single-storey car park.
- (e) In determining the maximum domestic and non-domestic plot ratios for the purposes of paragraph (c) above and the maximum plot ratio for the purpose of paragraph (d) above, any floor space that is constructed or intended for use solely as car park, loading/unloading bay, plant room and caretaker's office, or caretaker's quarters and recreational facilities for the use and benefit of all the owners or occupiers of the domestic building or domestic part of the building, provided such uses and facilities are ancillary and directly related to the development or redevelopment, may be disregarded. Any floor space that is constructed or intended for use solely as Government, institution or community facilities, as required by the Government, may also be disregarded.
- (f) Based on the individual merits of a development or redevelopment proposal, minor relaxation of the plot ratio and building height restrictions stated in paragraphs (c) and (d) above may be considered by the Town Planning Board on application under section 16 of the Town Planning Ordinance.

RESIDENTIAL (GROUP A)

Column 1	Column 2		
Uses always permitted	Uses that may be permitted with or		
	without conditions on application		
	to the Town Planning Board		
Ambulance Depot	Commercial Bathhouse/		
Flat	Massage Establishment		
Government Use (not elsewhere specified)	Eating Place		
House	Educational Institution		
Library	Exhibition or Convention Hall		
Market	Government Refuse Collection Point		
Place of Recreation, Sports or Culture	Hospital		
Public Clinic	Hotel		
Public Transport Terminus or Station (excluding open-air terminus or station)	Institutional Use (not elsewhere specified) Office		
Public Vehicle Park	Petrol Filling Station		
(excluding container vehicle)	Place of Entertainment		
Religious Institution (Ancestral Hall only)	Private Club		
Residential Institution	Public Convenience		

(Please see next page)

Public Transport Terminus or Station

Religious Institution (not elsewhere specified)

Shop and Services (not elsewhere specified)

(not elsewhere specified)

School (not elsewhere specified)

Public Utility Installation

Training Centre

School (in free-standing purpose-designed

Utility Installation for Private Project

building only)

Social Welfare Facility

<u>RESIDENTIAL (GROUP A)</u> (Cont'd)

In addition, the following uses are always permitted (a) on the lowest three floors of a building, taken to include basements; or (b) in the purpose-designed non-residential portion of an existing building, both excluding floors containing wholly or mainly car parking, loading/unloading bays and/or plant room:

Eating Place Educational Institution Institutional Use (not elsewhere specified) Off-course Betting Centre Office Place of Entertainment Private Club Public Convenience Recyclable Collection Centre School Shop and Services Training Centre

Planning Intention

This zone is intended primarily for high-density residential developments. Commercial uses are always permitted on the lowest three floors of a building or in the purpose-designed non-residential portion of an existing building.

<u>RESIDENTIAL (GROUP A)</u> (Cont'd)

Remarks

- (a) No new development, or addition, alteration and/or modification to or redevelopment of an existing building shall result in a total development and/or redevelopment in excess of a maximum plot ratio of 6.5 and a maximum building height of 160mPD, or the plot ratio and height of the existing building, whichever is the greater.
- (b) In determining the maximum plot ratio for the purposes of paragraph (a) above, any floor space that is constructed or intended for use solely as public vehicle parks, public transport facilities and Government, institution or community facilities, as required by the Government, may be disregarded.
- (c) In determining the maximum plot ratio for the purposes of paragraph (a) above, the area of any part of the site that is occupied or intended to be occupied by free-standing purpose-designed buildings (including both developed on ground and on podium level) solely for accommodating Government, institution or community facilities including school(s), as may be required by the Government, shall be deducted from calculation of the site area.
- (d) In determining the maximum plot ratio for the purposes of paragraph (a) above, any floor space that is constructed or intended for use solely as car park, loading/unloading bay, plant room and caretaker's office, or caretaker's quarters and recreational facilities for the use and benefit of all the owners or occupiers of the domestic building or domestic part of the building, provided such uses and facilities are ancillary and directly related to the development or redevelopment, may be disregarded.
- (e) Based on the individual merits of a development or redevelopment proposal, minor relaxation of the plot ratio and building height restrictions stated in paragraph (a) above may be considered by the Town Planning Board on application under section 16 of the Town Planning Ordinance.

Column 1	Column 2
Uses always permitted	Uses that may be permitted with or
	without conditions on application
	to the Town Planning Board
Flat	Ambulance Depot
Government Use (Police Reporting Centre,	Eating Place
Post Office only)	Educational Institution
House	Government Refuse Collection Point
Library	Government Use (not elsewhere specified)
Residential Institution	Hospital
School (in free-standing purpose-designed	Hotel
building only)	Institutional Use (not elsewhere specified)
Utility Installation for Private Project	Market
	Office
	Petrol Filling Station
	Place of Entertainment
	Place of Recreation, Sports or Culture
	Private Club
	Public Clinic
	Public Convenience
	Public Transport Terminus or Station
	Public Utility Installation
	Public Vehicle Park
	(excluding container vehicle)
	Recyclable Collection Centre
	Religious Institution
	Rural Committee/Village Office
	School (not elsewhere specified)
	Shop and Services
	Social Welfare Facility
	Training Centre
	-

RESIDENTIAL (GROUP B)

Planning Intention

This zone is intended primarily for sub-urban medium-density residential developments in rural areas where commercial uses serving the residential neighbourhood may be permitted on application to the Town Planning Board.

RESIDENTIAL (GROUP B) (Cont'd)

Remarks

(a) No new development, or addition, alteration and/or modification to or redevelopment of an existing building shall result in a total development and/or redevelopment in excess of the maximum plot ratio, site coverage and building height specified below, or the plot ratio, site coverage and height of the building which was in existence on the date of the first publication in the Gazette of the notice of the draft development permission area plan, whichever is the greater.

Maximum Building Height

Sub-area	Maximum <u>Plot Ratio</u>	Maximum <u>Site Coverage</u>	No. of Storeys	Building Height
R(B)1	1.0	40%	4 storeys over single-storey car park	15m
R(B)2	1.26	40%	6 storeys over single-storey car park	21m
R(B)3	2.1	40%	12 storeys excluding car park	36m (excluding car park)

- (b) In determining the maximum plot ratio and site coverage for the purposes of paragraph (a) above, any floor space that is constructed or intended for use solely as car park, loading/unloading bay, plant room and caretaker's office, or caretaker's quarters and recreational facilities for the use and benefit of all the owners or occupiers of the domestic building or domestic part of the building, provided such uses and facilities are ancillary and directly related to the development or redevelopment, may be disregarded.
- (c) Based on the individual merits of a development or redevelopment proposal, minor relaxation of the plot ratio, site coverage and building height restrictions stated in paragraph (a) above may be considered by the Town Planning Board on application under section 16 of the Town Planning Ordinance.

Column 1	Column 2
Uses always permitted	Uses that may be permitted with or
	without conditions on application
	to the Town Planning Board
Elot	Ambulance Denot
Fiat Covernment Use (Police Penerting Centre	Esting Disco
Bost Office only)	Educational Institution
House	Covernment Defuse Collection Doint
House	Government Lise (not cleavybers specified)
Othicy instantion for Private Project	Government Use (not elsewhere specified)
	Hospital
	Hotel
	Librory
	Library Detrol Filling Station
	Place of Bospection Sports on Culture
	Prace of Recreation, Sports of Culture
	Private Club Dyblic Clinic
	Public Clinic Dyblic Convenience
	Public Convenience Dublic Transport Terminus or Station
	Public Hallsport Terminus of Station
	Public Utility Installation Dublic Vehicle Derly
	(avaluding container vehicle)
	Recycluding container vehicle)
	Recyclable Collection Centre Religious Institution
	Religious Institution
	Residential Institution Burel Committee/Willege Office
	Sabaal
	Shop and Services
	Shop and Services
	Training Control
	Training Centre

RESIDENTIAL (GROUP C)

Planning Intention

This zone is intended primarily for low-rise, low-density residential developments where commercial uses serving the residential neighbourhood may be permitted on application to the Town Planning Board.

RESIDENTIAL (GROUP C) (Cont'd)

<u>Remarks</u>

- (a) No new development, or addition, alteration and/or modification to or redevelopment of an existing building shall result in a total development and/or redevelopment in excess of a maximum plot ratio of 0.4 and a maximum building height of 3 storeys (9m) including car park, or the plot ratio, site coverage and height of the building which was in existence on the date of the first publication in the Gazette of the notice of the draft development permission area plan, whichever is the greater.
- (b) In determining the maximum plot ratio for the purposes of paragraph (a) above, any floor space that is constructed or intended for use solely as car park, loading/unloading bay, plant room and caretaker's office, or caretaker's quarters and recreational facilities for the use and benefit of all the owners or occupiers of the domestic building or domestic part of the building, provided such uses and facilities are ancillary and directly related to the development or redevelopment, may be disregarded.
- (c) Based on the individual merits of a development or redevelopment proposal, minor relaxation of the plot ratio and building height restrictions stated in paragraph (a) above may be considered by the Town Planning Board on application under section 16 of the Town Planning Ordinance.

Column 1	Column 2
Uses always permitted	Uses that may be permitted with or without conditions on application to the Town Planning Board
Agricultural Use	Eating Place
Government Use (Police Reporting Centre	Flat
Post Office only)	Government Refuse Collection Point
House (Redevelopment: Addition	Government Use (not elsewhere specified) #
Alteration and/or Modification	House (not elsewhere specified)
to existing house only)	Institutional Use (not elsewhere specified) #
On-Farm Domestic Structure	Library
Rural Committee/Village Office	Market
	Petrol Filling Station
	Place of Recreation, Sports or Culture
	Public Clinic
	Public Convenience
	Public Transport Terminus or Station
	Public Utility Installation #
	Public Vehicle Park
	(excluding container vehicle)
	Recyclable Collection Centre
	Religious Institution #
	Residential Institution #
	School #
	Shop and Services
	Social Welfare Facility #
	Utility Installation for Private Project
	-

RESIDENTIAL (GROUP D)

In addition, the following uses are always permitted on the ground floor of a New Territories Exempted House:

Eating Place Library School Shop and Services

<u>RESIDENTIAL (GROUP D)</u> (Cont'd)

Planning Intention

This zone is intended primarily for improvement and upgrading of existing temporary structures within the rural areas through redevelopment of existing temporary structures into permanent buildings. It is also intended for low-rise, low-density residential developments subject to planning permission from the Town Planning Board.

Remarks

- (a) No addition, alteration and/or modification to or in-situ redevelopment of an existing temporary structure or an existing building (except to 'New Territories Exempted House' or to those annotated with #) shall result in a total development and/or redevelopment in excess of a maximum building area of 37.2m² and a maximum building height of 2 storeys (6m), or the building area and height of the building which was in existence on the date of the first publication in the Gazette of the notice of the draft development permission area plan, whichever is the greater.
- (b) No development including redevelopment for 'Flat' and 'House' uses, other than those to which paragraph (a) above shall apply, shall result in a development and/or redevelopment in excess of a maximum plot ratio of 0.2 and a maximum building height of 2 storeys (6m).
- (c) In determining the maximum plot ratio for the purposes of paragraph (b) above, any floor space that is constructed or intended for use solely as car park, loading/unloading bay, plant room and caretaker's office, or caretaker's quarters and recreational facilities for the use and benefit of all the owners or occupiers of the domestic building or domestic part of the building, provided such uses and facilities are ancillary and directly related to the development or redevelopment, may be disregarded.
- (d) Based on the individual merits of a development or redevelopment proposal, minor relaxation of the plot ratio and building height restrictions stated in paragraph (b) above may be considered by the Town Planning Board on application under section 16 of the Town Planning Ordinance.

RESIDENTIAL (GROUP E)

Column 1	Column 2
Uses always permitted	Uses that may be permitted with or
	without conditions on application
	to the Town Planning Board

Schedule I: for open-air development or for building other than industrial building

Ambulance Depot	Eating Place
Government Use (Police Reporting Centre,	Educational Institution
Post Office only)	Flat
Rural Committee/Village Office	Government Refuse Collection Point
Utility Installation for Private Project	Government Use (not elsewhere specified)
	House (other than rebuilding of New
	Territories Exempted House or
	replacement of existing domestic
	building by New Territories
	Exempted House permitted
	under the covering Notes)
	Institutional Use (not elsewhere specified)
	Library
	Market
	Office
	Petrol Filling Station
	Place of Entertainment
	Place of Recreation, Sports or Culture
	Private Club
	Public Clinic
	Public Convenience
	Public Transport Terminus or Station
	Public Utility Installation
	Public Vehicle Park
	(excluding container vehicle)
	Recyclable Collection Centre
	Religious Institution
	Residential Institution
	School
	Shop and Services
	Social Welfare Facility
	Training Centre

RESIDENTIAL (GROUP E) (Cont'd)

Column 1	Column 2
Uses always permitted	Uses that may be permitted with or
	without conditions on application
	to the Town Planning Board

Schedule II: for existing industrial development

Eating Place (Canteen only)	Office
Government Refuse Collection Point	Petrol Filling Station
Government Use (not elsewhere specified)	Public Convenience
Public Utility Installation	Public Vehicle Park
Recyclable Collection Centre	(excluding container vehicle)
Rural Workshop	Shop and Services (ground floor only)
Utility Installation for Private Project	Vehicle Repair Workshop
Warehouse (excluding Dangerous	Wholesale Trade
Goods Godown)	

Planning Intention

This zone is intended primarily for phasing out of existing industrial uses through redevelopment for residential use on application to the Town Planning Board. Whilst existing industrial uses will be tolerated, new industrial developments are not permitted in order to avoid perpetuation of industrial/residential interface problem.

Remarks

- (a) No new development (except 'New Territories Exempted House') shall exceed a maximum plot ratio of 1.0, a maximum site coverage of 40% and a maximum building height of 4 storeys over single-storey car park (15m).
- (b) No addition, alteration and/or modification to or redevelopment of an existing building (except redevelopment to 'New Territories Exempted House') shall exceed the plot ratio, site coverage and building height restrictions stated in paragraph (a) above, or the plot ratio, site coverage and height of the building which was in existence on the date of the first publication in the Gazette of the notice of the draft development permission area plan, whichever is the greater, subject to redevelopment to the plot ratio in the latter restriction shall be permitted only if the existing building is a domestic building.
- (c) In determining the maximum plot ratio and site coverage for the purposes of paragraphs (a) and (b) above, any floor space that is constructed or intended for use solely as car park, loading/unloading bay, plant room and caretaker's office, or caretaker's quarters and recreational facilities for the use and benefit of all the owners or occupiers of the domestic building or domestic part of the building, provided such uses and facilities are ancillary and directly related to the development or redevelopment, may be disregarded.
- (d) Based on the individual merits of a development or redevelopment proposal, minor relaxation of the plot ratio, site coverage and building height restrictions stated in paragraphs (a) and (b) above may be considered by the Town Planning Board on application under section 16 of the Town Planning Ordinance.

Column 1	Column 2
Uses always permitted	Uses that may be permitted with or
	without conditions on application
	to the Town Planning Board
Agricultural Use	Burial Ground
Government Use (Police Reporting Centre,	Eating Place
Post Office only)	Flat
House (New Territories Exempted	Government Refuse Collection Point
House only)	Government Use (not elsewhere specified) #
On-Farm Domestic Structure	House (not elsewhere specified)
Religious Institution	Institutional Use (not elsewhere specified) #
(Ancestral Hall only)	Market
Rural Committee/Village Office	Petrol Filling Station
	Place of Recreation, Sports or Culture
	Private Club
	Public Clinic
	Public Convenience
	Public Transport Terminus or Station
	Public Utility Installation #
	Public Vehicle Park
	(excluding container vehicle)
	Religious Institution
	(not elsewhere specified) #
	Residential Institution #
	School #
	Shop and Services
	Social Welfare Facility #
	Utility Installation for Private Project

VILLAGE TYPE DEVELOPMENT

In addition, the following uses are always permitted on the ground floor of a New Territories Exempted House:

Eating Place Library School Shop and Services

VILLAGE TYPE DEVELOPMENT (Cont'd)

Planning Intention

The planning intention of this zone is to reflect existing recognized and other villages, and to provide land considered suitable for village expansion and reprovisioning of village houses affected by Government projects. Land within this zone is primarily intended for development of Small Houses by indigenous villagers. It is also intended to concentrate village type development within this zone for a more orderly development pattern, efficient use of land and provision of infrastructures and services. Selected commercial and community uses serving the needs of the villagers and in support of the village development are always permitted on the ground floor of a New Territories Exempted House. Other commercial, community and recreational uses may be permitted on application to the Town Planning Board.

Remarks

- (a) No new development, or addition, alteration and/or modification to or redevelopment of an existing building (except development or redevelopment to those annotated with #) shall result in a total development and/or redevelopment in excess of a maximum building height of 3 storeys (8.23m) or the height of the building which was in existence on the date of the first publication in the Gazette of the notice of the draft development permission area plan, whichever is the greater.
- (b) Based on the individual merits of a development or redevelopment proposal, minor relaxation of the building height restriction stated in paragraph (a) above may be considered by the Town Planning Board on application under section 16 of the Town Planning Ordinance.

Column 1	Column 2	
Uses always permitted	Uses that may be permitted with or	
Uses always permitted	without conditions on application	
	to the Town Planning Board	
Ambulance Depot	Animal Boarding Establishment	
Animal Quarantine Centre	Animal Quarantine Centre	
(in Government building only)	(not elsewhere specified)	
Broadcasting, Television and/or Film Studio	Columbarium	
Cable Car Route and Terminal Building	Correctional Institution	
Eating Place (Canteen,	Crematorium	
Cooked Food Centre only)	Driving School	
Educational Institution	Eating Place (not elsewhere specified)	
Exhibition or Convention Hall	Firing Range	
Field Study/Education/Visitor Centre	Flat	
Government Refuse Collection Point	Funeral Facility	
Government Use (not elsewhere specified)	Helicopter Fuelling Station	
Hospital	Helicopter Landing Pad	
Institutional Use (not elsewhere specified)	Holiday Camp	
Library	Hotel	
Market	House (other than rebuilding of New	
Place of Recreation, Sports or Culture	Territories Exempted House or	
Public Clinic	replacement of existing domestic	
Public Convenience	building by New Territories	
Public Transport Terminus or Station	Exempted House permitted	
Public Utility Installation	under the covering Notes)	
Public Vehicle Park	Off-course Betting Centre	
(excluding container vehicle)	Office	
Recyclable Collection Centre	Petrol Filling Station	
Religious Institution	Place of Entertainment	
Research, Design and Development Centre	Private Club	
Rural Committee/Village Office	Radar, Telecommunications Electronic	
School	Microwave Repeater, Television	
Service Reservoir	and/or Radio Transmitter Installation	
Social Welfare Facility	Refuse Disposal Installation	
Training Centre	(Refuse Transfer Station only)	
Wholesale Trade	Residential Institution	
	Sewage Treatment/Screening Plant	
	Shop and Services (not elsewhere specified)	
	Utility Installation for Private Project	
	Zoo	

GOVERNMENT, INSTITUTION OR COMMUNITY

GOVERNMENT, INSTITUTION OR COMMUNITY (Cont'd)

Planning Intention

This zone is intended primarily for the provision of Government, institution or community facilities serving the needs of the local residents and/or a wider district, region or the territory. It is also intended to provide land for uses directly related to or in support of the work of the Government, organizations providing social services to meet community needs, and other institutional establishments.

Remarks

- (a) On land designated "Government, Institution or Community (1)", no new development, or addition, alteration and/or modification to or redevelopment of an existing building shall result in a total development and/or redevelopment in excess of a maximum gross floor area (GFA) of 35,000m² and a maximum building height of 50 metres above Principal Datum, or the GFA and height of the building which was in existence on the date of the first publication in the Gazette of the notice of the draft development permission area plan, whichever is the greater.
- (b) In determining the maximum GFA for the purposes of paragraph (a) above, any floor space that is constructed or intended for use solely as car park, loading/unloading bay, plant room and caretaker's office, or caretaker's quarters and recreational facilities for the use and benefit of all the owners or occupiers of the domestic building or domestic part of the building, provided such uses and facilities are ancillary and directly related to the development or redevelopment, may be disregarded.
- (c) Based on the individual merits of a development or redevelopment proposal, minor relaxation of the GFA and building height restrictions stated in paragraph (a) above may be considered by the Town Planning Board on application under section 16 of the Town Planning Ordinance.

Column 1	Column 2
Uses always permitted	Uses that may be permitted with or
	without conditions on application
	to the Town Planning Board
Aviary	Eating Place
Barbecue Spot	Government Refuse Collection Point
Field Study/Education/Visitor Centre	Government Use (not elsewhere specified)
Park and Garden	Holiday Camp
Pavilion	Place of Entertainment
Pedestrian Area	Place of Recreation, Sports or Culture
Picnic Area	Private Club
Playground/Playing Field	Public Transport Terminus or Station
Public Convenience	Public Utility Installation
Sitting Out Area	Public Vehicle Park
Zoo	(excluding container vehicle)
	Religious Institution
	Service Reservoir
	Shop and Services
	Tent Camping Ground
	Utility Installation for Private Project

OPEN SPACE

Planning Intention

This zone is intended primarily for the provision of outdoor open-air public space for active and/or passive recreational uses serving the needs of local residents as well as the general public.

OTHER SPECIFIED USES

Column 1 Uses always permitted Column 2 Uses that may be permitted with or without conditions on application to the Town Planning Board

For "Quarry" only

Quarry

Government Use Public Utility Installation

Planning Intention

This zone is intended primarily for quarry use.

For "Refuse Transfer Station" only

Public Utility Installation Refuse Transfer Station Government Use

Planning Intention

This zone is intended primarily for the provision of refuse transfer station.

For "Petrol Filling Station" only

Petrol Filling Station

Government Use Public Utility Installation

Planning Intention

This zone is intended primarily for the provision of petrol filling station.

Animal Boarding Establishment Broadcasting, Television and/or Film Studio Burial Ground Columbarium (within a Religious Institution or extension of existing Columbarium only) Crematorium (within a Religious Institution or extension of existing Crematorium only) Field Study/Education/Visitor Centre Firing Range Flat Golf Course Government Refuse Collection Point Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
 Animal Boarding Establishment Broadcasting, Television and/or Film Studio Burial Ground Columbarium (within a Religious Institution or extension of existing Columbarium only) Crematorium (within a Religious Institution or extension of existing Crematorium only) Field Study/Education/Visitor Centre Firing Range Flat Golf Course Government Refuse Collection Point Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
 Broadcasting, Television and/or Film Studio Burial Ground Columbarium (within a Religious Institution or extension of existing Columbarium only) Crematorium (within a Religious Institution or extension of existing Crematorium only) Field Study/Education/Visitor Centre Firing Range Flat Golf Course Government Refuse Collection Point Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
 Burial Ground Columbarium (within a Religious Institution or extension of existing Columbarium only) Crematorium (within a Religious Institution or extension of existing Crematorium only) Field Study/Education/Visitor Centre Firing Range Flat Golf Course Government Refuse Collection Point Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
 Columbarium (within a Religious Institution or extension of existing Columbarium only) Crematorium (within a Religious Institution or extension of existing Crematorium only) Field Study/Education/Visitor Centre Firing Range Flat Golf Course Government Refuse Collection Point Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
or extension of existing Columbarium only) Crematorium (within a Religious Institution or extension of existing Crematorium only) Field Study/Education/Visitor Centre Firing Range Flat Golf Course Government Refuse Collection Point Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
Crematorium (within a Religious Institution or extension of existing Crematorium only) Field Study/Education/Visitor Centre Firing Range Flat Golf Course Government Refuse Collection Point Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
or extension of existing Crematorium only) Field Study/Education/Visitor Centre Firing Range Flat Golf Course Government Refuse Collection Point Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
Field Study/Education/Visitor Centre Firing Range Flat Golf Course Government Refuse Collection Point Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
Firing Range Flat Golf Course Government Refuse Collection Point Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
Flat Golf Course Government Refuse Collection Point Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
Golf Course Government Refuse Collection Point Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
Government Refuse Collection Point Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
Government Use (not elsewhere specified) Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
Helicopter Landing Pad Holiday Camp House (other than rebuilding of New Territories Exempted House or
Holiday Camp House (other than rebuilding of New Territories Exempted House or
House (other than rebuilding of New Territories Exempted House or
Territories Exempted House or
replacement of existing domestic
building by New Territories
Exempted House permitted
under the covering Notes)
Petrol Filling Station
Place of Recreation, Sports or Culture
Public Transport Terminus or Station
Public Utility Installation
Public Vehicle Park
(excluding container vehicle)
Radar, Telecommunications Electronic
Microwave Repeater, Television
and/or Radio Transmitter Installation
Religious Institution
Residential Institution
School
School Service Deservoir
Service Reservoir Social Walfara Essility
Social Wellate Facility Utility Installation for Drivata Project

GREEN BELT

<u>GREEN BELT</u> (Cont'd)

Planning Intention

The planning intention of this zone is primarily for defining the limits of urban and sub-urban development areas by natural features and to contain urban sprawl as well as to provide passive recreational outlets. There is a general presumption against development within this zone.

Remarks

Any *diversion of stream*, filling of land or excavation of land, including that to effect a change of use to any of those specified in Columns 1 and 2 above or the uses or developments always permitted under the covering Notes (except public works co-ordinated or implemented by Government, and maintenance, repair or rebuilding works), shall not be undertaken or continued on or after the date of the first publication in the Gazette of the notice of the draft Lam Tei and Yick Yuen Outline Zoning Plan No. S/TM-LTYY/5 without the permission from the Town Planning Board under section 16 of the Town Planning Ordinance.

Column 1	Column 2
Uses always permitted	Uses that may be permitted with or
	without conditions on application
	to the Town Planning Board
Agricultural Use (other than Plant Nursery)	Barbecue Spot
Nature Reserve	Field Study/Education/Visitor Centre
Nature Trail	Government Refuse Collection Point
On-Farm Domestic Structure	Government Use (not elsewhere specified)
Picnic Area	Holiday Camp
Wild Animals Protection Area	House (Redevelopment only)
	Public Convenience
	Public Utility Installation
	Radar, Telecommunications Electronic
	Microwave Repeater, Television
	and/or Radio Transmitter Installation
	Tent Camping Ground
	Utility Installation for Private Project

CONSERVATION AREA

Planning Intention

This zoning is intended to protect and retain the existing natural landscape, ecological or topographical features of the area for conservation, educational and research purposes and to separate sensitive natural environment such as Country Park from the adverse effects of development.

There is a general presumption against development in this zone. In general, only developments that are needed to support the conservation of the existing natural landscape or scenic quality of the area or are essential infrastructure projects with overriding public interest may be permitted.

<u>Remarks</u>

- (a) No redevelopment, including alteration and/or modification, of an existing house shall result in a total redevelopment in excess of the plot ratio, site coverage and height of the house which was in existence on the date of the first publication in the Gazette of the notice of the draft development permission area plan.
- (b) Any *diversion of stream*, filling of land or excavation of land, including that to effect a change of use to any of those specified in Columssns 1 and 2 above or the uses or developments always permitted under the covering Notes (*except public works co-ordinated or implemented by Government, and maintenance or repair works*), shall not be undertaken or continued on or after the date of the first publication in the Gazette of the notice of the draft Lam Tei and Yick Yuen Outline Zoning Plan No. S/TM-LTYY/4 without the permission from the Town Planning Board under section 16 of the Town Planning Ordinance.

APPROVED DRAFT LAM TEI AND YICK YUEN OUTLINE ZONING PLAN NO. S/TM-LTYY/10A

EXPLANATORY STATEMENT

APPROVED DRAFT LAM TEI AND YICK YUEN OUTLINE ZONING PLAN NO. S/TM-LTYY/10A

EXPLANATORY STATEMENT

	CONTENTS	Page
1.	INTRODUCTION	1
2.	AUTHORITY FOR THE PLAN AND PROCEDURE	1
3.	OBJECT OF THE PLAN	23
4.	NOTES OF THE PLAN	3
5.	THE PLANNING SCHEME AREA	3 4
6.	POPULATION	4
7.	OPPORTUNITIES AND CONSTRAINTS	45
8.	GENERAL PLANNING INTENTION	5 6
9.	 LAND-USE ZONINGS 9.1 Commercial 9.2 Comprehensive Development Area 9.3 Residential (Group A) 9.34 Residential (Group B) 9.45 Residential (Group C) 9.56 Residential (Group D) 9.67 Residential (Group E) 9.78 Village Type Development 9.89 Government, Institution or Community 9.910 Open Space 9.1011 Other Specified Uses 9.1412 Green Belt 9.1213 Conservation Area 	67 7 8 79 810 910 911 1011 1112 1112 1113 1213 1214 1314
10.	TRANSPORT AND COMMUNICATION	13 15
11.	UTILITY SERVICES	14 15
12	CULTURAL HERITAGE	15 16
13.	IMPLEMENTATION	15 17
14.	PLANNING CONTROL	15 17

APPROVED DRAFT LAM TEI AND YICK YUEN OUTLINE ZONING PLAN NO. S/TM-LTYY/10A

(Being an Approved *a Draft* Plan for the Purposes of the Town Planning Ordinance)

EXPLANATORY STATEMENT

Note : For the purposes of the Town Planning Ordinance, this statement shall not be deemed to constitute a part of the Plan.

1. <u>INTRODUCTION</u>

This Explanatory Statement is intended to assist an understanding of the approved *draft* Lam Tei and Yick Yuen Outline Zoning Plan (OZP) No. S/TM-LTYY/10A. It reflects the planning intentions and objectives of the Town Planning Board (the Board) for various land-use zonings of the Plan.

2. <u>AUTHORITY FOR THE PLAN AND PROCEDURE</u>

- 2.1 The land within the Lam Tei and Yick Yuen OZP was previously included in the draft Lam Tei and Yick Yuen Development Permission Area (DPA) Plan No. DPA/TM-LTYY/1 which was exhibited for public inspection under section 5 of the Town Planning Ordinance (the Ordinance) on 18 June 1993.
- 2.2 On 30 April 1996, the draft Lam Tei and Yick Yuen DPA Plan No. DPA/TM-LTYY/1 was approved by the then Governor in Council under section 9(1)(a) of the Ordinance.
- 2.3 On 13 March 1995, under the power delegated by the then Governor, the then Secretary for Planning, Environment and Lands, directed the Board, under section 3(1)(a) of the Ordinance, to prepare an OZP for the area of Lam Tei and Yick Yuen.
- 2.4 On 7 June 1996, the draft Lam Tei and Yick Yuen OZP No. S/TM-LTYY/1 was exhibited for public inspection under section 5 of the Ordinance. The OZP was subsequently amended and exhibited for public inspection under section 7 of the Ordinance to reflect the changing circumstances.
- 2.5 On 23 May 2000, the Chief Executive in Council (CE in C), under section 9(1)(a) of the Ordinance, approved the draft Lam Tei and Yick Yuen OZP which was subsequently renumbered as S/TM-LTYY/3.
- 2.6 On 16 December 2003, the CE in C referred the approved Lam Tei and Yick Yuen OZP No. S/TM-LTYY/3 to the Board for amendment under section 12(1)(b)(ii) of the Ordinance. *The OZP was then amended twice and exhibited for public inspection under section 5 or 7 of the Ordinance to reflect the changing circumstances.* On 4 June 2004, the draft Lam Tei and Yick Yuen OZP No. S/TM LTYY/4 was exhibited for public inspection under section 5 of the Ordinance. On 15 April 2005, the draft Lam Tei and Yick Yuen OZP No.

S/TM-LTYY/5 was exhibited for public inspection under section 7 of the Ordinance.

- 2.7 On 14 March 2006, the CE in C, under section 9(1)(a) of the Ordinance, approved the draft Lam Tei and Yick Yuen OZP, which was subsequently renumbered as S/TM-LTYY/6.
- 2.8 On 8 July 2014, the CE in C referred the approved Lam Tei and Yick Yuen OZP No. S/TM-LTYY/6 to the Board for amendment under section 12(1)(b)(ii) of the Ordinance. The OZP was subsequently amended and exhibited for public inspection under section 5 of the Ordinance.
- 2.9 On 8 September 2015, the CE in C, under section 9(1)(a) of the Ordinance, approved the draft Lam Tei and Yick Yuen OZP, which was subsequently renumbered as S/TM-LTYY/8. On 18 September 2015, the approved Lam Tei and Yick Yuen OZP No. S/TM-LTYY/8 was exhibited for public inspection under section 9(5) of the Ordinance.
- 2.10 On 26 April 2016, the CE in C referred the approved Lam Tei and Yick Yuen OZP No. S/TM-LTYY/8 to the Board for amendment under section 12(1)(b)(ii) of the Ordinance. The reference back of the OZP was notified in the Gazette on 6 May 2016 under section 12(2) of the Ordinance. The OZP was subsequently amended and exhibited for public inspection under section 5 of the Ordinance.
- 2.11 On 26 May 2017, the draft Lam Tei and Yick Yuen OZP No. S/TM-LTYY/9 was exhibited fsor public inspection under section 5 of the Ordinance. The amendment on the Plan involved the excision of the northern part from the planning scheme area for incorporation into the draft Hung Shui Kiu and Ha Tsuen OZP No. S/HSK/1. During the two month exhibition period, a total of 5 valid representations were received. On 25 August 2017, the representations were published for three weeks for public comments and a total of 4 valid comments on the representations (comments) were received. On 17 April 2018, the Chief Executive, under section 8(2) of the Ordinance, agreed to extend the statutory time limit for the Board to submit the draft OZP to the CE in C for approval for a period of six months. After giving considerations to the representations and comments, the Board on 25 May 2018 decided not to propose any amendment to the draft OZP to meet the representations.
- 2.121 On 16 October 2018, the CE in C, under section 9(1)(a) of the Ordinance, approved the draft Lam Tei and Yick Yuen OZP, which was subsequently renumbered as S/TM-LTYY/10. On 26 October 2018, the approved Lam Tei and Yick Yuen OZP No. S/TM-LTYY/10 (the Plan)-was exhibited for public inspection under section 9(5) of the Ordinance.
- 2.12 On 3 November 2020, the CE in C referred the approved Lam Tei and Yick Yuen OZP No. S/TM-LTYY/10 to the Board for amendment under section 12(1)(b)(ii) of the Ordinance. The reference back of the OZP was notified in the Gazette on 13 November 2020 under section 12(2) of the Ordinance.
- 2.13 On XX XXX 2021, the draft Lam Tei and Yick Yuen OZP No. S/TM-LTYY/11 (the Plan) was exhibited for public inspection under section 5 of the Ordinance.

The amendments on the plan mainly involved the rezoning of (i) two areas to the north of Hong Po Road, to the east of Tsing Shan Firing Range and to the west of Castle Peak Road – Lam Tei from "Residential (Group E)" ("R(E)") and "Green Belt" ("GB") to "Residential (Group A)" ("R(A)"); and (ii) an area to the east of Tsing Shan Firing Range from "GB" to "Government, Institution or Community" ("G/IC").

3. <u>OBJECT OF THE PLAN</u>

- 3.1 The object of the Plan is to indicate the broad land-use zonings for the Lam Tei and Yick Yuen area so that development and redevelopment within the area can be put under statutory planning control. It also provides the planning framework for preparing more detailed non-statutory plans which form the basis for public works planning and site reservation for various uses.
- 3.2 The Plan is to illustrate the broad principles of development and control only. It is a small-scale plan and the road *transport* alignments and boundaries between the land-use zones may be subject to minor adjustments as detailed planning proceeds.
- 3.3 Since the Plan is to show broad land use zonings, there would be situations in which small strips of land not intended for building development purposes and carry no development right under the lease, such as the areas restricted as non-building area or for garden, slope maintenance and access road purposes, are included in the residential zones. The general principle is that such areas should not be taken into account in plot ratio and site coverage calculation. Development within residential zones should be restricted to building lots carrying development right in order to maintain the character and amenity of the Lam Tei and Yick Yuen area and not to overload the road network in this area.
- 3.4 The boundary of the area is delineated having regard to physical and topographical features such as roads, drainage channels and hills. Therefore, the area boundary does not necessarily follow the Heung boundaries which are used for administration purpose only. Also, the name of the Plan is geographical in nature and would not cause implications on development rights, particularly Small House applications.

4. <u>NOTES OF THE PLAN</u>

- 4.1 Attached to the Plan is a set of Notes which shows the types of uses or developments which are always permitted within the area *Planning Scheme Area (the Area)* and in particular zones and which may be permitted by the Board, with or without conditions, on application. The provision for application for planning permission under section 16 of the Ordinance allows greater flexibility in land use planning and control of development to meet changing needs.
- 4.2 For the guidance of the general public, a set of definitions that explains some of the terms used in the Notes may be obtained from the Technical Services
Division of the Planning Department (*PlanD*) and can be downloaded from the Board's website at http://www.info.gov.hk/tpb.

5. <u>THE PLANNING SCHEME AREA</u>

- 5.1 The Planning Scheme Area (the Area) is about 474.93 ha. It is located in the south-western part of the North West New Territories (NWNT). It is bounded by Hung Shui Kiu New Development Area (NDA) and Yuen Tau Shan in the north, Tan Kwai Tsuen in the north-east, Tai Lam Country Park in the east and south-east, the Tsing Shan Firing Range in the west, Fu Tai Estate, Tsz Tin Tsuen and Po Tong Ha in the south. The boundary of the Area is shown in a heavy broken line on the Plan.
- 5.2 The Area is situated to the north of Tuen Mun New Town. It extends northwards to the edge of Hung Shui Kiu NDA and is characterised by a land-use pattern which is a mixture of industrial workshops brownfield operations and residential dwellings. Traditional villages and agricultural land uses exist side by side with squatter housing and industrial workshops brownfield operations in temporary structures. The existing road networks in the Area comprise Castle Peak Road and several other minor roads branching off from it. Yuen Long Highway (YLH) and the Kong Sham Western Highway (KSWH) traverse the Area from the north-east to south-west, and from the north-west to south-east respectively.
- 5.3 In recognition of the traditional burial right of the indigenous villagers, the existing burial grounds in the Area have been preserved.
- 5.4 The present conditions in the Area are unsatisfactory because of the proliferation of container storage *brownfield operations* and other non-compatible uses, poor vehicular access, stream pollution and general visual squalor and urban sprawl.

6. <u>POPULATION</u>

- 6.1 Based on the 2016 Population **b***B*y-census, the population of the Area was estimated by the Planning Department PlanD as about 22,150 persons. *It is estimated that the total planned population of the Area would be about 89,100*. The population concentrates along Fuk Hang Tsuen Road, in and around nine recognized villages in the Area including Chung Uk Tsuen, Sun Fung Wai, Nai Wai, Tsing Chuen Wai, Tuen Tsz Wai, San Hing Tsuen, Lam Tei Tsuen, Tuen Mun San Tsuen (also known as Lam Tei San Tsuen) and To Yuen Wai as well as the non-indigenous villages such as Tsoi Yuen Tsuen, Fuk Hang Tsuen and Fu Tei Ha Tsuen.
- 6.2 Apart from usual natural growth in the existing villages, future expansion of population of the Area will be concentrated in the areas zoned for residential use along Castle Peak Road and areas to the north-east and north-west of Lam Tei Interchange. It is estimated that the planned population for the Area would be about 28,800 persons. There is no definite programme for this anticipated level of population built-up and the planned population growth and distribution will be

for long-term consideration.

7. <u>OPPORTUNITIES AND CONSTRAINTS</u>

- 7.1 **Opportunities**
 - 7.1.1 The completion of Route 3, the New Territories Circular Road, Yuen Long Highway YLH, the West Rail (WR) *Tuen Ma Line (TML)* and the Light Rail (LR) network has improved the accessibility of the Area. The KSWH provides a strategic road link for the Area. The Siu Hong Station of the WRTML is in close proximity to the southern boundary of the Area. This increases the development potential and provides opportunities for comprehensive development in the Lam Tei and Yick Yuen area.
 - 7.1.2 According to the Agriculture, Fisheries and Conservation Department, a few pockets of agricultural land in the Area are still under active cultivation and they are mainly confined to market gardening.
- 7.2 Constraints
 - 7.2.1 There is an ongoing sewerage project, including *The* construction of village sewerage along Lam Tei Main Street, for the Lam Tei area *was completed in 2019*. The contract was commenced in July 2015 with target completion in 2019 in phases. The Drainage Improvement Works in Nai Wai was completed in May 2012. Before the basic infrastructure is completed, only low density and small-scale developments can be permitted subject to the provision of satisfactory arrangements for water supply, sewerage, drainage, flood mitigation and access facilities.
 - 7.2.2 The WR TML, LR and KSWH in the Area pose constraints to the developments along their alignments in terms of vehicular/pedestrian access and environmental impact. Mitigation measures will be required when nearby development proceeds.
 - 7.2.3 There is an existing 400kV overhead power line running along the southern boundary of the Area. Furthermore, the alignment of the integral part of the Black Point 400kV Transmission System from Black Point to Sha Tin, runs across from Yick Yuen in the west passing through a narrow area between Chung Uk Tsuen and Sun Fung Wai to the east of the Area. According to the Hong Kong Planning Standards and Guidelines (HKPSG), only low-rise development can be allowed within the 50 metres wayleave corridor of the 400kV overhead power lines. Therefore, land-use designation on the OZP has to take account of this constraint.
 - 7.2.4 There are nine recognized villages in the Area and land has to be reserved to meet the forecasted Small House demand.
 - 7.2.5 The potential suburban development area in Lam Tei and Yick Yuen

should be confined to the plain area. The steep slopes of the Tsing Shan Firing Range and the Tai Lam Country Park are more suitable for conservation purposes.

7.2.6 The petrol filling station at Fuk Hang Tsuen Road is also a liquefied petroleum gas (LPG) filling station. The LPG filling station is regarded as a notifiable gas installation subject to the control of the Gas Safety Ordinance (Cap. 51). For developments in close proximity to the LPG filling station, the respective developers should conduct a detailed quantitative risk assessment to ascertain the risk levels posed by the LPG filling station and implement mitigation measures identified in the assessment to comply with the Government Risk Guidelines stated in the HKPSG.

8. <u>GENERAL PLANNING INTENTION</u>

8.1 The Area is situated along the Tuen Mun and Yuen Long Corridor located close to the Tuen Mun New Town. It extends northwards from the northern fringe of Tuen Mun New Town to Hung Shui Kiu NDA. Due to its geographical location and transport network, the general planning intention is to develop the Area for suburban development between the two existing urban centres of Tuen Mun New Town and Yuen Long Town. Castle Peak Road, the WR-TML and the LR traverse the Area from north to south. With the convenience of the public transport network, *high density*/relatively higher intensity of developments will be concentrated in the Lam Tei Local Centre area, along Castle Peak Road and near Lam Tei Interchange and public transport network. For areas further away from these centres and existing and planned infrastructural provisions, developments with less intensity will be envisaged. The general planning intention for the Area is:

(a) to encourage upgrading of the environment through comprehensively planned developments;

- (ab) to designate suburban residential developments to appropriate locations along committed and planned infrastructure corridors;
- (bc) to develop and maintain Lam Tei as a local centre in order to serve the residents in the Area;
- (ed) to reserve sufficient land for village expansion;
- (de) to designate land for strategic road and railway link including the LR and the WR TML; and
- (e) to encourage upgrading for the environment through comprehensively planned development by private sectors and infrastructural improvement works by Government departments; and
- (f) to retain the countryside character of the upland portion of the Area.

8.2 In the designation of various zones in the Area, considerations have been given to the natural environment, physical landform, existing settlements, land status, availability of infrastructure, local development pressures, Territorial Development Strategy Review and the North West New Territories (Yuen Long District) Development Statement Study under the North West New Territories Development Strategy Review. Other than the above, buildings and places of historical interest have been preserved in the Area as far as possible.

9. <u>LAND-USE ZONINGS</u>

- 9.1 <u>Commercial ("C")</u> : Total Area : 0.69 ha
 - 9.1.1 This zone is intended primarily for commercial developments, which may include shop, services, place of entertainment and eating place, functioning mainly as local shopping centre(s) serving the immediate neighbourhood.
 - 9.1.2 An elongated parcel of land to the east of San Hing Tsuen which is sandwiched by the LR and Castle Peak Road is zoned "C". At present, Lam Tei Vegetable Collection Depot and village type houses with commercial use are on this land. In order to be compatible with the character of the neighbourhood, developments in this zone are restricted to a maximum plot ratio of 3.6 and a maximum building height of 12 storeys including car park (36m). To provide flexibility for innovative design adapted to the characteristics of particular sites, minor relaxation of the above restrictions may be considered by the Board through the planning permission system. Each proposal will be considered on its individual planning merits.

9.2 <u>Comprehensive Development Area ("CDA")</u> : Total Area : 6.99 ha

- 9.2.1 This zone is intended for comprehensive development/redevelopment of the area for residential use with the provision of commercial, open space and other supporting facilities, if any. The zoning is to facilitate appropriate planning control over the development mix, scale, design and layout of development, taking account of various environmental, traffic, infrastructure and other constraints. Such zoning is to facilitate the phasing out of the sporadic rural industrial activities and temporary structures which have been incompatible with the adjoining developments. Although the existing uses are tolerated, any development/redevelopment proposals on sites under this zoning should be submitted to the Board in the form of a master layout plan for consideration, except as otherwise expressly provided that it is not required by the Board. When approved by the Board, a copy of the approved Master Layout Plan shall be made available for public inspection in the Land Registry pursuant to section 4A(3) of the Ordinance.
- 9.2.2 There are two sites zoned "CDA". The first site (about 5.08 ha) is located to the west of Fuk Hang Tsuen Road near Lam Tei Tsuen. The development shall not result in a total development or redevelopment

intensity in excess of a maximum domestic plot ratio of 2.0 for residential use and a maximum non-domestic plot ratio of 0.11 for commercial and retailing uses and a maximum building height of 15 storeys excluding car park (45m). The area is largely developed, now known as The Sherwood. The rest of the area consists mostly of temporary structures.

- 9.2.3 The other area zoned "CDA" (about 1.91 ha) is located to the north of Yuen Long Highway near Fuk Hang Tsuen. To put the development under better control, the site is zoned "CDA". The development shall not result in a total development or redevelopment in excess of a maximum plot ratio of 0.54 and a maximum building height of 6 storeys over single-storey car park. The site is predominantly vacant.
- 9.2.4 As the two "CDA" sites are located adjacent to existing main roads, noise mitigation measures such as the use of screening structures and the adoption of self-protective building layout would need to be implemented. To provide flexibility for innovative design adapted to the characteristics of particular sites, minor relaxation of the above restrictions for the two "CDA" sites may be considered by the Board through the planning permission system. Each proposal will be considered on its individual planning merits.

9.3 <u>Residential (Group A) ("R(A)")</u>: Total Area : 21.52 ha

- 9.3.1 The "R(A)" zone is intended primarily for high-density residential developments. Under this zoning, commercial uses are always permitted on the lowest three floors of a building or in the purpose-designed non-residential portion of an existing building.
- 9.3.2 Two sites to the north of Tuen Mun New Town are intended for public housing developments. Government, institution or community (GIC) facilities including school(s) and social welfare facilities as well as public transport termini will be provided within these sites. These sites are subject to a maximum plot ratio of 6.5 and a maximum building height of 160mPD.
- 9.3.3 In determining the maximum plot ratio of the development and/or redevelopments, any floor space that is constructed or intended for use solely as public vehicle parks, public transport facilities and GIC facilities, as required by the Government, may be disregarded to facilitate the provision of these facilities.
- 9.3.4 An Air Ventilation Assessment Expert Evaluation (AVA-EE) has been carried out for these sites and concluded that design measures, including building separations and setbacks, would alleviate the potential air ventilation impacts on the surrounding wind environment. A quantitative AVA shall be carried out at the detailed design stage. Requirements of the design measures and quantitative AVA shall be incorporated in the planning brief(s) for implementation as appropriate.

- 9.3.5 Planning brief(s) setting out the planning parameters and the design requirements of public housing development will be prepared to guide the future development.
- 9.3.6 To provide flexibility for innovative design adapted to the characteristics of particular sites, minor relaxation of the plot ratio and building height restrictions may be considered by the Board through the planning permission system. Each proposal will be considered on its individual merits.

9.34 Residential (Group B) ("R(B)") : Total Area : 9.54 ha

- 9.34.1 This zone is intended primarily for sub-urban medium-density residential developments in rural areas where commercial uses serving the residential neighbourhood may be permitted on application to the Board.
- 9.34.2 As the zoned areas are located adjacent to Castle Peak Road and Fuk Hang Tsuen Road, noise mitigation measures such as the use of screening structures and the adoption of self-protective building layout would need to be implemented.
- 9.34.3 There are 3 sub-areas within this zone.
 - (a) <u>Residential (Group B) 1 ("R(B)1")</u>: Total Area: 4.04 ha

The parcel of land located between the Nullah and Castle Peak Road near San Hing Tsuen is zoned for "R(B)1". Residential developments within this zone are restricted to a maximum plot ratio of 1.0, a maximum site coverage of 40% and a maximum building height of 4 storeys over single-storey car park (15m).

(b) <u>Residential (Group B) 2 ("R(B)2")</u>: Total Area: 3.20 ha

The parcel of land located to the south-east of Hung Shui Kiu NDA namely Bauhinia Garden and Wo Ping San Tsuen is zoned for this purpose. Residential developments within this zone are restricted to a maximum plot ratio of 1.26, a maximum site coverage of 40% and a maximum building height of 6 storeys over single-storey car park (21m).

(c) <u>Residential (Group B) 3 ("R(B)3")</u>: Total Area: 2.30 ha

The parcel of land located at the junction of Fuk Hang Tsuen Road and Lam Tei Main Street is zoned for this purpose to reflect a completed residential development (i.e. Botania Villa). This residential scheme constitutes part of the Lam Tei Local Centre development. The residential developments within this zone are restricted to a maximum plot ratio of 2.1, a maximum site coverage of 40% and a maximum building height of 12 storeys (36m) excluding car park.

- 9.34.4 To provide flexibility for innovative design adapted to the characteristics of particular sites, minor relaxation of the above restrictions may be considered by the Board through the planning permission system. Each proposal will be considered on its individual planning merits.
- 9.45 <u>Residential (Group C) ("R(C)")</u> : Total Area : 2.52 ha
 - 9.45.1 This zone is intended primarily for low-rise, low-density residential developments where commercial uses serving the residential neighbourhood may be permitted on application to the Board.
 - 9.45.2 Under this zoning, residential developments are restricted to a maximum plot ratio of 0.4 and a maximum building height of 3 storeys (9m) including car park. To provide flexibility for innovative design adapted to the characteristics of particular sites, minor relaxation of the above restrictions may be considered by the Board through the planning permission system. Each proposal will be considered on its individual planning merits.
 - 9.45.3 Four areas on the two sides of the KSWH near Fuk Hang Tsuen Road are zoned for this purpose. At present, these areas are occupied by temporary structures for domestic purpose and rural workshops. As these four areas are located adjacent to KSWH and Fuk Hang Tsuen Road, noise mitigation measures such as the use of screening structures and the adoption of self-protective building layout would need to be implemented.
- 9.56 Residential (Group D) ("R(D)") : Total Area : 20.36 ha
 - 9.56.1 This zone is intended primarily for improvement and upgrading of existing temporary structures within the rural areas through redevelopment of existing temporary structures into permanent buildings. It is also intended for low-rise, low-density residential developments subject to planning permission from the Board. Within this zoned area, replacement or new houses are encouraged to be constructed in permanent materials. Replacement housing for temporary structures shall not result in a total redevelopment in excess of a maximum building area of 37.2m² and a maximum building height of 2 storeys (6m).
 - 9.56.2 Apart from the intention of residential upgrading, very low-rise and low-density residential development may be permitted on application to the Board. To be in line with the development intensity of existing domestic accommodation within the zoned area, residential development other than New Territories Exempted House shall not result in a total development in excess of a maximum plot ratio of 0.2, and a maximum building height of 2 storeys (6m). Generally, the applicant has to prove to the Board that the proposed development would have no or minimal adverse effects on the environment. To provide flexibility for innovative design adapted to the characteristics of particular sites, minor relaxation of the above restrictions may be considered by the Board through the planning permission system. Each proposal will be

considered on its individual planning merits.

- 9.56.3 Areas surrounding Tsoi Yuen Tsuen and Fuk Hang Tsuen are zoned "*R(D)*". For the areas located adjacent to industrial establishments, necessary mitigation measures should be provided to minimize industrial/residential (I/R) interface problems.
- 9.5.4 Areas surrounding Tsoi Yuen Tsuen and Fuk Hang Tsuen are zoned "R(D)". At present, the areas are mainly occupied by temporary structures for both domestic purpose and rural workshops.
- 9.67 <u>Residential (Group E) ("R(E)")</u> : Total Area 11.56 2.00 ha
 - 9.67.1 This zone is intended primarily for phasing out of existing industrial uses through redevelopment for residential use on application to the Board. Whilst existing industrial uses will be tolerated, new industrial developments are not permitted in order to avoid perpetuation of I/R interface problem.
 - 9.67.2 An *Three* areas to the south-west of San Hing Tsuen is *are* zoned "R(E)". These areas is *are* at present intermixed with structures for residential use, open storage and workshops. Since it may not be possible to phase out all the industrial uses at once, it is important to ensure that the residential development will be environmentally acceptable and not subject to I/R interface problems. The applicant will be required to submit adequate information to demonstrate that the new development will be environmentally acceptable, and suitable mitigation measures, if required, will be implemented to address the potential I/R interface problems. In addition, the applicant will have to prove to the Board that the proposed development would have no or minimal adverse impact on the area in terms of environmental quality, land-use compatibility, infrastructural provision and traffic requirement.
 - 9.67.3 New development within this zone is restricted to a maximum plot ratio of 1.0, a maximum site coverage of 40% and a maximum building height of 4 storeys over single-storey car park (15m). To provide flexibility for innovative design adapted to the characteristics of particular sites, minor relaxation of the above restrictions may be considered by the Board through the planning permission system. Each proposal will be considered on its individual planning merits.

9.78 Village Type Development ("V") : Total Area : 102.78 ha

9.78.1 The planning intention of this zone is to reflect existing recognized and other villages, and to provide land considered suitable for village expansion and reprovisioning of village houses affected by Government projects. Land within this zone is primarily intended for development of Small Houses by indigenous villagers. It is also intended to concentrate village type development within this zone for a more orderly development pattern, efficient use of land and provision of infrastructures and services. Selected commercial and community uses serving the needs of the

villagers and in support of the village development are always permitted on the ground floor of a New Territories Exempted House. Other commercial, community and recreational uses may be permitted on application to the Board.

- 9.78.2 The boundaries of the "V" zones are drawn up having regard to the existing village environs, the anticipated Small House demands for the next ten years, topography, site constraints and the provision of public services. Village expansion areas and other infrastructural improvements will be guided by detailed layout plans whenever applicable.
- 9.78.3 All recognized villages within the Area are zoned "V". These include San Hing Tsuen, Tuen Tsz Wai, Tsing Chuen Wai, Chung Uk Tsuen, Nai Wai, Sun Fung Wai, Lam Tei Tsuen, Tuen Mun San Tsuen and To Yuen Wai.
- 9.78.4 As the zoned areas are located adjacent to Castle Peak Road, Shun Tat Street and Yuen Long Highway, noise mitigation measures such as the use of screening structures and the adoption of self-protective building layout would need to be implemented.

9.89 Government, Institution or Community ("G/IC") : Total Area : 5.78 6.31 ha

- 9.89.1 This zone is intended primarily for the provision of Government, institution or community facilities serving the needs of the local residents and/or a wider district, region or the territory. It is also intended to provide land for uses directly related to or in support of the work of the Government, organizations providing social services to meet community needs, and other institutional establishments.
- 9.89.2 The Director of Environmental Protection (DEP) should be informed of any noise sensitive users within the "G/IC" zone which might be subject to noise impact and noise mitigation measures may be required.
- 9.89.3 There are four *five* sites zoned "G/IC" in the Area and three of them are existing uses. The first site is located at San Hing Tsuen covering an ex-primary school and Tze Tong. The second site covers the existing Miu Fat Buddhist Monastery and its affiliated secondary school. The third site is a waterworks installation located near Lam Tei Quarry. The last site located between the Nullah and Castle Peak Road is reserved for a primary school in the long term.

9.9.4 A site along the western boundary of the Area is zoned "G/IC" reserved for the provision of a fresh water service reservoir and a salt water service reservoir.

9.89.45A site to the east of Fuk Hang Tsuen Road is zoned "G/IC(1)". This area is intended for development of elderly care services. Developments within this sub-area are restricted to a maximum gross floor area of 35,000m² and a maximum building height of 50 metres above Principal Datum. To provide flexibility for innovative design adapted to the

characteristics of the site, minor relaxation of the above restrictions may be considered by the Board through the planning permission system. Each proposal will be considered on its individual planning merits.

- 9.910 Open Space ("O") : Total Area : 1.93 ha
 - 9.910.1This zone is intended primarily for the provision of outdoor open-air public space for active and/or passive recreational uses serving the needs of local residents as well as the general public.
 - 9.910.2The following areas are zoned "O":
 - (a) Chung Uk Children's Playground at Chung Uk Tsuen;
 - (b) Nai Wai Soccer Pitch at Nai Wai;
 - (c) Fuk Hang Playground on both sides of the junction of Fuk Hang Tsuen Road and Castle Peak Road;
 - (d) a public open space at Lam Tei Main Street; and
 - (e) a strip of land to the east of Bauhinia Garden opposite to Hung Shui Kiu NDA.
 - 9.910.3Additional open space will be provided within the residential zonings and "CDA" zones. The designation of open space at Lam Tei Quarry after its decommissioning would be subject to review as detailed planning proceeds.
- 9.101 Other Specified Uses ("OU") : Total Area : 29.66 ha

This zone covers sites designated for specified uses.

(a) <u>"OU" annotated "Quarry"</u>

This zone is intended primarily for quarry use. The existing Lam Tei Quarry is zoned for this purpose to reflect the existing use of the site for quarry operation and its ancillary uses.

(b) <u>"OU" annotated "Refuse Transfer Station"</u>

This zone is intended primarily for the provision of refuse transfer station. A site at the end of Shun Tat Street and adjacent to Yuen Long Highway is zoned for this use and is currently occupied by the North West New Territories Refuse Transfer Station and its ancillary uses. The site is allocated to DEP.

(c) <u>"OU" annotated "Petrol Filling Station"</u>

This zone is intended primarily for the provision of petrol filling station. Four sites for five existing petrol filling stations (PFSs) are zoned for this purpose to reflect the existing uses. Two PFSs are located to the north of Bauhinia Garden opposite to Hung Shui Kiu NDA. The third one is located near the junction of Castle Peak Road and Fuk Hang Tsuen Road in Lam Tei. The fourth one is located at the island site between Castle Peak Road and the LR while the fifth one is located near to the junction of Castle Peak Road and Ng Lau Road.

9.142 Green Belt ("GB") : Total Area : 110.68 98.19 ha

- 9.142.1The planning intention of this zone is primarily for defining the limits of urban and sub-urban development areas by natural features and to contain urban sprawl as well as to provide passive recreational outlets. There is a general presumption against development within this zone. The zoned areas may include foothills, lower hill slopes, spurs, isolated knolls, woodland, traditional burial ground or vegetated land which occur at the urban fringe. Limited developments may be permitted with or without conditions on application to the Board, and each application will be considered on its individual merits taking into account the relevant Town Planning Board Guidelines.
- 9.142.2The "GB" zone covers an area to the north-west of Tsing Chuen Wai, and the south-western corner and south-eastern portion of the Area. The strip of land along the western side of Yuen Long Highway is also zoned "GB" in order to provide a buffer area for the adjoining uses.
- 9.142.3As *diversion of stream*, filling of land and excavation of land may cause adverse drainage impacts on the adjacent areas and adverse impacts on the natural environment, permission from the Board is required for such activities.

9.1213 Conservation Area ("CA") : Total Area : 152.43 ha

- 9.123.1This zoning is intended to protect and retain the existing natural landscape, ecological or topographical features of the area for conservation, educational and research purposes and to separate sensitive natural environment such as Country Park from the adverse effects of development. There is a general presumption against development in this zone. In general, only developments that are needed to support the conservation of the existing natural landscape or scenic quality of the area or are essential infrastructure projects with overriding public interest may be permitted. The area under this zoning is mainly Government land and permitted burial grounds.
- 9.123.2Uses related to conservation purposes such as nature reserve and nature trail are permitted as of right. However, significant new developments are not permitted in this zone. Only a selective range of uses such as public convenience and tent camping ground which would have insignificant impact on environment and infrastructural provision could be permitted upon approval by the Board.
- 9.123.3The majority of the areas under this zoning are the flanks of the western

and eastern boundaries of the Area. They are sloping land adjoining Tai Lam Country Park and the Tsing Shan Firing Range.

9.123.4As diversion of stream, Ffilling of land and excavation of land may cause adverse drainage impacts on the adjacent areas and adverse impacts on the natural environment. In view of the conservation value of the area within this zone, permission from the Board is required for such activities.

10. TRANSPORT AND COMMUNICATION

10.1 Road and Rail Networks

- 10.1.1 Castle Peak Road, an important local distributor, provides pedestrian and vehicular access to the Area. Several roads branch off from Castle Peak Road serving the Area are Fuk Hang Tsuen Road, Shun Tat Street, Lam Tei Main Street and San Hing Road. Parallel to Castle Peak Road is the LR network which connects the Area with Yuen Long Town and Tin Shui Wai New Town in the north and Tuen Mun New Town in the south. The completion of Route 3 and the WR TML also greatly improve the accessibility of the Area. The KSWH provides a strategic road which links the Shenzhen Bay Bridge (which is also known as the Hong Kong-Shenzhen Western Corridor) at its landing point in Ngau Hom Shek with the Yuen Long Highway at Lam Tei.
- 10.1.2 It is also one of the planning intentions that the junctions especially those along Castle Peak Road should be improved through comprehensive planning.

10.2 <u>Transport Provision</u>

The Area is at present served by both rail-based and road-based public transport including the WR TML, LR, existing buses, public light buses and taxi services linking to Tin Shui Wai New Town and Yuen Long Town in the north and Tuen Mun New Town in the south.

11. <u>UTILITY SERVICES</u>

11.1 <u>Water Supply</u>

The existing water treatment works Water Treatment Works capacity available in NWNT will soon be has already been fully committed. Further treatment works capacity, if required, would be made available from the future extension to Ngau Tam Mei the Water Treatment Works in NWNT. A site zoned "G/IC" located along the western boundary of the Area is reserved for the provision of a new fresh water service reservoir and a new salt water service reservoir. Extension of water supply system will be required if there is a substantial increase in the future water demand arising from large residential developments.

11.2 <u>Sewerage and Sewage Treatment</u>

At present, public sewer along Castle Peak Road is provided to serve the Area. The DEP has completed a consultancy study on the review of Tuen Mun Sewerage Master Plan in 2003. The study includes the formulation of a new Sewerage Master Plan to extend the public sewer to the Area in the long-term. Developments in the Area must make own provision for suitable and satisfactory on-site sewage treatment prior to the availability of public sewer with adequate capacity for connection. Connection to public sewer shall be made when public sewer with adequate capacity is in place and connection to public sewer is required by the DEP.

11.3 <u>Electricity</u>

The Area has long been supplied with electricity. It is anticipated that there will not be any problem in the provision of electricity supply to the Area.

11.4 <u>Gas</u>

Gas pipelines have already been laid along Yuen Long Highway and Castle Peak Road. Piped gas supply to the Area is dependent on the demand of future developments.

12. <u>CULTURAL HERITAGE</u>

- 12.1 Within the *boundary of the* Area, there are graded is a Grade 1 historic buildings, i.e. To Ancestral Hall at Tuen Tsz Wai-(Grade 1) and Entrance Gate of Sun Fung Wai (Grade 3), as well as the Fu Tei Ha, Nai Wai Kiln, San Hing Tsuen, Tsing Chuen Wai and Tuen Tsz Wai Sites of Archaeological Interest. All of the above graded historic buildings and sites of archaeological interest are worthy of preservation.
- 12.2 The Antiquities Advisory Board (AAB) also released a list of new items in addition to the list of 1,444 historic buildings. These items are subject to grading assessment by the AAB. Details of the list of 1,444 historic buildings and the new items have been uploaded onto the website of the AAB at http://www.aab.gov.hk.
- 12.3 Prior consultation with the Antiquities and Monuments Office (AMO)-of the Leisure and Cultural Services Department should be made if any development, redevelopment or rezoning proposals might affect the above graded historic buildings/structures, new items pending grading assessment, sites of archaeological interest and their immediate environs. If disturbance to the site(s) of archaeological interest is unavoidable, an archaeological impact assessment shall be conducted to evaluate the archaeological impact imposed by the proposed works. If necessary, a qualified archaeologist shall be engaged to apply for a licence from the Antiquities Authority under the Antiquities and Monuments Ordinance (Cap. 53) for an archaeological field investigation and proposed appropriate mitigation measures to the satisfaction of the AMO.

13. <u>IMPLEMENTATION</u>

- 13.1 The Plan provides a broad land-use framework for development control and implementation of planning proposals. More detailed plans will be prepared as a basis for public works planning and private developments.
- 13.2 At present there is no overall programme for the provision of infrastructure within the Area. The implementation process will be gradual and may stretch over a long period depending on the availability of resources. It will be undertaken through the participation of both the public and private sectors.
- 13.3 The provision of infrastructure, e.g. road widening and laying of services will be implemented through the Public Works Programme and Rural Public Works as and when resources are available. Private developments will be effected principally through private sector initiatives to develop or redevelop their properties in accordance with the zoned uses indicated on the Plan, provided that their proposals meet Government requirements.

14. <u>PLANNING CONTROL</u>

- 14.1 The types of permitted developments and uses within the Area are listed in the Notes to the Plan. Unless otherwise specified, all building, engineering and other operations incidental to and all uses directly related and ancillary to the permitted developments and uses within the same zone are always permitted and no separate permission is required.
- 14.2 Uses of land or building which were in existence immediately before the first publication in the Gazette of the notice of the draft DPA Plan and which are not in compliance with the terms of the Plan may have adverse impact on the environment, drainage and traffic of the Area. Although no action is required to make such use conform to the Plan, any material change of such use or any other development (except minor alteration and/or modification to the development of the land or building in respect of such use which is always permitted) must be always permitted in terms of the Plan or, if permission is required, in accordance with a permission granted by the Board. The Board will consider these applications on their individual merits. Those alteration and/or modification works which may lead to an environmental improvement or upgrading to the Area may be considered favourably by the Board.
- 14.3 Planning applications to the Board will be assessed on individual merits. In general, the Board, in considering the planning applications, will take into account all relevant planning considerations which may include the departmental outline development plans and layout plans, and the guidelines published by the Board. The outline development plans and the layout plans are available for public inspection at the Planning Department *PlanD*. Guidelines published by the Board are available from the Board's website, the Secretariat of the Board and the Technical Services Division of the Planning applications can be downloaded from the Board's website and are available from the Secretariat of the Board of the Board's website and are available from the Secretariat of the Board are available for planning applications can be downloaded from the Board's Website and are available from the Secretariat of the Board, and the Technical Services Division and the relevant District Planning

Office of the Planning Department *PlanD*. Applications should be supported by such materials as the Board thinks appropriate to enable it to consider the applications.

14.4 Any development, other than those referred to in paragraph 14.1 above or in conformity with this Plan or with the permission of the Board, undertaken or continued on or after 18 June 1993 on land included in a plan of the Lam Tei and Yick Yuen DPA, may be subject to enforcement proceedings under the Ordinance. Any *diversion of stream and/or* filling of land and excavation of land in the relevant zones on or after the exhibition of the specific plan referred to in the Notes of the relevant zones without the permission from the Board may also be subject to enforcement proceedings.

TOWN PLANNING BOARD OCTOBER 2018 _____ 2021

Appendix V of RNTPC Paper No. 6/21

Agreement No. CE 68/2017 (CE) Site Formation and Infrastructural Works for the Development at San Hing Road and Hong Po Road, Tuen Mun - Feasibility Study

FINAL REPORT

198172/B&V/025/Issue 4 July 2021





土木工程拓展署 Civil Engineering and Development Department



Agreement No. CE 68/2017 (CE)

Site Formation and Infrastructural Works for the Development at San Hing Road and Hong Po Road, Tuen Mun – Feasibility Study

Final Report

198172/B&V/025/Issue 4

Civil Engineering and Development Department 5/F, Civil Engineering and Development Building 101 Princess Margaret Road, Homantin Kowloon, Hong Kong Black & Veatch Hong Kong Limited 43/F, AIA Kowloon Tower, 100 How Ming Street, Kwun Tong, Kowloon, Hong Kong

July 2021

DOCUMENT CONTROL		Agreement No. CE68/2017 (CE)	No: 198172/B&V/025		
AMENDMENT RECORD		Site Formation and Infrastructural Works for the Development at San Hing Road and Hong Po Road, Tuen Mun – Feasibility Study	Prepared by:	B&V	
Final Day	ant		Client: CEDD	Initial:	YLT
ғпа кер	ort		Client: CEDD	Date:	July 2021
Pages	Date	Issue No	Descriptio	on	Initials
All	October 2020	1	Issue to Government Depart	ments for Comments	YLT
All	February 2021	2	Issue to Government Depart	ments for Comments	YLT
All	June 2021	3	Issue to Government Depart	ments for Comments	YLT
All	July 2021	4	Issue to Government Depart	ments for Comments	YLT

*The Registered Recipient is responsible for destroying or marking as 'superseded' all superseded documents.

CONTENTS

1	INTRODUCTION	1
1.1	BACKGROUND	1
1.2	PROJECT DESCRIPTION	2
1.3	PURPOSE AND STRUCTURE OF THIS REPORT	3
2	PROPOSED DEVELOPMENT LAYOUT PLANS AND DEVELOPMENT PARAMETERS	4
2.1	General	4
2.2	SITE DESCRIPTION	4
2.3	BACKGROUND OF THE PLANNING OF SITES	5
2.4	PROPOSED DEVELOPMENT LAYOUT PLANS AND DEVELOPMENT PARAMETERS	6
3	PRELIMINARY AIR VENTILATION ASSESSMENT (AVA)	
3.1	General	8
3.2	SUMMARY ON SITE WIND AVAILABILITY AND AIR VENTILATION PERFORMANCE	8
3.3 3.4	RECOMMENDATIONS	10
5.4		11
4	PRELIMINARY TRAFFIC AND TRANSPORT IMPACT ASSESSMENT (TTIA)	12
4.1	General	
4.2	EXISTING ROAD NETWORKS	
4.5 1 1	Δαδασαγία ενα Τραξείς Ευδείας	12
4.5	PROPOSED ACCESS ARRANGEMENTS AND PUBLIC TRANSPORT FACILITIES	
4.6	OPERATIONAL TRAFFIC IMPACT ASSESSMENT	
4.7	PARKING PROVISIONS	19
4.8	PUBLIC TRANSPORT ASSESSMENT	19
4.9	SUMMARY	20
5	PRELIMINARY PEDESTRIAN CONNECTIVITY ASSESSMENT (PCA)	21
5.1	General	21
5.2	Existing Pedestrian Flows	21
5.3	EXISTING, PLANNED AND PROPOSED NEARBY PEDESTRIAN NETWORK	
5.4 5 5	REFERENCE PEDESTRIAN FLOWS	
5.5 5.6	SUMMARY	22
6	PRELIMINARY GEOTECHNICAL ASSESSMENT (GA)	
C 1		2, 27
6.1 6.2	GENERAL	
6.3	GROUNDWATER REGIME	
6.4	Geotechnical Consideration and Constraints	
6.5	Existing Man–made Slopes and Retaining Walls	28
6.6	NATURAL TERRAIN HAZARDS STUDY (NTHS)	30
6.7	PRELIMINARY NATURAL TERRAIN HAZARD ASSESSMENT	
6.8	MITIGATION STRATEGY	
0.9	Ο υμμακι	

7	PRELIMINARY SITE FORMATION ASSESSMENT (SFA)	
7.1	General	
7.2	SITE FORMATION DESIGN CONSIDERATION	
7.3	PROPOSED SITE FORMATION WORKS	35
7.4	EARTHWORK INVENTORY	
7.5	Summary	
8	PRELIMINARY DRAINAGE IMPACT ASSESSMENT (DIA)	
8.1	General	
8.2	Existing Drainage Condition	
8.3	Assessment Approach	
8.4	RECOMMENDATION	
8.5	Summary	
9	PRELIMINARY SEWERAGE IMPACT ASSESSMENT (SIA)	
9.1	General	41
9.2	Approach and Methodology	
9.3	ESTIMATED SEWAGE FLOW	
9.4	RECOMMENDATION	43
9.5	Summary	
10	PRELIMINARY WATER SUPPLY AND UTILITY IMPACT ASSESSMENT (WSUIA)	
10.1	General	45
10.2	EXISTING WATER SUPPLY SYSTEMS	45
10.3	WATER DEMAND OF THE DEVELOPMENT	45
10.4	IMPACT TO PROPOSED DEVELOPMENTS ON WATER SUPPLY	46
10.5	RECOMMENDATIONS FOR FRESH WATER SUPPLY	
10.6	RECOMMENDATIONS FOR SALT WATER SUPPLY	
10.7	EXISTING UTILITIES	
10.8	FINDINGS AND RECOMMENDATIONS ON UTILITIES	
10.9	SUMMARY	
11	PRELIMINARY SUSTAINABILITY ASSESSMENT (SA)	
11.1	General	52
11.2	APPROACHES FOR SUSTAINABILITY ASSESSMENT	52
11.3	Scenario	53
12	PRELIMINARY LAND REQUIREMENT STUDY (LRS)	
12.1	General	55
12.2	LAND RESUMPTION	55
12.3	GRAVES, KAM TAPS (URNS) OR SHRINES	56
12.4	AFFECTED HOUSES	56
12.5	AFFECTED AGRICULTURAL LANDS AND ORCHARDS	56
12.6	Summary	57
13	ENVIRONMENTAL IMPACT ASSESSMENT (EIA)	
13.1	General	58
13.2	ENVIRONMENTAL IMPACT ASSESSMENT ORDINANCE (EIAO)	58
13.3	AIR QUALITY IMPACT	59

13.4	NOISE IMPACT	60
13.5	WATER QUALITY IMPACT	62
13.6	SEWERAGE AND SEWAGE TREATMENT IMPLICATIONS	63
13.7	ECOLOGICAL IMPACT	63
13.8	IMPACT FROM ELECTRIC AND MAGNETIC FIELDS	65
13.9	LANDSCAPE AND VISUAL IMPACT	66
13.10	WASTE MANAGEMENT IMPLICATIONS	69
13.11	LAND CONTAMINATION IMPACT	70
13.12	IMPACT ON CULTURAL HERITAGE	71
13.13	ENVIRONMENTAL MONITORING AND AUDIT REQUIREMENTS	72
13.14	SUMMARY	73
14	PRELIMINARY LAND CONTAMINATION AND REMEDIATION STUDY (LCRS)	74
14.1	GENERAL	74
14.2	SITE APPRAISAL FINDINGS	74
14.3	SITE INVESTIGATION AND SAMPLING PLAN	74
14.4	Remediation	75
15	IMPLEMENTATION PROGRAMME	76
15.1	GENERAL	76
15.2	IMPLEMENTATION PROGRAMME	76
16	CONCLUSIONS AND RECOMMENDATION	79
16.1	GENERAL	79
16.2	AIR VENTILATION	79
16.3	TRAFFIC AND TRANSPORTATION	79
16.4	Pedestrian Connectivity	80
16.5	Geotechnical	81
16.6	SITE FORMATION	82
16.7	DRAINAGE, SEWERAGE, WATER SUPPLY AND UTILITIES	82
16.8	ENVIRONMENTAL, LANDSCAPE AND VISUAL, LAND CONTAMINATION AND SUSTAINABILITY	83
16.9	LAND REQUIREMENT	83

END OF TEXT

FIGURES

198172/B&V/FR/0101	Location Plan
198172/B&V/FR/0201	Preliminary Site Layout Plan
198172/B&V/FR/0202	Extent of Railway Protection Zone
198172/B&V/FR/0203	Outline Zoning Plan
198172/B&V/FR/0301	Mitigated Site Layout Plan
198172/B&V/FR/0401	Location of Key Road Junctions
198172/B&V/FR/0402	Location on Key Road Links
198172/B&V/FR/0501	Pedestrian Flow Paths (Sheet 1 of 2)
198172/B&V/FR/0502	Pedestrian Flow Paths (Sheet 2 of 2)
198172/B&V/FR/0601	Ground Investigation Records
198172/B&V/FR/0601a	Geological Profile (Sections 1-1 to 2-2)
198172/B&V/FR/0601b	Geological Profile (Section 3-3)
198172/B&V/FR/0601c	Geological Profile (Section 4-4)
198172/B&V/FR/0601d	Geological Profile (Section 5-5)
198172/B&V/FR/0602	Registered and Non-Registered Geotechnical Features
198172/B&V/FR/0603	Catchments for Natural Terrain Hazard Assessments
198172/B&V/FR/0604a	Proposed Hazard Mitigation Strategy
198172/B&V/FR/0604b	Proposed Hazard Mitigation Measures Cross–Section
198172/B&V/FR/0701	Proposed Site Formation Plan
198172/B&V/FR/0701a	Cross-Section of Site Formation Plan (Sheet 1 of 4)
198172/B&V/FR/0701b	Cross-Section of Site Formation Plan (Sheet 2 of 4)
198172/B&V/FR/0701c	Cross-Section of Site Formation Plan (Sheet 3 of 4)
198172/B&V/FR/0701d	Cross-Section of Site Formation Plan (Sheet 4 of 4)
198172/B&V/FR/0801	Catchment Plan
198172/B&V/FR/0802	Proposed Drainage Layout Plan
198172/B&V/FR/0803	Proposed U-channel Catchment Plan
198172/B&V/FR/0804	Proposed Hydraulic Network (After Development)
198172/B&V/FR/0901	Proposed Sewerage Layout Plan
198172/B&V/FR/1001	Fresh Water Flow System Diagram (Fresh Water)
198172/B&V/FR/1002	Proposed Fresh Water Supply Arrangement
198172/B&V/FR/1003a	Proposed Fresh Water Mains Layout Plan (Sheet 1 of 3)
198172/B&V/FR/1003b	Proposed Fresh Water Mains Layout Plan (Sheet 2 of 3)
198172/B&V/FR/1003c	Proposed Fresh Water Mains Layout Plan (Sheet 3 of 3)
198172/B&V/FR/1004	Proposed Salt Water Supply Arrangement
198172/B&V/FR/1005a	Proposed Salt Water Mains Layout Plan (Sheet 1 of 2)
198172/B&V/FR/1005b	Proposed Salt Water Mains Layout Plan (Sheet 2 of 2)
198172/B&V/FR/1301a	Locations of Considered Road Traffic Noise Mitigation Measures
	(Sheet 1 of 2)
198172/B&V/FR/1301b	Locations of Considered Road Traffic Noise Mitigation Measures
	(Sheet 2 of 2)
198172/B&V/FR/1302	Ecological Mitigation and Enhancement Plan
198172/B&V/FR/1303	Photomontage VSR1 Car Park, Tuen Tsz Wai
198172/B&V/FR/1304	Photomontage VSR2 from Public Open Area between Tuen Mun
	San Tsuen and Botania Villa

198172/B&V/FR/1305	Photomontage VSR3 from Pui Lin Garden, Siu Hang Tsuen
198172/B&V/FR/1306	Photomontage VSR4 from Tsing Lun Road Footbridge
198172/B&V/FR/1307	Photomontage VSR5 from Kei Lun Wai Children's Playground
198172/B&V/FR/1308	Photomontage VSR6 from Po Tong Ha
198172/B&V/FR/1309	Photomontage VSR7 from Raised Public Road Outside Siu Hong MTR Station
198172/B&V/FR/1310	Photomontage VSR8 from Tsing Chung Koon Road
198172/B&V/FR/1311a	Photomontage VSR9 from Miu Fat Monastery Viewing Platform (Sheet 1 of 2)
198172/B&V/FR/1311b	Photomontage VSR9 from Miu Fat Monastery Viewing Platform (Sheet 2 of 2)
198172/B&V/FR/1312	Photomontage VSR10 from Tuen Mun Trail, Route 1
198172/B&V/FR/1313	Photomontage VSR11 from between Castle Peak Trail and Por Lo Shan
198172/B&V/FR/1314a	Photomontage VSR12 from Chung Shan Hillside with Graves (Sheet 1 of 2)
198172/B&V/FR/1314b	Photomontage VSR12 from Chung Shan Hillside with Graves (Sheet 2 of 2)
198172/B&V/FR/1315	Photomontage VSR13 from West Railway heading South-west
198172/B&V/FR/1316	Photomontage VSR14 from Hong Po Road Roundabout
198172/B&V/FR/1317a	Outline Landscape Plan (Sheet 1 of 3)
198172/B&V/FR/1317b	Outline Landscape Plan (Sheet 2 of 3)
198172/B&V/FR/1317c	Outline Landscape Plan (Sheet 3 of 3)
198172/B&V/FR/1401	Location of Potential Contaminated Sites
198172/B&V/FR/1501a	Stages of Development (Sheet 1 of 2)
198172/B&V/FR/1501b	Stages of Development (Sheet 2 of 2)

TABLES

Table 2.1 – Existing Ground Level and Proposed Site Formation Level of Proposed	
Development	4
Table 2.2 – Proposed Development Parameters#	6
Table 2.3 – Detail Breakdown of the GFA#	7
Table 4.1 – Road Link Performance for Year 20311	4
Table 4.2 – Road Link Performance for Year 20361	5
Table 4.3 – Junction Performace for Year 20311	6
Table 4.4 – Juncction Performance for Year 20361	7
Table 4.5 – Junction Performance with Improvement 1	9
Table 5.1 – Description of Pedestrian Facility LOS2	3
Table 5.2 – Pedestrian Flow Paths and LOS Assessment2	4
Table 6.1 – Existing Registered Man–Made Features Identified to be Potentially Affected by	
the Proposed Works	9
Table 6.2 – Existing Non-Registered Geotechnical Features Identified to be Potentially	
Affected by the Proposed Works	0
Table 6.3 – Screening Results for Natural Terrain Hazards Assessment	0
Table 6.4 – Summary of Debris Mobility Modelling	1
Table 6.5 – Summary of Design Requirements3	1

Table 7.1 – Summary of Estimated Quantities of C&D Materials Generated During Site	
Formation	36
Table 8.1 – Determination of Flood Level in the Fluvial–Tidal Zone	39
Table 9.1 – Hydraulic Performance of SPSs	42
Table 9.2 – Average Dry Weather Flow	43
Table 9.3 – Hydraulic Performance of SPSs	44
Table 10.1 – Summary of Estimated Water Demand	45
Table 10.2 – Summary of Fresh Water Supply Schemes	47
Table 10.3 – Summary of Salt Water Supply Schemes	47
Table 10.4 – Review on Existing Utilities	49
Table 10.5 – Review on Proposed Utilities	50
Table 11.1 - Summary of Anticipated Impacts for each Sustainability Parameter	54
Table 12.1 – Total Area of Private Land Lots to be Affected	55
Table 12.2 – Existing Licenses to be Affected	55
Table 12.3 – Existing Private Land Lots to be Affected	55
Table 12.4 – Existing Graves/ Urns to be Affected	56
Table 12.5 – The Area of Affected Agricultural Lands and Orchards	56
Table 13.1 – Summary of Schedule 2 Designated Project	58
Table 13.2 – Summary of Predicted Cumulative Construction Dust Impact (After	
Implementation of Mitigation Measures)	59
Table 13.3 – Summary of Predicted Concentrations of Representative Air Pollutants during	
Operation Phase in Year 2029	60
Table 13.4 – Summary of Predicted Concentrations of Representative Air Pollutants during	
Operation Phase in Year 2033	60
Table 13.5 – Summary of Mitigated Road Traffic Noise Levels	61
Table 13.6 – Summary of the Habitat Loss	64
Table 13.7 – Significance of Landscape Impacts before Mitigation within Assessment Area	67
Table 13.8 – Landscape and Visual Mitigation Measures	68
Table 13.9 – Summary of EM&A Requirements	72
Table 15.1 – Summary of Implementation Programme within the PDA	77

LIST OF APPENDICES

- Appendix A Summary of Prevailing Winds, Major Breezways and Layout Plans
- Appendix B Access Arrangement and Public Transport Facilities
- Appendix C Layout of Proposed Junction Improvement

	Name	Signature	Date
Prepared	Alan TAM	PP Jorte	July 2021
Checked	Tony LEE		July 2021
Reviewed	Jeremy WONG	Teght	July 2021
Authorized	Edwin LO	A-	July 2021
L		0	

LIST OF ABBREVIATIONS

Abbreviation	Meaning
АМО	Antiquities and Monuments Office
API	Aerial Photograph Interpretation
AQO	Air Quality Objectives
AOI	Area of Influence
ASR	Air Sensitive Receiver
AVA	Air Ventilation Assessment
BDTM	Base District Traffic Model
C&D	Construction & Demolition
CDF	Channelized debris flow
CEDD	Civil Engineering and Development Department
CLP	China Light and Power Co Ltd.
CTS	Comprehensive Transport Study
DIA	Drainage Impact Assessment
DEA	Data envelopment analysis
DP	Designated Project
DSD	Drainage Services Department
EB	Eastbound
EIAO	Environmental Impact Assessment Ordinance
EMSD	The Electrical and Mechanical Services Department
EPD	Environmental Protection Department
FS	Feasibility Study
FW	Fresh Water
GB (Zoning)	Green Belt
GCC	General conditions of contract
GEO	Geotechnical Engineering Office
GFA	Gross Floor Area
GI	Ground Investigation
GLA	Government Land Allocations
GLL	Government Land License
GMB	Green Minibus
G/IC (Zoning)	Government, Institution or Community
HD	Housing Department
HKBN	Hong Kong Broadband Network Limited
НКССС	Hong Kong and China Gas Company Limited (Towngas)
НКНА	Hong Kong Housing Authority
HKPSG	Hong Kong Planning Standards and Guidelines
HPR Site	Hong Po Road Site
нкт	Hong Kong Telecom
HyD	Highways Department
I, D&C	Investigation, Design and Construction

Agreement No. CE 68/2017(CE) Site Formation and Infrastructural Works for the Development at San Hing Road and Hong Po Road, Tuen Mun – Feasibility Study

Abbreviation	Meaning
LandsD	Lands Department
LVIA	Landscape & Visual Impact Assessment
LCA	Landscape Characteristic Area
LOPSW P/S	Lok On Pai Salt Water Pumping Station
LR	Landscape Resource
LRT	Light Rail Transit
MTR	Mass Transit Railway
NB	Northbound
NEC	New Engineering Contract
NSR	Noise Sensitive Receiver
NTHS	Natural Terrain Hazard Study
NWNT	North West New Territories
OH Catchments	Open hillslope Catchments
OHL	Overhead Line
OPRS	On-site Pre-school Rehabilitation Services
OZP	Outline Zoning Plan
рси	Passenger Car Unit
PES	Preliminary Environmental Study
РН	Public Housing
PFC	Public Fill Committee
PLB	Public Light Bus
PME	Powered Mechanical Equipment
PPOF	Person Per Occupied Flat
РН	Public Housing
PTI	Public Transport Interchange
PTTIA	Preliminary Traffic and Transportation Impact Assessment
R (Zoning)	Residential
RC	Reserve Capacity
RCHE	Residential Care Home for the Elderly
RFC	Ratio of Flow to Capacity
RSP	Respirable Suspended Particulates
SB	Southbound
SDM	Stormwater Drainage. Manual
SHR Site	San Hing Road Site
SHR Site Extension	San Hing Road Site Extension
SPS	Sewage Pumping Station
SI	Site Investigation
SSF	Subsidised Sale Flats
S/R	Service Reservoir
SW	Salt Water
SWL	Sound Power Level
TD	Transport Department

Agreement No. CE 68/2017(CE) Site Formation and Infrastructural Works for the Development at San Hing Road and Hong Po Road, Tuen Mun – Feasibility Study

Abbreviation	Meaning
TML	Tuen Ma Line
TPZ	Tree Protection Zone
TGLA	Temporary Government Land Allocation
TGN37	Guidelines on Empirical Design of Flexible Barriers for Mitigating Natural. Terrain Open Hillslope Landslide Hazards
TPDM	Transport Planning & Design Manual
TPEDM	Territorial Population and Employment Data Matrices
TPRP	Tree Preservation and Removal Proposal
TSFR	Tsing Shan Firing Range
TSP	Total Suspended Particulates
VEB	Village Environ Boundary
VEP	Variation of Environmental Permit
VSRs	Visually Sensitive Receivers
V/C Ratio	Volume to Capacity Ratio
WB	Westbound
WIS	Western Interceptor Sewer
WRL	West Rail Line
WSD	Water Supplies Department

1 INTRODUCTION

1.1 Background

- 1.1.1 The Government is committed to increase the supply of land for housing development in a persistent manner to meet the keen housing demands of the public. To meet this policy objective, the sites at San Hing Road (SHR) and Hong Po Road (HPR) have been identified for potential public housing development. The Site Formation and Infrastructural Works for the Development at San Hing Road and Hong Po Road, Tuen Mun – Feasibility Study (the Study) commenced in February 2018 to ascertain the engineering feasibility and environmental acceptability of the proposed public housing development at these sites.
- 1.1.2 Black & Veatch Hong Kong Limited (B&V)¹ was commissioned by Civil Engineering and Development Department (CEDD) to examine the technical feasibility on developing public housing at SHR and HPR, Tuen Mun (the Project), location of which is shown in *Figure No. 198172/B&V/FR/0101.*
- 1.1.3 The Proposed Development Area (PDA) mainly comprises SHR Site, SHR Site Extension, HPR Site, the proposed Road L7 and the realigned Hong Po Road. The PDA is situated in Tuen Mun between Lam Tei Light Rail Transit (LRT) Station and Siu Hong West Rail Line² / Light Rail Transit (WRL/LRT) Interchange Station. SHR Site, SHR Site Extension and HPR Site are currently connected by Hong Po Road which is a single two-way road of 6m wide with substandard footpaths and road bends and is remote from the public transport network. On the south side of the PDA is Po Tong Ha and Tsz Tin Tsuen, while the west side is the Tsing Shan Firing Range (TSFR). The private housing site, Villa Pinada, which is a low density private residential development, is located to the north side of the PDA and is situated between HPR Site and SHR Site Extension.
- 1.1.4 The PDA currently falls within areas zoned "Residential (Group E)" ("R(E)") and "Green Belt" ("GB") on the approved Lam Tei and Yick Yuen Outline Zoning Plan (OZP) No. S/TM–LTYY/10 (the LTYY OZP) and "Residential (Group E) 1" ("R(E)1"), "GB", an area shown as 'Road' and slightly encroaches onto an area zoned "Village Type Development" ("V") on the approved Tuen Mun OZP No. S/TM/35 (the TM OZP).
- 1.1.5 The total study area under the Study is about 29.7 ha. Since it exceeds 20 ha, Environmental Impact Assessment (EIA) is required in accordance with the Item 1 under Schedule 3 of the EIA Ordinance (Cap. 499), i.e. "Engineering feasibility study of urban development projects with a study area covering more than 20 ha or involving a total population of more than 100,000". In this regard, the Study with an EIA Study is required to ascertain the engineering feasibility and environmental acceptability of the proposed public housing development.
- 1.1.6 While the sites at San Hing Road and Hong Po Road have the potential for public housing development, the proposed sites are subject to various development

¹ Currently named as "Binnies Hong Kong Limited"

² Currently named as "Tuen Ma Line"

constraints including inadequate infrastructure. A holistic review is required to confirm the feasibility of amending the land use of the sites for the proposed development. In order to support the subsequent proposed amendments to the OZPs by the Government, the Study will examine the feasibility on developing public housing and associated Government, Institutional and Community (GIC) facilities by conducting preliminary engineering and environmental assessments to formulate Infrastructure proposal, and the implementation strategies and programme for the infrastructure to suit the proposed public housing development.

1.2 Project Description

- 1.2.1 Throughout the study period under this Agreement, the boundaries of the proposed public housing sites have been refined for the purpose of optimizing land utilisation and addressing site constraints.
- 1.2.2 The aim of this Study is to assess the feasibility of site formation and infrastructural works to support the proposed public housing and school developments. The scope of the works in general include:
 - a) Site formation works;
 - b) Slope works and other geotechnical works;
 - c) Land decontamination works;
 - d) Roadworks (e.g. Proposed Road L7 and the realigned Hong Po Road);
 - e) Waterworks (including service reservoirs);
 - f) Sewerage works (including sewage pumping station);
 - g) Drainage works;
 - h) Landscaping works;
 - i) Public Transport Interchanges (PTIs);
 - j) Environmental mitigation measures;
 - k) Archaeological field survey and mitigation measures ; and
 - l) Other infrastructure works including utilities and road junction improvement works, etc.
- 1.2.3 A series of technical assessments need to be carried out under the Study to determine the feasibility of site formation and infrastructural works supporting the proposed public housing development. These assessments include, but not limited to:
 - a) Preliminary air ventilation assessment;
 - b) Preliminary traffic and transport impact assessment;
 - c) Preliminary pedestrian connectivity assessment;
 - d) Preliminary geotechnical assessment;
 - e) Preliminary site formation assessment;
 - f) Preliminary drainage impact assessment;

- g) Preliminary sewerage impact assessment;
- h) Preliminary water supply and utility impact assessment;
- i) Preliminary sustainability assessment;
- j) Preliminary land requirement study;
- k) Environmental impact assessment study; and
- l) Preliminary land contamination and remediation study.

1.3 Purpose and Structure of this Report

- 1.3.1 The purpose of this Final Report is to set out the key findings, recommendations and conclusions of the Study. It also includes the recommended scope of the site formation and infrastructural works.
- 1.3.2 The Final Report includes:
 - (a) findings, proposals and recommendations of the Study including the scope of the recommended site formation and infrastructural works;
 - (b) the results of the technical assessments; and
 - (c) the implementation strategy and frameworks, including recommendation on implementation and phasing for the infrastructure works.
- 1.3.3 The structure of the report is as follows:
 - Section 1 introduces the project background, the main tasks and purpose of this report;

Section 2 presents the preliminary development proposal;

- Sections 3 to 15 summarise the findings and recommendations of various technical assessments for the proposed public housing development, including air ventilation, traffic and transport, pedestrian connectivity, geotechnical, site formation, drainage, sewerage, water supply, utilities, sustainability, land requirement, environment, landscape and visual and land contamination and remediation;
- > Section 16 presents a tentative implementation programme; and
- Section 17 provides the conclusion to the report.

2 PROPOSED DEVELOPMENT LAYOUT PLANS AND DEVELOPMENT PARAMETERS

2.1 General

2.1.1 This chapter describes the proposed layout plans and the development parameters for the proposed development sites.

2.2 **Site Description**

- 2.2.1 The Site mainly comprises the SHR Site, SHR Site Extension and HPR Site, the proposed Road L7 and the realigned Hong Po Road. The preliminary layouts of the proposed sites are illustrated in *Figure No. 198172/B&V/FR/0201*. To support the proposed development, a number of infrastructure works including site formation works, construction of public housing and schools, provision of social welfare facilities, construction of roads, construction of drainage, sewerage and waterworks facilities, and construction of landscaping works etc., will be implemented.
- 2.2.2 The Site currently comprises slopes, open storage, workshops, low-rise village houses settlements, agricultural area, ice-making factories, warehouses and temporary structures, graves, urns and permitted burial grounds. The existing / proposed ground levels are summarized in *Table 2.1* below:

Table 2.1	1 – Existing Ground Level and Pro Proposed Develop	posed Site Formation Level of pment
	Existing Ground Level	Proposed Site Formation Level

Location	Existing Ground Level (mPD)	Proposed Site Formation Level (mPD)	
SHR Site	Approx. 5.3 to 14.8	Approx. 6 to 12	
SHR Site Extension	Approx. 9 to 16	Approx. 9.8 to 12.2	
HPR Site	Approx. 14 to 40	Approx. 8 to 21	

2.2.3 Tsing Shan Firing Range (TSFR) adjoins the north-western boundary of the Site. The Site is rural in nature and is characterized by brownfield operation intermixed with scattered low-rise/ village type developments. The Site is accessible by Hong Po Road, which is a single two-way road of 6m with substandard footpaths and road bends.

SHR Site

- 2.2.4 The SHR Site is located to the south of San Hing Tsuen and bounded by the San Hing Road to the north, Ng Lau Road to the east and Hong Po Road to the south. The SHR Site currently comprises brownfield operations intermixed with village type developments. The eastern part of the SHR Site falls within the Railway Protection Zone of MTR Corporation Limited. The Railway Protection Zone is shown in *Figure* No. 198172/B&V/FR/0202.
- 2.2.5 The existing village type developments are located very close to the brownfield sites and hence are affected by the brownfield operations to a certain extent. There are several drainage channels located within the SHR Site. CLP's 400kV Overhead Line (OHL) and the pylons are located to the south of SHR Site. The Tuen Mun Nullah is located at the east side of SHR Site. Details of the existing drainage system and

proposed drainage works will be discussed in **Section 8**.

SHR Site Extension

- 2.2.6 The SHR Site Extension is sandwiched by a well-established low rise residential development named Villa Pinada and northwestern boundary of the SHR Site. It is currently occupied by a marble factory and squatter structures. Immediately north of SHR Site Extension is the Permitted Burial Ground No. 6 (BURGD6).
- 2.2.7 There is an existing open channel running from the northwest to the south within the SHR Site Extension. A drainage channel is located at the south of the SHR Site Extension. These existing drainage channels would require proper diversion during the proposed site formation works. 400kV OHL and the pylons are located to the south side of the SHR Site Extension.

<u>HPR Site</u>

2.2.8 The HPR Site is located to the west of Villa Pinada and to the east of TSFR. To the north and west of HPR Site are the BURGD7 and BURGD8 respectively. The HPR Site is currently occupied by open storage, orchard and scattered residential dwellings. A DSD adit shaft building, which falls within the boundary of TSFR, is located to the northwest of the site.

2.3 Background of the Planning of Sites

- 2.3.1 Housing is one of the most important livelihood issues in Hong Kong to be addressed by the Government. Hence, increasing land supply for housing development has been one of the major focuses in the Policy Addresses for last few years. The 2011–2014 Policy Addresses had identified that increasing production of public housing as well as land supply for private housing development would be a key to tackle issues arising from the raising housing demand. Various measures had been recommended with an aim to expanding the land resources, which include exploring the possibility of converting agricultural land in some parts of the New Territories, currently used for industrial purposes or temporary storage into land for housing development. Under the Long Term Housing Strategy 2019, the split ratio of public / private housing of 70:30 is adopted continuously, and supply target for public housing is 301,000 for ten year from 2020/21 to 2029/30.
- 2.3.2 Currently, SHR Site mainly falls within an area zoned "Residential (Group E)" ("R(E)") on the LTYY OZP, with minor portion falls within an area zoned "Residential (Group E) 1" ("R(E)1") and an area shown as 'Road' on the approved Tuen Mun Outline Zoning Plan No. S/TM/35.
- 2.3.3 SHR Site Extension mainly falls within an area zoned "Green Belt" ("GB") on the LTYY OZP, with minor portion falls within an area zoned "GB" on the TM OZP.
- 2.3.4 HPR Site mainly falls within an area zoned "GB" on the LTYY OZP, with minor portion falls within an area zoned "GB" and an area shown as 'Road' on the TM OZP.
- 2.3.5 The details of the OZPs are shown in *Figure No. 198172/B&V/FR/0203*.

2.4 Proposed Development Layout Plans and Development Parameters

- 2.4.1 The PDA boundary has been refined in the feasibility study taking into account the site, land, traffic and transport, environmental, geotechnical and associated infrastructure constraints to develop an optimised development layout
- 2.4.2 The preliminary site layout plan is shown in *Figure No. 198172/B&V/FR/0201*, and are described as follows:
 - The area between SHR Site Extension and HPR Site was adjusted along the boundary of the open storage and the village;
 - The area between SHR Site and SHR Site Extension was excluded to minimize disturtance to existing graves;
 - The area to the north of western side of SHR Site was excluded to minimize disturbance to existing graves; and
 - The area at the eastern side of SHR Site was excluded due to the presence of existing pylon and to minimize disturbance to the existing residential clusters.
- 2.4.3 Based on the proposed development layouts, the development parameters for the proposed development, which are subject to review at later investigation and design stages, are summarized in **Table 2.2** & *Table 2.3* below:

Site	Zoning on the Current OZP	Proposed Land Use	No. of Flats/ Domestic Population	GIC Facilities	(m²)	Tentative Completion Year
SHR Site	"R(E)" and "GB" on the approved Lam Tei and Yick Yuen OZP No. Public S/TM- Housing LTYY/10 (PH) &			100–p Child Care Centre (CCC)	1,166	
				Neighbourhood Elderly Centre (NEC)	1333.2	
				60–p Special Child Care Centre (SCCC)	902	
		9,400/ 26,300	On-site Pre-school Rehabilitation Services (OPRS) Office Base	363	2031	
			Two 6-classrooms kindergartens	1,000		
				Retail	10,620	
	"R(E)",			Others*	21,942.8	
	"GB","V" and area			<u>Total Non-</u> <u>Domestic GFA</u>	<u>37,327.0</u>	
	shown as 'Road' on			Two 36 -classroom primary schools	14,000	
SHR Site Extension	the	the		Retail	1,000	
	approvedTuenPublicMun OZPHousingNo.(PH)S/TM/35		Others*	6,298		
		Housing	1,700/ 4,700	<u>Total Non-</u> Domestic GFA	<u>7,298.0</u>	2030
		(11)		Two 36 -classroom primary schools	21,000	

 Table 2.2 - Proposed Development Parameters#

Agreement No. CE 68/2017(CE) Site Formation and Infrastructural Works for the Development at San Hing Road and Hong Po Road, Tuen Mun – Feasibility Study

Final Report 198172/B&V/025/Issue 4

Site	Zoning on the Current OZP	Proposed Land Use	No. of Flats/ Domestic Population	GIC Facilities (m ²)		Tentative Completion Year
				One 30-classroom		
				secondary school		
				Retail	8,000	
				150-p Residential Care Home for the Elderly (RCHE) + 30-p Day Care Unit (DCU)	4,406.6	
HPR Site		Public Housing	10,500/	and Youth Services Centre (ICYSC)	1,388.2	2033
	(PH)	30,000	Neighbourhood Elderly Centre (NEC)	1333.2		
				Others*	21160.5	
				Two 9-classrooms kindergartens	1,520	
				<u>Total Non–</u> Domestic GFA	<u>37,808.5</u>	

Remarks:

subject to review at later investigation and design stages.

*allowed for provision of additional retail facilities or welfare facilities (to be reviewed in the detailed design stage).

Table 2.3 - Deta	il Breakdown	of the GFA [#]

Development Parameters	SHR Site	SHR Site Extension	HPR Site		
Overall Plot Ratio (PR)	6.5 (Assuming a domestic PR of 6 and a non-domestic PR of 0.5)				
Net Site Area	About 7.5 ha	About 1.5 ha	About 7.6 ha		
Population per flat		About 2.8			
No. of Units	9,400	1,700	10,500		
Max. Building Height (mPD)	About +150	About +150	About +150		
No. of Domestic Storeys	32-46	43	43-44		
School Provision	 Two 36-classroom primary schools 	 Two 36-classroom primary Schools One 30-classroom secondary school 	-		
Completion Year	2031	2030	2033		

Remarks:

subject to review at later investigation and design stages.

3 PRELIMINARY AIR VENTILATION ASSESSMENT (AVA)

3.1 General

- 3.1.1 Qualitative assessment (Expert Evaluation) of wind environments with the proposed public housing development have been carried out based on the preliminary layout plan.
- 3.1.2 Preliminary AVA has been conducted for the Project to establish guiding principles in air ventilation terms and evaluate the proposed developments in accordance with the Technical Guide for AVA for developments in Hong Kong.

3.2 Summary on Site Wind Availability and Air Ventilation Performance

- 3.2.1 After reviewing the development layout, the existing conditions of adjacent areas, the far-field and near-field topography, the site wind availability including the major breezeways, and future developments in the surrounding areas, it is considered that there would be some potential air ventilation impacts on the areas in the vicinity of the Site when the cumulative effect is reviewed in annual (NNE, NE, E, SE, SSE, S) and summer (SSE, S, SSW, SW) prevailing wind conditions.
- 3.2.2 Air ventilation to the nearest large development Villa Pinada, San Hing Tsuen, Yan Tin Estate, the proposed schools within the Sites, San Tat Lane and San Hing Road would be affected at different degrees subject to either annual or summer prevailing wind conditions in NE to SW wind directions.
- 3.2.3 The effect of surrounding topography on the wind environment of the Site is investigated. It is considered that a majority portion of strong winds from prevailing directions of NE to SW will pass through the Site as the wind directions are aligned with the direction of the valley running more or less in the north–south direction. Part of the prevailing winds coming from SE, SSE and SW will be weakened by the shielding effects of the adjacent mountains; while magnifications in wind speed due to downhill air movement and valley wind is expected from some vegetative hill slopes and some valleys near the Site.
- 3.2.4 The effect of building morphology on the wind environment of the Site due to the committed developments in vicinity is investigated. The existing Siu Hong Court and Yan Tin Estate may obstruct S and SSE wind to the Site. Tsing Lun Road and Tsz Tin Road serve as breezeways which allow wind penetration to the Project Site. A summary of prevailing winds and major breezeways is shown on **Appendices A-1** and **A-2**.
- 3.2.5 The layouts of each Sites are shown on **Appendices A-3** to **A-5** and the good design features are summarized below:

<u>HPR Site</u>

- 3.2.6 Most of the building separation are at least 15m.
- 3.2.7 Localised wind corridors which facilitate prevailing winds coming from North-east and South-west directions:
 - Between Blocks 1 and 2 which aligned in NE to SW direction
- Between Blocks 2 and 3 which aligned in NE to SW direction
- Between Blocks 8 and 9 which aligned in NE to SW direction

SHR Site Extension

- 3.2.8 The building gaps are designed in NE to SW direction which facilitate the penetration of prevailing wind from north-east and south-west quadrant.
- 3.2.9 Localised wind corridors which facilitate prevailing winds coming from North-east and South-west directions:
 - Between the residential blocks and the proposed Primary School Site C at the east which aligned in NE to SW direction
 - Between the residential blocks and the proposed Secondary school Site A which aligned in NE to SW direction

SHR Site

- 3.2.10 The building blocks (Blocks 5 to 9) are aligned in NE to SW direction which facilitate the penetration of prevailing wind from north-east and south-west quadrant.
- 3.2.11 Localised wind corridors which facilitate prevailing winds coming from North-east and South-west directions:
 - Opening under the podium between Blocks 3 and 4 which allow wind penetration
 - Between Blocks 6 and 7 which aligned in NE to SW direction
 - Block 2 and Block 3 are taller than Block 1 by at least 25 meters to convey the upper wind down to the pedestrian level under prevailing E direction annual wind
- 3.2.12 Considering the wind performance in prevailing wind directions, the following issues are identified:
 - (a) The leeward side of the inner site area within HPR Site under annual and summer wind conditions;
 - (b) The extended podium to the south east of Blocks 2 and 11 of HPR Site is predicted to lower the performance of the potential wind path between Blocks 1 and 2;
 - (c) The leeward side of the inner site area within SHR Site Blocks 5 to 9 under annual and summer wind conditions;
 - (d) The extended podium to the south east of Block 9 is predicted to lower the performance of the potential wind path between Blocks 5 and 8 of SHR Site under NE to SW wind;
 - (e) Villa Pinada, the largest development in the surrounding area, would be affected in summer prevailing wind conditions subject to SSE and S winds. Its access road from the proposed Road L7 and the houses along the access would also be affected in annual prevailing wind conditions;

- (f) Proposed Primary School Site B (as shown in Appendix A) would be affected under summer wind conditions subject to SW and SSW winds;
- (g) Proposed Primary School Sites D & E (as shown in Appendix A) would be affected under annual wind conditions (SE & SSE wind) and summer wind conditions (SSE wind);
- (h) San Hing Tsuen would be affected under summer wind conditions(SW, SSW and S);
- (i) Air ventilation to discrete sections of the proposed Road L7 may be affected in annual prevailing conditions subject to NNE and NE winds; and
- (j) The downhill wind from Yuen Tau Shan would be less available in the development area as compare to the existing conditions.

3.3 Recommendations

- 3.3.1 In order to minimise the potential ventilation impacts, the conceptual layout of the proposed housing development has been carefully designed after a qualitative assessment of wind performance of the sites in consideration of the existing condition and the proposed indicative schemes.
- 3.3.2 It is recommended to maintain the building separation and localized wind corridors mentioned in 3.2.6 to 3.2.11 as to facilitate the wind penetration. Moreover, the extended podium structures mentioned in 3.2.12 are suggested to be removed to enhance the performance of the potential wind paths under different prevailing wind conditions. The proposed mitigation measures are shown in *Figure No.* **198172/B&V/FR/0301.**
- 3.3.3 It is also recommended to maximize the building block separations as far as practicable in the detailed design stage. Furthermore, quantitative study is suggested to quantify the air ventilation performance.
- 3.3.4 In addition, the following design principles to improve urban climate are specified for considerations at detailed design stage:
 - (a) Building permeability (in accordance with APP-152 Sustainable Building Design Guidelines requirement);
 - (b) Maintain podium free design as far as practicable;
 - (c) Building setback (in accordance with APP-152 Sustainable Building Design Guidelines requirement);
 - (d) Greenery target of 30% while 20% as a minimum;
 - (e) Avoid continuous long facades; and
 - (f) Reference could also be made to recommendations of design measures in the

Hong Kong Planning Standards and Guidelines.

- 3.3.5 This evaluation is qualitative assessment only. Further quantitative AVA is required at detailed design stage for design optimization, so as to assess the ability of wind penetration and to recommend design guidelines for future development from air ventilation perspective.
- 3.3.6 Initial Study (IS) by Computational Fluid Dynamics (CFD), is recommended to quantify the air ventilation performance. Should the proposed mitigation measures cannot be incorporated into the future design, the quantitative AVA is to prove that the future schemes perform no worse than those with the listed mitigation measures.
- 3.3.7 Mitigation measures, either planning measures (e.g. provision of amenity area/greening zone) or engineering measures e.g. (installation of shelter/canopy to alleviate downwashed winds) shall be investigated to address wind comfort at pedestrian level.

3.4 Summary

3.4.1 From the air ventilation point of view, no insurmountable issue is anticipated for the proposed development with the incorporation of good building design features.

4 PRELIMINARY TRAFFIC AND TRANSPORT IMPACT ASSESSMENT (TTIA)

4.1 General

4.1.1 This chapter provides a summary of the key findings in the Preliminary TTIA. The existing and planned road networks and traffic impacts due to the proposed developments are presented.

4.2 Existing Road Networks

- 4.2.1 The potential public housing sites are located at the north of Hong Po Road and the west of Castle Peak Road near Lam Tei Interchange.
- 4.2.2 Hong Po Road is currently a local access road connecting to the existing developments such as Villa Pinada. It connects to Tsing Lun Road and Ng Lau Road which further leads to Lam Tei Interchange.

4.3 Existing Traffic Conditions

4.3.1 An AOI was established in order to assess the traffic and transport impact arising from the proposed developments. Manual classified traffic count surveys at the concerned junction were conducted on a typical weekday in June 2018 during the AM (07:30 – 09:30) and PM (17:00 – 19:00) peak periods. The locations of these key junctions and road links within AOI are presented in *Figure Nos.* 198172/B&V/FR/0401 and 0402 respectively. The AM and PM peak periods were identified to be 08:00 – 09:00 and 17:30 – 18:30 respectively.

4.4 Approach for Traffic Forecast

<u>Overview</u>

- 4.4.1 In order to carry out the traffic forecast, the base year and future year traffic models for weekday AM & PM peaks are developed using the 2014–based Territorial Population Employment Data Matrices (TPEDM), 2015 Base District Traffic Model (BDTM), 2017 Annual Traffic Census (ATC) Report and latest planning assumptions for providing traffic forecast.
- 4.4.2 Taking into consideration the target population intake of SHR Site development (i.e. year 2031), SHR Site Extension development (i.e. year 2030) and HPR Site (i.e. year 2033). Year 2031 and 2036 are adopted for the traffic impact assessment of interim stage upon the population intake of SHR Site and SHR Site Extension, and 3 years after full intake of population respectively.

4.5 **Proposed Access Arrangements and Public Transport Facilities**

Preliminary Road Layout

Proposed Road L7

4.5.1 Proposed Road L7 is proposed to be connected to the realigned Hong Po Road which links Lam Tei Interchange and Ming Kum Road, providing a direct east-west transport corridor. The concept of the proposed Road L7 is to provide an alternative access route to the proposed development sites so as to minimize the traffic impact to the surrounding road network.

- 4.5.2 Without the proposed Road L7, all the traffic generated by the proposed developments to the south could only travel via Tuen Mun Road through Lam Tei Interchange or Tsing Lun Road. The proposed Road L7 could diversify the development traffic to/from the south to Ming Kum Road or Tsun Wen Road and hence alleviate the traffic impact to the road network.
- 4.5.3 The proposed Road L7 can serve as an alternative route to bypass Tuen Mun Road (Fu Tei Section), bringing traffic relief to a certain extent owing to the growing development in the NWNT.

Realigned Hong Po Road

4.5.4 Realignment and modification of Hong Po Road is proposed to align with the alignment of proposed Road L7. The realigned section of Hong Po Road will be a 10.3m wide two-way carriageway with footpath on both sides.

Vehicular Access Arrangement

- 4.5.5 The vehicular accesses for HPR Site are proposed at both southeast and southwest side of the site at Hong Po Road. Signalised junctions are proposed at the southeast side whereas run–in/out arrangement is proposed at the southwest side to connect the vehicular accesses with public road.
- 4.5.6 Three vehicular access points are proposed for SHR Site. An access road is proposed to bisect SHR Site and SHR Site Extension. The vehicular access of SHR Site is located at the west side of the site connecting to the access road, whereas the vehicular access of SHR Site Extension is located at the east side of the site connecting to the same access road. Similar to the vehicular access for HPR Site, it is proposed to provide a left turn filter lane on the exit arm of site access to channelise the left turning traffic for this junction. Connection to the major road network is expected through this proposed access road to Hong Po Road.
- 4.5.7 The vehicular access for School Sites next to SHR Site is proposed on the south side of SHR Site at Hong Po Road connecting to a cul-de-sac .
- 4.5.8 The remaining vehicular access for SHR Site is proposed at Ng Lau Road. It is located at the northern end of the site so as to maximise the sightline of Ng Lau Road eastbound to the vehicular access.
- 4.5.9 The details of the access arrangement and public transport facilities is shown in **Appendix B**.

Public Transport Facilities

4.5.10 For SHR Site, the PTI is proposed at the south of the SHR Site access by a one-way access road connecting to Tsing Lun Road. Four bus bays will be provided within the SHR PTI for the provision of future franchised buses. One Green Minibus (GMB) bay and one taxi bay are recommended to provide comprehensive coverage of the public transport services to accommodate the demand from SHR Site. The one-way traffic arrangement would allow bus traffic turn in/out of the terminus without any conflicting traffic movement.

4.5.11 The HPR PTI is proposed at the southern side of HPR Site with access at Hong Po Road. Six bus bays are proposed to be provided within the HPR PTI. One GMB bay and one taxi bay are recommended to provide comprehensive coverage of the public transport services to accommodate the demand from SHR Site Extension and HPR Site. To shorten the walking distance from HPR Site to the HPR PTI, ramp and stair are suggested to connect the HPR access road and the northern footpath of the PTI.

4.6 Operational Traffic Impact Assessment

4.6.1 Junction capacity and road link assessment for design year 2031 and 2036 for reference scenarios are carried out. The operational Traffic Impact Assessment identifies critical issues and recommends the associated traffic improvement schemes to alleviate the identified traffic problems as necessary.

Road Capacity Assessment

4.6.2 Road link capacity is assessed for the key road links within the AOI. The assessment results for both the reference and design scenarios of design years 2031 and 2036. The results are presented in *Table 4.1* and *Table 4.2* respectively.

			Link		V/C F	Ratio	
Index	Road Link	Direction	Capacity	2031 Re	ference	2031 Design	
			(pcu/hr)	AM	РМ	AM	PM
I 1	Yuen Long Highway (Lam Tei	EB	4000	0.93	0.91	0.93	0.91
	Interchange Section)	WB	4000	0.97	0.96	0.97	0.96
12	Tuen Mun Road (Lam Tei Interchange – Teing Tin Road	NB	6100	0.94	0.95	0.98	0.97
LZ	Section)	SB	6100	1.05	0.96	1.09	0.98
1.2	Tsing Lun Road (Section	EB	2800	0.70	0.59	0.80	0.64
L3	Tei Interchange)	WB	2800	0.51	0.48	0.69	0.57
I.A	Lam Tei Interchange (Entry Arm from Tuen Mun Road)	NB	3600	0.11	0.18	0.20	0.21
L4	Lam Tei Interchange (Exit Arm to Tuen Mun Road)	SB	1800	0.36	0.11	0.50	0.17
T.C.	Proposed Road L7 (Section bet	EB	900	N/A	N/A	0.29	0.28
L2	oad Site Access)	WB	900	N/A	N/A	0.14	0.14
	Tsing Tin Road (Section	EB	4800	0.42	0.30	0.42	0.30
L6	between Tsun Wen Road and Tuen Mun Road)	WB	4800	0.35	0.33	0.35	0.33
17	Lam Tei Interchange (Exit Arm to Castle Peak Road)	NB	2800	0.47	0.45	0.49	0.46
L7	Lam Tei Interchange (Entry Arm from Castle Peak Road)	SB	2800	0.41	0.38	0.43	0.39
	Tsing Lun Road (Section	NB	2800	0.60	0.72	0.62	0.72
L8	Hong Road)	SB	2800	0.55	0.49	0.58	0.49

Table 4.1 - Road Link Performance for Year 2031

Agreement No. CE 68/2017(CE) Site Formation and Infrastructural Works for the Development at San Hing Road and Hong Po Road, Tuen Mun – Feasibility Study

			Link	V/C Ratio				
Index	Road Link	Direction	Capacity	2031 Reference		2031 Design		
			(pcu/hr)	AM	РМ	AM	PM	
10	Castle Peak Road – Lam Tei	NB	2800	0.68	0.76	0.69	0.77	
L9 Cas		SB	2800	0.76	0.80	0.79	0.81	
110	Castle Deals Dead Linguan	NB	2800	0.15	0.20	0.23	0.25	
LIU	Casue Feak Road – Liligilali	SB	2800	0.29	0.29	0.29	0.29	

		Link		V/C Ratio				
Index	Road Link	Direction	Capacity	2036 Re	eference	2036 I	Design	
			(pcu/hr)	AM	РМ	AM	РМ	
L1	Yuen Long Highway (Lam Tei	EB	4000	0.98	1.03	0.98	1.03	
	Interchange Section)	WB	4000	0.98	1.05	0.98	1.05	
L2	Tuen Mun Road (Lam Tei Interchange – Tsing Tin Road	NB	6100	0.96	1.02	1.04	1.07	
	Section)	SB	6100	0.98	1.01	1.06	1.04	
L3	Tsing Lun Road (Section	EB	2800	0.76	0.61	0.95	0.69	
	Tei Interchange)	WB	2800	0.46	0.51	0.77	0.71	
L4	Lam Tei Interchange (Entry Arm from Tuen Mun Road)	NB	3600	0.07	0.16	0.23	0.25	
	Lam Tei Interchange (Exit Arm to Tuen Mun Road)	SB	1800	0.29	0.06	0.55	0.17	
L5	Proposed Road L7 (Section	EB	900	N/A	N/A	0.59	0.40	
	Po Road Site Access)	WB	900	N/A	N/A	0.46	0.27	
L6	Tsing Tin Road (Section	EB	4800	0.45	0.37	0.46	0.37	
	Tuen Mun Road)	WB	4800	0.39	0.38	0.40	0.39	
17	Lam Tei Interchange (Exit Arm to Castle Peak Road)	NB	2800	0.48	0.44	0.50	0.45	
L7	Lam Tei Interchange (Entry Arm from Castle Peak Road)	SB	2800	0.43	0.41	0.44	0.42	
1.0	Tsing Lun Road (Section	NB	2800	0.67	0.72	0.71	0.74	
LQ	Hong Road)	SB	2800	0.56	0.54	0.61	0.55	
19	Castle Peak Road – Lam Toi	NB	2800	0.71	0.75	0.73	0.76	
L7	Cashe Feak Road – Laili Tel	SB	2800	0.85	0.86	0.90	0.88	
110	Castle Peak Road - Lingnan	NB	2800	0.14	0.14	0.26	0.24	
L10	Casue i cak Roau – Lingildii	SB	2800	0.34	0.36	0.34	0.36	

Table 4.2 – Ro	ad Link Perform	nance for Year 2036

4.6.3 The assessment results show that all the above road links will be operated within capacity (i.e. V/C ratios below 1.0) in year 2031 and 2036 except Yuen Long Highway (Lam Tei Interchange Section) (L1) and Tuen Mun Road (Lam Tei Interchange -Tsing Tin Road Section) (L2), which have V/C ratio between 1.0 and 1.2.

4.6.4 For L1, the V/C ratio with and without the proposed development in year 2036 is the

same in this report. The cumulative traffic impact is mainly contributed by other planned development in the vicinity. With the implementation of other potential improvement works to the Yuen Long Highway proposed under other projects, it is anticipated that the overloading situation of this strategic route will be alleviated in the long-term. The traffic impact on this strategic road link due to the development should be timely reviewed in subsequent stages of the proposed development in conjunction with other traffic/transport infrastructure studies conducted by the Government.

4.6.5 For L2, the proposed development will slightly affect the performance of this strategic road link in year 2036. The cumulative traffic impact is also contributed by other planned developments in the vicinity, such as Tuen Mun Area 54 (TM54) Development, Hung Shui Kiu NDA, Yuen Long South Development, etc. The section of Tuen Mun Road between Lam Tei Interchange and Tsing Tin Road is bounded by the columns of the overhead Light Rail track, flyover and footbridge on both sides of the road which impose constraint to the widening of this road section. With the implementation of planned transport infrastructure projects such as Route 11 and Tuen Mun Bypass or alternatives, it is anticipated that the overloading of this strategic route will be alleviated in long-term. The traffic impact on this strategic road link due to the development should be timely reviewed in subsequent stages of the proposed development in conjunction with other traffic/ transport infrastructure studies conducted by the Government.

Critical Junction Assessment

4.6.6 Based on the forecasted peak hour traffic flows, the operational performance of the identified critical junctions have been assessed for scenarios with the proposed development under the design year 2031 & 2036. The results are summarized in *Table 4.3* & *Table 4.4* respectively.

		Junction	Reserve Capacity (RC)/ Ratio of Flow to Capacity (RFC)			
Index	Junction	Туре	2031 Reference		2031 Design	
			AM	РМ	AM	РМ
J1	Lam Tei Interchange / Castle Peak Road – Lingnan	Signal	5%	0%	-10%	-10%
J2	Lam Tei Interchange	Roundabout	0.58	0.44	0.73	0.51
J3	Tsing Lun Road / Ng Lau Road	Signal	45%	80%	40%	80%
J4	Tsing Lun Road / Hong Po Road	Roundabout	0.58	0.54	0.88	0.71
J5	Tsing Lun Road / Siu Hong Road	Signal	50%	65%	45%	65%
J6	Tsing Lun Road / Tsz Tin Road/ Access Road to Siu Hong Court	Signal	10%	20%	10%	20%
J7	Tsing Lun Road / Tsing Chung Koon Road	Signal	35%	20%	30%	20%
J8	Tsun Wen Road / Tsing Lun Road	Signal	50%	30%	45%	30%
J9	Tsing Tin Road / Tsun Wen Road	Signal	40%	55%	40%	55%
J10	Tin King Road / Ming Kum Road / Tsing Tin Road	Signal	45%	45%	40%	40%
J11	Tsun Wen Road / Leung Wan Street	Signal	45%	45%	40%	45%
J12	Hing Kwai Street / Tsun Wen Road / Ming Kum Road	Signal	40%	55%	30%	35%

 Table 4.3 - Junction Performace for Year 2031

Agreement No. CE 68/2017(CE) Site Formation and Infrastructural Works for the Development at San Hing Road and Hong Po Road, Tuen Mun – Feasibility Study

		Junction	Reserve Capacity (RC)/ Ratio of Flow to Capacity (RFC)			
Index	Junction	Туре	2031 Reference 2031 Des			Design
			AM	РМ	AM	РМ
J13	Proposed Road L7 / Hing Fu Street	Signal	45%	90%	25%	55%
J14	Proposed Road L7 / Road L54A	Signal	>100%	>100%	50%	50%
J15	Proposed Road L7 / Road L54D	Signal	>100%	65%	40%	40%
J16	Proposed Road L7 / Hong Po Road Site Access	Signal	n/a	n/a	n/a	n/a
J17	Proposed Road L7 / SHR Site Access Road	Signal	n/a	n/a	75%	>100%
J18	Tsing Lun Road Near Tuen Mun Hospital	Signal	20%	20%	20%	15%
J19	Tsing Lun Road Near Siu Hong Court	Signal	20%	20%	20%	20%
J20	Propsoed Road L7 / SHR Extension Site Access	Priority	n/a	n/a	0.68	0.39
J21	Tin King Road / Ming Kum Road	Signal	5%	0%	-10%	-10%

Note:

RC: Reserve Capacity for signalised junctions; RFC: Ratio of Flow to Capacity for priority junctions and roundabout. Figures for Reserve Capacity are rounded to nearest 5%.

		Junction	Reserve Capacity (RC)/ Ratio of Flow to Capacity (RFC)			
Index	Junction	Туре	2036 Re	eference	2036 Design	
			AM	РМ	AM	РМ
J1	Lam Tei Interchange / Castle Peak Road – Lingnan	Signal	-10%	-10%	-30%	-25%
J2	Lam Tei Interchange	Roundabout	0.52	0.44	0.80	0.62
J3	Tsing Lun Road / Ng Lau Road	Signal	55%	80%	15%	65%
J4	Tsing Lun Road / Hong Po Road	Roundabout	0.54	0.58	1.02	0.89
J5	Tsing Lun Road / Siu Hong Road	Signal	35%	65%	30%	60%
J6	Tsing Lun Road / Tsz Tin Road/ Access Road to Siu Hong Court	Signal	10%	15%	5%	10%
J7	Tsing Lun Road / Tsing Chung Koon Road	Signal	20%	15%	15%	15%
J8	Tsun Wen Road / Tsing Lun Road	Signal	30%	20%	25%	20%
J9	Tsing Tin Road / Tsun Wen Road	Signal	15%	40%	15%	40%
J10	Tin King Road / Ming Kum Road / Tsing Tin Road	Signal	50%	45%	45%	40%
J11	Tsun Wen Road / Leung Wan Street	Signal	20%	30%	15%	30%
J12	Hing Kwai Street / Tsun Wen Road / Ming Kum Road	Signal	30%	50%	5%	20%
J13	Proposed Road L7 / Hing Fu Street	Signal	35%	75%	5%	35%
J14	Proposed Road L7 / Road L54A	Signal	>100%	>100%	15%	25%
J15	Propsoed Road L7 / Road L54D	Signal	50%	40%	0%	15%
J16	Proposed Road L7 / HPR Site Access	Signal	n/a	n/a	30%	50%
J17	Propsoed Road L7 / SHR Site Access Road	Signal	n/a	n/a	25%	80%
J18	Tsing Lun Road Near Tuen Mun Hospital	Signal	25%	20%	20%	15%
J19	Tsing Lun Road Near Siu Hong Court	Signal	25%	25%	25%	20%
J20	Propsoed Road L7 / SHR Extension Site Access	Priority	n/a	n/a	0.87	0.45
J21	Tin King Road / Ming Kum Road	Signal	10%	-5%	-15%	-20%

Table 4.4 - Juncction Performance for Year 2036

Note:

RC: Reserve Capacity for signalised junctions; RFC: Ratio of Flow to Capacity for priority junctions and roundabout. Figures for Reserve Capacity are rounded to nearest 5%.

- 4.6.7 As shown in the above tables, it can be seen that all the assessed key junctions will be operating with ample reserve capacities, except for the junction of Lam Tei Interchange / Castle Peak Road Lingnan (J1) and Tin Kin Road / Ming Kum Road (J21) which would be overloaded in year 2031 with the proposed developments.
- 4.6.8 In year 2036, the assessed key junctions would generally operate with ample reserve capacity, except for the junction of Lam Tei Interchange / Castle Peak Road Lingnan (J1), Tsing Lun Road / Hong Po Road (J4), Proposed Road L7/ Road L54D (J15) and Tin Kin Road / Ming Kum Road (J21) would be overloaded with the proposed developments. The junction of Tsing Lun Road / Tsz Tin Road/ Siu Hong Court (J6), Tsun Wen Road / Ming Kum Road (J12) and Proposed Road L7 / Hing Fu Street (J13) would be operating close to or at capacity in year 2036.

Recommended Junction Improvement Measures

- 4.6.9 The Preliminary Traffic and Transport Impact Assessment has recommended the following junction improvement measures. The proposed improvement schemes are shown in **Appendix C**:
 - Lam Tei Interchange / Castle Peak Road Lingnan (J1);
 - In addition to the junction works at J1 under TM54 development, modification works of the junction layout is proposed to further alleviate the traffic condition of J1. It is proposed to convert the second farside lane of Castle Peak Road – Lingnan northbound from straight ahead movement to left turning movement to facilitate the increase of traffic due to the proposed developments. Besides, the left turning lane of Castle Peak Road – Lingnan southbound is proposed to split phase with the other movements.
 - Tsing Lun Road / Hong Po Road (J4); It is proposed to widen the the entry arm of Tsing Lun Road westbound carriageway to increase the capacity of this approach arm to ease the capacity of J4.
 - Tsing Lun Road / Tsz Tin Road / Access Road to Siu Hong Court (J6); To alleviate the traffic condition of J6, it is proposed to convert the farside lane of Tsing Lun Road southbound from right turning movement to straight-ahead and right turning movement. The northbound carriageway of Tsing Lun Road would be reduced to two lanes, where left turning cum straight-ahead and right turning cum straight-ahead movement would be provided for the nearside and farside lane respectively.
 - Hing Kwai Street/ Tsing Wen Road / Ming Kum Road (J12); To improve the junction improvement of J12, it is proposed to widen the carriageway of Hing Kwai Street southbound to provide an additional left turning lane. The corresponding section of eastern kerb of Hing Kwai Street will be setback.
 - Proposed Road L7 / Hing Fu Street (J13); To improve the junction improvement of J13, it is proposed to widen the carriageway of proposed Road L7 southbound to provide an additional traffic lane. The corresponding section of eastern and western kerb of proposed Road L7 will be setback. The nearside lane of Hing Fu Street eastbound is proposed to convert from left turning to left turning and right turning movement.

- Proposed Road L7 / Road L54D (J15); and It is proposed to widen the carriageway of proposed Road L7 southbound to provide an additional traffic lane. The corresponding section of eastern kerb of proposed Road L7 will be setback.
- Tin King Road / Ming Kum Road (J21). To ease the traffic condition of J21, it is proposed to convert the farside lane of Ming Kum Road southbound to right turning movement only to provide split phase for right turning and straight-ahead movement. The farside lane of Tin King Road eastbound is proposed to convert from right turning movement to left turning and right turning movement.
- 4.6.10 With the proposed junction improvement measures, the junction performance at junctions J1, J4, J6, J12, J13, J15 and J21 have been reassessed and the results are summarized in *Table 4.5* below. The junctions mentioned above will be operating within capacity in year 2031 and 2036.

			Reserve Capacity (RC)/ Ratio of Flow to Capacity (RFC)			
Index	Junction	Junction Type	2031 l (w improv	Design ith ement)	2036 (w improv	Design ith rement)
			AM	РМ	AM	РМ
J1	Lam Tei Interchange / Castle Peak Road – Lingnan	Signal	45%	35%	20%	20%
J4	Tsing Lun Road / Hong Po Road	Roundabout	0.75	0.61	0.88	0.76
J6	Tsing Lun Road / Tsz Tin Road/ Access Road to Siu Hong Court	Signal	20%	20%	15%	15%
J12	Hing Kwai Street / Tsun Wen Road / Ming Kum Road	Signal	50%	40%	25%	25%
J13	Proposed Road L7 / Hing Fu Street	Signal	45%	55%	15%	35%
J15	Proposed Road L7 / Road L54D	Signal	65%	65%	25%	40%
J21	Tin King Road / Ming Kum Road	Signal	20%	25%	20%	15%

Table 4.5 - Junction Performance with Improvement

4.7 Parking Provisions

4.7.1 Parking provisions for the development sites would be estimated according to the latest parking standards in Hong Kong Planning Standards and Guidelines (HKPSG). It is proposed to provide the upper bound of the parking provisions under the latest standards in the HKPSG as far as practicable.

4.8 Public Transport Assessment

- 4.8.1 The area adjacent to SHR site are well served by public transport facilities including MTR, LRT, franchised bus and GMB. However, SHR Site Extension and HPR Site are remote from the existing public transport network.
- 4.8.2 The public transport demand for the proposed developments is estimated based on the person trip rates surveyed at similar nearby developments and the modal split of the corresponding District Council Constituency Area making reference to the result of 2016 Population By-census. Based on the estimated public transport demand for the proposed developments, the feeder bus service to MTR Station, LRT service, bus

service and WRL service have been reviewed.

Review on Feeder Service to MTR Station

4.8.3 As the walking distance between SHR Site, SHR Site Extension and HPR Site is more than 500m from Siu Hong Station, it is suggested to provide feeder bus routes to Siu Hong Station PTI (North) and Tuen Mun Town Centre departing from HPR PTI and SHR Site bus layby. These routes are suggested to serve HPR Site, SHR Site and SHR Site Extension to and from Siu Hong Station and the town centre.

Review on LRT Service

4.8.4 It is estimated that approximately 4,885 passengers per hour at peak hour would be generated by the proposed public housing development to use the LRT services. It is suggested to provide feeder bus routes to Siu Hong Station PTI (North) departing from HPR PTI and SHR Site bus layby. These routes are suggested to serve HPR Site, SHR Site and SHR Site Extension to and from Siu Hong Light Rail Station.

Review on Bus Service

- 4.8.5 With the assumption that bus passengers of SHR Site would go to SHR PTI, the estimated peak bus passenger demand generated from SHR Site is 3,085. Considering that re-routing the existing bus routes to stop at SHR PTI during the peak hours would increase the bus travel time and have an adverse impact to the existing passengers, new bus routes providing regular bus services are recommended to provide direct services to Kowloon and Hong Kong Island which allow for connection to other districts via Tuen Mun Bus-Bus Interchange, subject to the decision of Transport Department and bus companies at later stage.
- 4.8.6 With the assumption that bus passengers of SHR Site Extension and HPR Site would go to HPR PTI, the estimated peak bus passenger demand generated from SHR Site Extension and HPR Site is 3,540. New regular / special terminating routes servicing HPR Site / SHR Site Extension to other districts via Tuen Mun Road Bus-bus Interchange are proposed, subject to the decision of Transport Department and bus companies at later stage.

Review on WRL Service

4.8.7 It is estimated that approximately 4,685 passengers per hour at peak hour would be generated by the proposed public housing developments to use the WRL services. The passenger demand from the proposed public housing developments would induce only a marginal inrease in the maximum hourly passenger flow during the AM peak at the WRL.

4.9 Summary

4.9.1 With the implementation of the proposed improvement works, it is anticipated the proposed development will not induce insurmountable problem to the traffic network from traffic point of view.

5 PRELIMINARY PEDESTRIAN CONNECTIVITY ASSESSMENT (PCA)

5.1 General

5.1.1 This chapter will assess the need and feasibility of providing continuous, barrier–free system in the form of at–grade or grade–separated pedestrian links for inter–connecting amongst the proposed public housing sites to the proposed PTIs and nearby public transport systems including WRL/ LRT/ Bus.

5.2 Existing Pedestrian Flows

- 5.2.1 A pedestrian count survey was carried out on a typical weekday in June 2018 during AM peak (07:30 09:30) and PM peak (17:00 19:00) to investigate the existing pedestrian flow pattern at the existing pedestrian network.
- 5.2.2 The observed AM peak period was identified between 08:00 and 09:00. Therefore, only the AM peak pedestrian flows are presented in this report to present the conservative results
- 5.2.3 According to the survey result, there are low pedestrian activities at the local streets within the study area.

5.3 Existing, Planned and Proposed Nearby Pedestrian Network

- 5.3.1 The following critical existing / planned / proposed footpaths / crossing will be assessed:
 - Existing San Hing Road Footpath;
 - Existing Siu Hong Road Footpath;
 - Existing Tsing Lun Road northern footpath near the Tsing Lun Road roundabout;
 - Existing Hong Po Road footpath section east of Villa Pinada;
 - Existing Hong Po Road footpath section west of Villa Pinada;
 - Existing Stairs and footbridge crossing Tsing Lun Road (hereinafter called as "FB1");
 - Proposed footbridge alongside the Tsing Lun Road Northern Footpath and connecting to the existing footbridge FB1 (hereinafter called as "**FB2**");
 - Proposed walkway connecting FB1 and the footbridge at Siu Hong Court near Siu Hong Court Bus Terminus (hereinafter called as "CW1");
 - Proposed realigned Hong Po Road between the HPR Site PTI and SHR Site PTI;
 - Proposed access road from HPR Residential Site to HPR PTI;
 - Proposed access road from HPR Residential Site to the realigned Hong Po Road;
 - Proposed access road from SHR Site Extension Residential Site to the realigned Hong Po Road;

- Proposed access road from SHR Site Extension Residential Site to SHR Residential Site;
- Proposed access road from SHR Residential Site to Siu Hong Station;
- Proposed access road leading from the realigned Hong Po Road to Tsing Lun Road;
- Proposed access road leading from Tsing Lun Road to Siu Hong Station; and
- Proposed access road leading from Siu Hong Road to Siu Hong Station.
- 5.3.2 In additional, there are several accesses proposed to maintain pedestrian linkage to existing land lots.
 - A. Proposed access to Tsz Tin Tsuen from the realigned Hong Po Road;
 - B. Proposed access to Villa Pinada from the realigned Hong Po Road; and
 - C. Proposed footpath from the existing village houses at the north of SHR Site Extension to the proposed access road to the school sites at SHR Site Extension.

5.4 Reference Pedestrian Flows

- 5.4.1 Under the latest planned road alignment in Siu Hong Road and Tsing Lun Road, the existing pedestrian network surrounding Siu Hong Court would be changed upon the population intake of TM54. The changes are extracted from the TM54 Review and are summarized in below:
 - (1) Junction of Tsing Lun Road / Siu Hong Road would become a signalized junction with crossing facilities at all arms of the junction;
 - (2) A new footpath alongside the LRT tracks and Siu Hong Court has been proposed by Housing Department; and
 - (3) A vertical mechanical transport system (including escalators and elevator) would be constructed and integrated with the existing stairs and footbridge connecting Siu Hong Road and Siu Hong Station PTI (North).
- 5.4.2 The 2036 reference pedestrian flows are derived based on the existing pedestrian pattern and pedestrian flows (factored up by TPEDM growth), and future pedestrian pattern and pedestrian flows estimated from planned adjacent development in the vicinity (e.g. TM54).

5.5 Pedestrian Flow Paths and Pedestrian Connectivity Assessment

5.5.1 The pedestrian demands in the peak hours are derived from the pedestrian flows generated by public transport passengers from SHR Site Extension, SHR Site and HPR Site, as well as the pedestrians from TM54 Sites 1&1A, onto the reference pedestrian flows.

5.5.2 Level of Service (LOS) is used to assess the performance of footpaths, stairs and footbridge. The description of the LOS obtained from the TPDM published by TD is summarized in *Table 5.1*.

	Flow	Rate	
LOS	(ped/m	in/m)	Description
	Footpath	Stairs	
A	<16	<16	Pedestrians basically move in the desired paths without altering their movements in response to other pedestrians. Walking speeds are freely selected, and conflicts between pedestrian are unlikely.
В	16-23	16-20	Sufficient space is provided for pedestrians to freely select their walking speeds, to bypass other pedestrians and to avoid crossing conflicts with others. At this level, pedestrians begin to be aware of other pedestrians and to respond to their presence in the selection of walking paths.
С	23-33	20-26	Sufficient space is available to select normal walking speeds and to bypass other pedestrians primarily in unidirectional stream. Where reverse direction or crossing movement exist, minor conflicts will occur, and speed and volume will be somewhat lower.
D	33-49	26-36	Freedom to select individual walking speeds and bypass other pedestrians is restricted. Where crossing or reverse-flow movements exist, the probability of conflicts is high and its avoidance requires changes of speeds and position. The LOS provides reasonable fluid flow; however considerable friction and interactions between pedestrians are likely to occur.
Е	49-75	36-49	Virtually, all pedestrians would have their normal walking speeds restricted. At the lower range of this LOS, forward movement is possible only by shuffling. Space is insufficient to pass over slower pedestrians. Cross- and reverse-movement are possible only with extreme difficulties. Design volumes approach the limit of walking capacity with resulting stoppages and interruptions to flow.
F	>75	>49	Walking speeds are severely restricted. Forward progress is made only by shuffling. There are frequent and unavoidable conflicts with other pedestrians. Cross-and-reverse movements are virtually impossible. Flow is sporadic and unstable. Space is more characteristics of queued pedestrians than of moving pedestrian streams.

Table 5.1 - Description of Pedestrian Facility LOS

Source: TPDM Volume 6 Chapter 10 – Pedestrian Action Plan

Pedestrian Network of HPR Site, SHR Site Extension and SHR Site

- 5.5.3 The 2-way pedestrian flow paths of HPR Site, SHR Site Extension and SHR Site are shown in *Figure Nos.* **198172**/*B*&*V*/*F***R**/**0501** and **0502**. The pedestrian flow paths (i.e. red arrows) simulate the different routes the pedestrians might take when accessing to the HPR PTI, SHR PTI, Lam Tei LRT Station or Siu Hong Station. It was also assumed some pedestrians would access between TM54 Site 1&1A and HPR PTI via proposed Road L7.
- 5.5.4 To mitigate the impact of the existing signalized crossing at junction of Tsing Lun Road/Ng Lau Road on the Tsing Lun Road roundabout and to alleviate the pedestrian condition of northern Siu Hong Road footpath, a footbridge system (FB2) with staircase, lifts and escalators is proposed to connect the open space next to the proposed SHR SPS at SHR Site to the existing footbridge near Ng Lau Road (FB1). A covered walkway (CW1) and modification of existing walkway, if necessary, are proposed to connect the existing footbridge near Ng Lau Road and the Siu Hong

Station. The proposed footbridge system will improve the pedestrian connectivity and its attractiveness would encourage the pedestrians to access to Siu Hong Station. The proposed footbridge system is shown in *Figure No.* **198172**/*B*&*V*/*FR*/**0502**.

5.5.5 The pedestrian flow paths and the corresponding LOS are summarized in *Table 5.2.* Based on the assessment results, all of the footpaths/ footbridge will achieve LOS(C) or above. Therefore, sufficient walking space will be available during peak hours and individual's walking speed and direction would not be constrained.

Footpath/ Footbridge	Location	Effective width (m) ⁽¹⁾⁽²⁾	LOS
P0	Proposed Road L7	3	А
P1a	Left footpath connecting HPR PTI and HPR Site	3.5	В
P1b	Right footpath connecting HPR PTI and HPR Site	3.5	В
Р2	HPR Site southern footpath connecting to realigned Hong Po Road	5	А
Р3	HPR northern footpath (between access road to HPR site and Villa Pinada)	3	В
P4	Realigned Hong Po Road northern footpath (between access road to Villa Pinada and SHR Site Extension)	3	В
Р5	Western footpath of the access road to SHR Site Extension connecting to the realigned Hong Po Road	5	В
P6	Northern footpath of the access road of SHR Site Extension connecting to the school sites	2.7	В
Ρ7	The outermost path of the access road of SHR Site Extension connecting to SHR Site	2	А
P8	Western footpath of SHR Site connecting to the realigned Hong Po Road	3	В
Р9	Realigned Hong Po Road northern footpath near SHR Site	3.5	А
P10	Realigned Hong Po Road next to SHR Site	4	А
P11	Footpath between the two sections of SHR Site connecting to the realigned Hong Po Road	4.5	В
P13	Footpath between the two sections of SHR Site connecting to the realigned Hong Po Road	4.5	В
P14	Footpath located in the north of the right primary school of SHR Site	3	В

Table 5.2 - Pedestrian Flow Paths and LOS Assessment

Agreement No. CE 68/2017(CE) Site Formation and Infrastructural Works for the Development at San Hing Road and Hong Po Road, Tuen Mun – Feasibility Study

Footpath/ Footbridge	Location	Effective width (m) ⁽¹⁾⁽²⁾	LOS
P15	Footpath located in the north direction of the right section of SHR Site	1.75	В
P16	Footpath inside the SHR Site leading to SHR PTI	4	А
P17	Footpath inside SHR Site within the proposed SPS boundary	3.5	В
P18	Western footpath of Tsing Lun Road (between Tsing Lun Road roundabout and Siu Hong Road)	1.75	А
P19	Northern footpath of Siu Hong Road	2	С
P20	Southern footpath of Siu Hong Road	2	В
P21	Northern footpath of Siu Hong Road	1.4	С
P22	Footpath between Siu Hong Court and LRT track	1.78	В
P25	Northern footpath of realigned Ng Lau Road	2	А
P26	Northern footpath of Ng Lau Road	3	А
P27	Northern footpath of Ng Lau Road	2	А
P28	Access road to Lam Tei LRT Stop eastern footpath (between Ng Lau Road and Lam Tei Stop)	2	С
FB1 ⁽³⁾	Existing footbridge near Ng Lau Road	3	С
FB2 ⁽³⁾	Proposed footbridge alongside Tsing Lun Road northern footpath connecting to existing FB1	3	А
CW1	Proposed covered walkway and modification of existing walkway, if necessary connecting the existing FB1 to the existing vertical mechanical transport system (including escalators and elevator) to Siu Hong Station	3	С

Notes:

1. Clear Width = horizontal clearance excluding the distance from kerb to railing (if any).

2. Effective Width = Clear width – dead width (0.5m at each side of walkway.)

3. The design width of the footbridge will be further discussed with relevant government department in the detail design stage. For assessing the proposed pedestrian facilities, a minimum clear width of 4m is adopted in the assessment as conservative. (The clear width of existing FB1 is 4m)

Other Recommendations

- 5.5.6 Pedestrian crossing points have been proposed at key road junctions and pedestrian desire routes to ensure safe and convenient pedestrian facilities are provided to serve the future community.
- 5.5.7 Pedestrian walkways and cycle tracks will be constructed along the proposed Road L7 and the realigned Hong Po Road to promote green commuting. The proposed cycle tracks will be connected to the nearby residential areas (e.g. Tuen Mun New Town) and public transport facilities (e.g. Lam Tei LRT Station and Siu Hong Station) where practicable. Upgrading/modification works to existing footbridge at Ng Lau Road and associated covered walkway are proposed to improve the walking environment and

promote green commuting. All these measures are proposed to facilitate walking and cycling across the PDA and in the wider community. This would also in turn help to reduce road-based traffic and hence their associated vehicular noise and pollutants emission.

- 5.5.8 Covered walkway is proposed for Footpaths P3, P4, P5, P8, P9, P10, P11, P13, P14, P16, P17 and CW1 as shown in *Figure Nos.* **198172/B&V/FR/0501** and **0502** to enhance the walking environment.
- 5.5.9 Proposed accesses will be provided to maintain pedestrian linkage to the existing houses in the vicinity of the proposed public housing developments including:
 - A. Proposed access to Tsz Tin Tsuen from the realigned Hong Po Road;
 - B. Proposed access to Villa Pinada from the realigned Hong Po Road; and
 - C. Proposed footpath from the existing village houses at the north of SHR Site Extension to the proposed access road to the school sites at SHR Site Extension.

5.6 Summary

5.6.1 With the implementation of the proposed pedestrian system and associated improvement works, it is anticipated that all of the footpaths/ footbridge under assessment would be operated within capacity. The proposed development will not induce insurmountable problem to the pedestrian network from traffic point of view.

6 PRELIMINARY GEOTECHNICAL ASSESSMENT (GA)

6.1 General

6.1.1 This chapter aims to summarize the geological and geotechnical information about the housing sites and their vicinity, identify potential geotechnical impacts that may arise from the proposed developments, and recommend the necessary improvements or upgrading works, if any.

6.2 Ground Condition

- 6.2.1 Based on the review of geological profiles, the details of the proposed sites are shown as follows:
- 6.2.2 HPR Site is underlain by a layer of fill (various from 0m to 3.1m thick), alluvium (various from 0m to 3.5m thick) or colluvium (various from 0.5m to 2.6m thick), completely decomposed Tuffite/ Tuff Breccia/ Tuff/ Siltstone/ Meta-Andesite (various from 6.6m to 21.3m thick), and highly decomposed Tuffite/Tuff (various from 0.25m to 1.0m thick). Bedrock is encountered at a depth varying from +37.8mPD to -14.6mPD.
- 6.2.3 SHR Site Extension is underlain by a layer of fill (various from 2.0m to 2.5m thick), alluvium (various from 0m to 2.0m thick), completely decomposed Tuff/ Siltstone/ Tuffaceous Siltstone/Tuff Breccia/Tuffite (various from 15.6m to 22.0m thick), and highly decomposed Rhyolite/ Tuff Breccia/ Tuff/ Tuffaceous Siltstone (various from 0m to 7.9m thick). Bedrock is encountered at a depth varying from -9.8mPD to -17.8mPD.
- 6.2.4 SHR Site is underlain by a layer of fill (various from 0m to 9.1m thick), alluvium (various from 1.5m to 6.2m thick), completely decomposed Tuffaceous Siltstone/ Siltstone/ Tuff/ Tuff Breccia/ Tuffite (various from 7.2m to 31.7m thick) and highly decomposed Rhyolite/ Tuff Breccia/ Tuff/ Tuffaceous Siltstone (various from 0.5m to 6.2m thick). Bedrock is encountered at a depth varying from -4.4mPD to -36.1mPD.
- 6.2.5 Proposed Road L7 is underlain by a layer of fill (various from 0m to 3.6m thick), alluvium (various from 1.5m to 3.6m thick) or colluvium (various from 1.5m to 5.1m thick), completely decomposed Meta-tuff/ Andesite/ Tuff/ Tuffite/ Tuff Breccia (various from 6.7m to 22.2m thick), and highly decomposed Andesite/ Meta-tuff/ Tuffite/ Tuff/ Tuffaaceous Siltstone/ Rhyolite (various from 0.5m to 5.8m thick). Bedrock is encountered at a depth varying from +15.6mPD to -19.5mPD.
- 6.2.6 The inferred geological profile of the proposed development is presented in *Figure Nos.* 198172/*B*&*V*/*FR*/601, 601a to d.

6.3 Groundwater Regime

6.3.1 Groundwater monitoring records for relevant existing vertical drillholes have been retrieved and inspected. The highest groundwater level in the vicinity of the HPR Site, the SHR Site, the SHR Site Extension, and along the alignment of the proposed Road L7 is found to be +19.61mPD (in drillhole 48524/DH1), +7.26mPD (in drillhole 42914/DH186), +9.26mPD (in drillhole 63271/DH01) and +27.47 mPD (in drillhole

32517/BH1) respectively. The corresponding highest groundwater level is 1.71m, 2.08m, 0.98m and 8.96m below the ground surface, respectively.

- 6.3.2 5 nos. of piezometers with Halcrow Buckets were installed at the 5 proposed drillholes for monitoring of groundwater table. Groundwater was encountered at the 5 drillholes. Response tests after piezometer installation and the subsequent groundwater monitoring for 7 consecutive days have been carried out from September 2019 to December 2019. The groundwater table is at a level approximately from +8.55 mPD to +27.80 mPD.
- 6.3.3 Groundwater shall be further monitored at standpipes or piezometers during the later design and construction stage. Groundwater may be encountered during construction of the proposed works. Shoring or properly designed excavation and lateral support system will be required to maintain stability of the trenches or working pits. Dewatering and working pits may also be necessary.

6.4 Geotechnical Consideration and Constraints

- 6.4.1 According to the existing GI record, the proposed sites are covered by 0m to 9.1m of fill layer and 0m to 6.2m of alluvium layer or 0m to 5.1m of colluvium. These superficial material are sandy in nature and should not be highly compressible material, it is considered feasible for supporting light duty retaining structures. No steep rock head depression are founded in existing borehole record, deep foundation for the proposed development is considered feasible.
- 6.4.2 However, given the borehole record for the Site is very limited, the site should be further characterized by more GI conducted in later stage of the Project. From the current available information, no adverse geological and geotechnical features are observed. The proposed works for the housing development is considered feasible.

6.5 Existing Man-made Slopes and Retaining Walls

6.5.1 17 registered man-made features are identified to be located within HPR Site, 3 registered man-made features are identified to be located within SHR Site, 2 registered man-made features are identified to be located within SHR Site Extension and 4 registered man-made features are identified to be located within the proposed Road L7. Failure of these slopes may affect the nearby infrastructure and future development. Therefore, further assessment is recommended. The details of the registered man-made features are summarized in *Table 6.1* below and shown in *Figure No. 198172/B&V/FR/0602.*

Tanatan	Geotechnical	Proposed to be Further
Location	Feature No.	Assessed/Demolished/ Modified
	6NW-C/R 71	To be modified
	6NW-C/R 97	Further assessment is required
SHK Site	6NW-C/FR 162	Demolished under site formation works of
		project CV/2011/01
CUD Site Extension	6NW-C/C 53	To be modified
SHR SILE EXTENSION	6NW-C/R49	To be modified
	5NE-D/C 7	To be modified
	5NE-D/C 97	To be modified
	5NE-D/C 103	To be modified
	5NE-D/C 104	To be demolished
	5NE-D/C 105	To be demolished
	5NE-D/F 3	To be modified
	5NE-D/F 25	To be modified
	5NE-D/R 5	To be demolished
HPR Site	5NE-D/R 6	To be demolished
	5NE-D/R 7	To be demolished
	5NE-D/R 9	To be demolished
	6NW-C/C 9	To be modified
	6NW-C/C 13	To be demolished
	6NW-C/C 68	To be modified
	6NW-C/C 69	To be modified
	6NW-C/F 8	To be modified
	6NW-C/F 9	To be modified
	5NE-D/C119	To be modified
Droposed Boad J 7	5NE-D/C 126	To be modified
FTOPOSEU KOAU L7	5NE-D/FR 34	To be modified
	5NE-D/R 11	To be modified

Table 6.1 – Existing Registered Man–Made Features Identified to be Potentially Affected by the Proposed Works

6.5.2 After a series of site visit by the use of existing access, it is found that there are 8 non-registered geotechnical features within/ in close proximity to the site boundary which will be potentially be affected by the proposed development. The locations of these non-registered geotechnical features are shown in *Figure No.* **198172/B&V/FR/0602.** Nevertheless, all the features are less than 1.5m high with the angle of elevation less than 15 degrees. *Table 6.2* summarises the non-registered geotechnical features that are identified to be potentially affected by the proposed works.

Table 6.2 – Existing Non-Registered Geotechnical Features Identified to be
Potentially Affected by the Proposed Works

r otominany milotica by morroposed morne				
Location	Geotechnical Feature No.	Proposed to be		
		Demolished/ Retained		
HPR Site	Slope A	To be demolished		
	Slope B	To be demolished		
	Slope C	To be demolished		
	Slope D	To be retained		
	Slope E	To be retained		
SHR Site	Slope F	To be demolished		
	Slope G	To be modified		
	Slope H	To be demolished		

6.5.3 To facilitate the future development, new features will be constructed within the proposed developments.

6.6 Natural Terrain Hazards Study (NTHS)

- 6.6.1 This NTHS is prepared to study, identify and evaluate all potential risks arising from natural terrain hazards including boulder fall hazards for the proposed developments and the infrastructure and assess the need for hazard mitigation measures in accordance with guidelines given in GEO Report No. 138 and other government guidance.
- 6.6.2 Screenings were carried out to determine the need for NTHS in accordance with "Alert Criteria" stated in GEO Report No. 138 (Second Edition, 2016). Based on the results of the screenings, no NTHS is required for SHR Site, SHR Site Extension, the northern and southwestern parts of HPR Site and proposed Road L7. The preliminary footprints of Catchment A, A1, B and C are shown in *Figure No.* **198172/B&V/FR/0603**. The results are listed in *Table 6.3* below:

Tuble old Bereening Rebuild for Natural Ferrain Hazarab hobebonient					
	Alert Cri	Need for NTHS			
Catchment(1)Angular elevation of natural terrain $\ge 20^{\circ}$		(2) Within 50 m of ground sloping at > 15 ⁰	[Satisfying (1) <u>AND (</u> 2) => Yes; otherwise => NO]		
А	Yes	Yes	Yes		
A1	No	Yes	No		
В	Yes	Yes	Yes		
С	Yes	Yes	Yes		

Table 6.3 - Screening Results for Natural Terrain Hazards Assessment

6.7 Preliminary Natural Terrain Hazard Assessment

Debris Mobility Modelling (DMM)

6.7.1 A total of 9 cases of DMM were carried out in this study to assess the mobility and the runout distance of Open Hillslope Landslide (OHL) and Channelized Debris Flow (CDF). A summary of the results are shown in *Table 6.4* below:

Catchmont	SectionCaseHazardRheologicalSourceVolume (m3)		Soction	irce ie (m³)	Maximum	Debris Runout	Debris		
Catchinent	Section	No.	Туре	Model	Initial	Final	(MS ⁻¹)	Distance (m)	Boundary
	А	A1	CDF	Voellmy	56.32	56.32	11.32	536.78	Yes
А	В	B1	CDF	Voellmy	55.67	55.67	12.34	471.30	Yes
	С	C1	CDF	Voellmy	56.30	56.30	19.79	372.14	Yes
	D	D1	OHL	Frictional	55.57	55.57	13.81	154.42	No
р	D	D2	OHL	Frictional	42.02	42.02	7.64	36.22	No
D	F	E1	CDF	Voellmy	55.40	55.40	16.12	144.30	No
	Ľ	E2	CDF	Voellmy	43.78	43.78	6.38	29.8	No
	F	F1	OHL	Frictional	42.43	42.43	12.90	72.30	No
L L	г	F2	OHL	Frictional	42.53	42.53	4.85	24.53	No

Table 6.4 - Summary of Debris Mobility Modelling

6.7.2 The results show that debris of the potential natural hazard of Catchment A would travel and reach the boundary of the proposed development. Debris at Catchment B and C could not reach the site boundary probably due to an abandoned quarry located at the lower terrain of Catchment B and C and acted as a platform.

6.8 Mitigation Strategy

6.8.1 Based on the revised Design Event Approach framework and requirements as outlined in GEO Report No. 138 (2nd Edition), The design requirements / level of hazard mitigation for the hillside catchments within the Study Area were determined based on the combination of catchment type and facility group of the at-risk facility. The finding is summarised in *Table 6.5*.

CATCHMENT	HILLSIDE CATCHMENT CLASSIFICATION	PRESENCE OF HAZARD?	HAZARD TYPE	FACILITY GROUP	MEET QUALIFYING CRITERIA (TGN 37)	LEVEL OF MITIGATION REQUIREMENTS
А	Channelized Depression	Yes	CDF	1	No	Level 2
Noto:						

Table 6.5 – Summary of Design Requirements

(1) Level 1 – Primary protection based on empirical provisions for OH catchments.

(2) Level 2 – Enhanced protection with enhanced measures designed by analysis.

- 6.8.2 Natural terrain hazard mitigation measure is therefore required for Catchment A. Active mitigation measure (Rigid barrier) along the slope toe within the site boundary is proposed to mitigate the potential channelized debris flow and boulder fall hazards of Catchment A.
- 6.8.3 In view of the above findings, a schematic layout of recommended mitigation strategy for CDF (Channelized debris flow), RF (Rock fall), and BF (Boulder fall) hazards has been proposed and presented in *Figure Nos.* 198172/B&V/FR/0604a and b.
- 6.8.4 The findings of the NTHS and the proposed mitigation measures should be reviewed in later stage of the project when more geological /geotechnical information is collected from detailed field mapping and ground investigation.
- 6.8.5 TSFR is located adjacent to the proposed development site and the proposed Natural Terrain Hazards Mitigation Measures. To avoid encroachment onto the TSFR, it is recommended to construct fencing and warning signs along the boundary between

firing range and the site.

6.9 Summary

6.9.1 In view of the preliminary findings from the geotechnical assessment, no insurmountable issue is anticipated for the proposed development in the geotechnical aspect.

7 PRELIMINARY SITE FORMATION ASSESSMENT (SFA)

7.1 General

- 7.1.1 This section is to recommend suitable formation level, profile and designs of the proposed sites and recommend the works required for the proposed sites.
- 7.1.2 The constraints for site formation of SHR Site ,SHR Site Extension, HPR Site and proposed Road L7 are discussed in the SFA Report. Different types of earth–retaining system are proposed to support and achieve the formation of the proposed sites. It is considered that the site formation and infrastructural works for the proposed development in this site are technically feasible.

7.2 Site Formation Design Consideration

Design Philosophy

- 7.2.1 A balanced cut and fill approach is adopted as far as practicable in designing the proposed site formation level. This could reduce the construction vehicular trips due to import or disposal of fill materials and minimize the possible dust emission during the trips.
- 7.2.2 The existing area of SHR Site has a gentle topography. The ground level of the site ranges from approximately +14.8mPD at northern side to +5.3mPD at southern side, and to +10mPD at eastern side. The Site is proposed to be formed as a single platform.
- 7.2.3 The existing area of SHR Site Extension has a gentle topography. The ground level of the site ascends from +9mPD at the southern corner to +16mPD at the northern corner. The Site is proposed to be formed as a single platform.
- 7.2.4 The existing area of HPR Site has an uneven topography. The ground level of the site ascends from +14mPD at southeastern corner to +40mPD at northwestern corner. In view of the significant level difference, the HPR Site is proposed to be formed in two stepped platforms.
- 7.2.5 A gentle gradient of 1% to 3% would be formed within the platforms to provide adequate fall for stormwater drainage. Temporary peripheral drains/ channels would be provided around the proposed site areas to collect surface runoff for discharge to the nearby drainage facilities.
- 7.2.6 The proposed road formation levels are designed to satisfy all road and traffic requirements. Also, the road formation levels take account of existing and planned utilities, including drains, sewers and cables etc., beneath the roads. To protect the underground installations, adequate cover (1 m minimum) to underground installations is allowed in designing the road formation level.

Existing Structures

7.2.7 Although all the existing buildings and structures within the proposed site areas will be dismantled during the site formation works, cluster of squatter structures, such as area to the east of HPR Site and area to the north of SHR Site Extension, will be retained. It is essential to ensure the flooding risk in the vicinity of the proposed sites would not be increased. Therefore, surface channels would also be installed along the site boundary to ensure the surface runoff in the vicinity would be properly drained to the downstream drainage facilities.

Existing Drainage Facilities

- 7.2.8 The Development would potentially cause the following impact to the surface runoff:
 - Change in land use from unpaved to paved surface would increase the amount of runoff entering into the drainage system;
 - Change in formation level and cross-fall within the Site would alter the overland flow pattern and discharge point into the drainage system resulting in modification of the existing catchment plan; and
 - Formation of the proposed development would cut off the flow of some existing streams.
- 7.2.9 The surface runoff of the Site is proposed be discharged to the existing drainage network adjacent to the Site. Necessary upgrading works will be proposed to cater the additional runoff arising from the change in land use.
- 7.2.10 It is essential to ensure the flooding risk in the vicinity of the Site would not be increased. Therefore, surface channels would also be installed along the site boundary to ensure the surface runoff in the vicinity would be properly drained to the downstream drainage facilities.
- 7.2.11 The potential drainage impact caused by the proposed Development and details of the proposed drainage system will be discussed in Section 8.

Existing Sewage Facilities

- 7.2.12 The sewage generated from SHR Site, SHR Site Extension and HPR Site will be conveyed to the proposed sewage pumping station at SHR Site via gravity sewers and subsequently discharging to the sewage pumping station in Tuen Mun Area 54 (TM54 SPS).
- 7.2.13 Since the proposed trunk sewer and the twin-cell 4,000m(W) x 2,000m(H) stormwater box culvert would both run along the proposed realigned Hong Po Road, the alignment of box culvert shall be taken into account during the design of the trunk sewer and vice versa.

Existing Utilities

- 7.2.14 The existing CLP's 400kV OHL and pylons are running along the southern boundary of SHR Site, SHR Site Extension and HPR Site. These facilities shall be taken into account during the planning of site formation and infrastructure works. The disturbance to the existing OHL and pylons shall be kept to minimal during construction and operation stages.
- 7.2.15 Close coordination with CLP shall be carried out during the construction stage and the code of practices published by relevant departments and CLP shall be strictly followed.

7.3 Proposed Site Formation Works

Proposed Site Formation Works at SHR Site

- 7.3.1 Subject to the gentle topographic nature of the eastern section of SHR Site, this section will be formed at the level between +6.0mPD to +7.6mPD by slope grading.
- 7.3.2 Considering the western part of the SHR Site which is generally flat, the site will be formed at the level between +7.5mPD to +12.0mPD by slope grading. The proposed works are presented in *Figure No.* **198172/B&V/FR/0701**.

Proposed Site Formation Works at SHR Site Extension

- 7.3.3 Subject to the gentle topographic nature of the SHR Site Extension, this section will be formed at the level between +9.8mPD to +12.2mPD by slope grading.
- 7.3.4 In SHR Site Extension, each school site will be formed with uniform level with slight slope grading. The proposed formation levels of the proposed secondary school, western primary school and eastern primary school are +12.2mPD, +12mPD and +10.5mPD respectively.
- 7.3.5 Regarding the proposed public access road serving the school sites, L–shaped retaining walls of about 3.0 to 4.9m retaining height are proposed along the northern boundary of the proposed road to preserve the existing road located at a higher elevation immediately to the north. The proposed works are presented in *Figure No.* **198172/B&V/FR/0701**.

Proposed Site Formation Works at HPR Site

- 7.3.6 Since the western part of the HPR Site is hilly and the eastern part is generally flat, the site will be formed in various platforms which rise diagonally from the southeast to the northwest direction. The uppermost platform will be formed at the level between +21.0mPD to +27.0mPD. The lowest platform will be formed at the level between +8.0mPD to +12.0mPD. An estate road at the level between +16.5mPD to +21.0mPD will be formed by retaining structures separating the two platforms.
- 7.3.7 Considering the access to the existing DSD adit shaft building, the gradient and connectivity of the proposed estate road within the housing development to proposed PTI and further to the realigned HPR road, retaining structures are proposed along the western boundary of HPR Site.
- 7.3.8 The construction of bored pile walls (approximately retained height ranging from 10m 15m) along the western boundary of HPR Site will serve as permanent retaining structures and will be completed before the commencement of excavation of site formation works.
- 7.3.9 After the bored pile wall is completed, excavation will be carried out in the site up to the specified site formation level. Open cut and temporary retaining structures will be used for excavation during the construction of permanent L-shaped retaining walls (approximately retained height ranging from 2m 10m) and gravity retaining walls (approximately retained height ranging from 0m 3m) within the site.
- 7.3.10 The proposed works within HPR Site are presented in *Figure No. 198172/B&V/FR/0701*.

Proposed Site Formation Works on Proposed Road L7

- 7.3.11 When determining the longitudinal profile of proposed Road L7, the profile of the realigned Hong Po Road and L54D was considered as the proposed Road L7 is proposed to be connected to the realigned Hong Po Road and Road L54D to form a looped road network. Furthermore, considering the road safety, the gradient of the proposed Road L7 shall comply with Transport Planning and Design Manual (TPDM). As such, it is unavoidable that the longitudinal profile of proposed Road L7 is generally lower than the existing ground profile.
- 7.3.12 The village zone, existing graves and existing pylons also impose a certain degree of constraint to the design of proposed Road L7. In order to overcome these constraints, different types of earth-retaining system are proposed along the northbound and southbound of the carriageway, including bored pile wall (approximate retained height ranging from 10m 15m), L-shaped retaining wall and cut slope with soil nail. The approximate retained height of the L-shaped retaining wall are recommended to be ranged from 2m 10m.

7.4 Earthwork Inventory

7.4.1 An estimation on the quantities of excavated materials and imported fill for each proposed site and the associated roadworks are shown in *Table 7.1* below:

Disease	Inert C&D	Inert C&D		Disposed Offsite (m ³)		Tentative
Phases	Material (m ³) Site (m ³)		Inert	Non– inert	Disposal Period	
SHR Site Extension (Residential Site)	8,300	1,900	8,300	0	1,900	2025 Q2 – 2026 Q4
SHR SPS	200	500	200	0	500	2026 Q4 – 2028 Q2
Realigned Hong Po Road	800	1,900	800	0	1,900	2026 Q4 - 2028 Q1
Proposed Road L7	72,500	3,500	68,000	4,500	3,500	2026 Q2 - 2028 Q3
SHR Site – Residential Site	71,900	6,400	61,050	10,850	6,400	2025 Q4 - 2028 Q1
HPR Site	308,700	9,200	117,800	190,900	9,200	2026 Q4 - 2029 Q3
School Site at SHR Site	3,700	2,400	3,700	0	2,400	2026 Q4 - 2028 Q3
School Site at SHR Site Extension	12,900	4,800	12,900	0	4,800	2026 Q4 - 2029 Q3
Infrastructure Works	70,650	7,850	43,700	26,950	7,850	2027 Q1 – 2030 Q3
Total	549,650	38,450	316,450	233,200	38,450	-

Table 7.1 - Summary of Estimated Quantities of C&D Materials GeneratedDuring Site Formation

- 7.4.2 Early consultation with EPD and Public Fill Committee of CEDD is required to identify the feasible sites for dumping the unsuitable material and imported fill material.
- 7.4.3 The proposed site formation works would be constructed in parallel with the associated infrastructure works for the development, including roadworks, drainage and sewerage works. Temporary drainage works with proper desilting facilities would be provided, as appropriate, during the construction to avoid possible water pollution and increase in flooding risk.
- 7.4.4 It is recommended to carry out a more detailed site formation / geotechnical assessment in the detailed design stage when the design parameters are confirmed.

7.5 Summary

7.5.1 According to the findings of the preliminary site formation assessment, no insurmountable issue is anticipated for the proposed site formation works of the proposed developments.

8 PRELIMINARY DRAINAGE IMPACT ASSESSMENT (DIA)

8.1 General

8.1.1 This chapter aims to assess potential drainage impacts that may arise from the proposed developments, and recommends the necessary improvement or upgrading works, if any.

8.2 Existing Drainage Condition

- 8.2.1 An unnamed stream course, hereby named as Chung Shan Creek [CSC54], conveys runoff from Chung Shan to the San Hing Nullah near Tsing Lun Road. HPR Site, SHR Site Extension and SHR Site are located in the south–east section of CSC54.
- 8.2.2 The contributing catchment of CSC54 has an area of approx. 1.87 km², which is located along Hong Po Road. The existing catchment plan of the study area is shown in *Figure No.* 198172/B&V/FR/0801.
- 8.2.3 A portion of the CSC54 in the south of SHR Site had been upgraded to a 7 m wide rectangular drainage channel under Contract No. DC/2001/05 Completion Works for Rural Planning and Improvement Strategy Minor Rural Improvement Works Package 4 Phase 1 TM–004 Chung Shan Drainage Improvement, Tuen Mun.
- 8.2.4 This 7 m wide rectangular drainage channel is connected to a downstream twin-cell 4000 x 2000 box culvert under Contract No. CV/2011/01 Site Formation and Infrastructural Works near Tsing Lun Road and Tsz Tin Road in TM54, and subsequently to a 4000 x 3000 twin-cell box culvert. The runoff will eventually be discharged to the two single-cell 4000 x 3000 box culverts for discharging into the San Hing Nullah, which is located upstream of the Tuen Mun Nullah.

8.3 Assessment Approach

- 8.3.1 The proposed development at HPR Site, SHR Site Extension and SHR Site would potentially have the following hydraulic impact to the downstream stormwater drainage system:
 - Change in land use from unpaved to paved surface would increase the amount of runoff entering into the stormwater drainage system.
 - Change in formation level and cross-fall in the Study Area would alter the overland flow pattern and discharge point into the stormwater drainage system
 - Formation of the proposed development would cut off the flow of the existing stream CSC 54 which might result in flooding at some low-lying area.
- 8.3.2 InfoWorks CS Version 11.0 has been adopted for assessing the potential drainage impact arising from the proposed development under this Assignment.
- 8.3.3 The design criteria for flood level depends on a combination of rainstorm event, tidal level, climate change, as well as the catchment characteristics. With reference to Table 11 of SDM, the determination of the flood level is provided in *Table 8.1* below:

Flood Level Return Period	Case a	Case b
50 years	50–year rain + 10–year sea level	10–year rain + 50–year sea level
200 years	200–year rain + 10–year sea	10–year rain + 200–year sea level
	level	

- 8.3.4 A new drainage system would be constructed to convey the runoff from the proposed developments and upstream catchments to Tuen Mun Nullah.
- 8.3.5 The proposed drainage pipes, manholes, catch pits, u–channels and box culverts throughout the proposed development area are shown in *Figure No.* **198172/B&V/FR/0802** and summarized as follows:
 - 2,000 mm (H) x 3,000 mm (W) box culvert along the eastern boundary of HPR Site;
 - 2,000 mm (H) x 3,000 mm (W) box culvert along the proposed access road by the northern and eastern sides of SHR Site Extension;
 - 2,000 mm (H) x 4,000 mm (W) box culvert along the realigned Hong Po Road between HPR Site and SHR Site;
 - Twin 2,000 mm (H) x 4,000 mm (W) box culverts along the realigned Hong Po Road between SHR Site to BC–5 under Contract No. CV/2011/01; and
 - Proposed drainage pipes and periperal U-channels to collect surface runoff connecting to the proposed major drains above.
- 8.3.6 U-channels would be installed for proper conveyance of the stormwater runoff to the adjacent downstream drainage system. Sand traps are suggested for installation at the inlet of the drainage pipe and box culvert to minimise blockage and maintenance requirement.
- 8.3.7 Drainage reserves are required for proper maintenance of the drainage pipes and box culvert to be handed over to DSD.
- 8.3.8 Based on the hydraulic assessment, there is no adverse drainage impact to the downstream drainage system arising from the proposed developments upon commissioning of the proposed drainage works.

8.4 Recommendation

- 8.4.1 The proposed drainage works is provided in *Figure No. 198172/B&V/FR/0802*. The proposed drainage layout is considered adequate in conveying the stormwater runoff to the downstream Tuen Mun Nullah. Interior peripheral U-channels would be installed to properly convey stormwater runoff from the Site to the downstream drainage system. The proposed U-channel catchment plan within the PDA is shown in *Figure No. 198172/B&V/FR/0803*. The proposed hydraulic network after development is shown in *Figure No. 198172/B&V/FR/0804*.
- 8.4.2 An existing stream, which is a tributary to CSC54, is likely to be impeded by the development of SHR Site Extension. A 2,000 mm (H) x 3,000 mm (W) box culvert would be constructed to divert the concerned stream to the proposed twin 2,000 mm (H) x 4,000 mm (W) box culvert along the realigned Hong Po Road. The downstream

of this diversion point would be backfilled during site formation for the development of SHR Site. The alignment of the diversion drains should be laid along the proposed access road for the schools and connected to the drainage system at the proposed roundabout located to the southeast of SHR Site. In addition, internal peripheral U– channels would be installed to convey the runoff from the development area to the downstream drainage system. Peripheral U–channels would also be installed inside the site boundary to collect runoff to avoid potential ponding due to the development of SHR Site Extension. The details of the abovementioned diversion is shown in *Figure No.* 198172/B&V/FR/0804.

8.4.3 Three existing streams, which are tributaries to CSC54, are likely to be impeded by the development of HPR Site. The two northern streams would be connected to the proposed 2,000mm (H) x 4,000 mm (W) box culvert at the realigned Hong Po Road via a proposed 2,000 mm (H) x 3,000 mm (W) box culvert. The western stream would connect to the proposed drainage pipe along the realigned Hong Po Road via the proposed drainage pipe along the maintenance access road along the south western side of HPR Site.

8.5 Summary

8.5.1 Following the implementation of the proposed drainage works, no insurmountable issue is anticipated for the proposed developments from drainage viewpoints.

9 PRELIMINARY SEWERAGE IMPACT ASSESSMENT (SIA)

9.1 General

9.1.1 This chapter aims to assess potential sewerage impacts that may arise from the proposed developments, and recommends the necessary improvement or upgrading works, if any.

Existing and Planned Sewerage System

- 9.1.2 Ongoing and planned sewerage projects in the Tuen Mun sewage catchment have been reviewed and the relevant parties liaised to establish the baseline sewerage conditions in the vicinity of the proposed developments.
- 9.1.3 The latest development parameters of the proposed SHR Site, SHR Site Extension and HPR Site, along with the latest information on the existing and planned sewerage network in Tuen Mun were obtained and reviewed.
- 9.1.4 The hydraulic model of the ultimate development scenario from the approved TM54 SIA Review under Agreement No. CE 55/2007(DS) was adopted.
- 9.1.5 The proposed SHR SPS would collect sewage from SHR Site, SHR Site Extension and HPR Site for conveyance to the downstream sewerage system.
- 9.1.6 Only minor sewerage impact to the downstream is anticipated. In particular, the sewers at the upstream of the TM54 SPS (Sections M15.1, M16.1, M17.1, M18.1 & M19.1) are proposed to be upgraded from DN750 to DN900 and the capacity of the TM54 SPS is proposed to be upgraded from 0.59 m³/s to 0.82 m³/s in order to handle the sewage generated from the northern Tuen Mun catchment including the planned development in TM54 and the updated Sites. The required/ anticipated completion time of the upgrading of TM54 SPS is proposed to be completed by September 2030.
- 9.1.7 Based on SIA assessment, it is concluded that both the TM54 and WIS SPS would have sufficient freeboard at the wet well to cater for the peak design flow event and no flooding occurs along the upstream and downstream sewerage of the SPSs. Hence, downstream of TM54 SPS (i.e. rising mains and gravity sewers etc.) are not required to be upgraded.
- 9.1.8 With the proposed upgrading of the TM54 SPS, the twin 500 diameter rising mains, downstream of TM54 SPS would operate with a peak velocity of 1.83 m/s which is considered adequate for sewage rising mains. Therefore, only minor modification of internal pipework within the SPS is anticipated. The current Environmental Permit (EP) of TM54 SPS comprises the construction and operation of a SPS with a capacity of 90,000 m³/day; hence, upgrading of the TM54 SPS would not require a variation of EP.
- 9.1.9 Hence, downstream of TM54 SPS (i.e. rising mains and gravity sewers etc.) are not required to be upgraded.
- 9.1.10 The Pillar Point Sewage Treatment Works (STW) at further downstream has adequate capacity to cater for the sewage flow to be generated from updated development in TM54 and the proposed SHR Site, SHR Extension Site and HPR Site.

9.2 Approach and Methodology

- 9.2.1 A sewerage hydraulic model submitted and approved by EPD in the TM54 SIA Review was adopted for this SIA with an objective to examining the potential sewerage impact arising from the proposed developments.
- 9.2.2 The adopted sewerage hydraulic model was originally developed on the sewage catchment of Tuen Mun under Agreement No. CE 55/2007 (DS) Village Sewerage in Tseung Kwan O and Tuen Mun (Ka Loon Tsuen and Lung Kwu Tan). Under CE 55/2007 (DS), various design scenarios were assessed, which included the years of 2008, 2016, 2021, 2031 and ultimate development scenario. The population estimation for this hydraulic model was based on the 2006-based Territory Population and Employment Data Matrix (TPEDM).
- 9.2.3 The hydraulic model was checked against the latest record plans, as-built and construction drawings, and information on SPSs obtained from DSD. The hydraulic model was re-run with the revised population and employment planned for the proposed development to assess the potential sewerage impact.
- 9.2.4 The sewage generated from the proposed developments will convey to the proposed TM 54 SPS for discharge to the WIS at Ming Kum Road and subsequently to the Pillar Point STW. New sewers are proposed to connect the proposed SHR SPS. A twin 450 diameter rising mains are proposed to connect the SHR SPS to TM 54 SPS. The proposed local sewerage network is shown in *Figure No. 198172/B&V/FR/0901*. Upon checking of the hydraulic performance as shown in *Table 9.1* below, the proposed local sewerage network has adequate handling capacity to convey the sewage flow from the SHR Site, SHR Site Extension and HPR Site to the proposed SHR SPS. The sewage flow would then be conveyed via the TM54 SPS to WIS.

Site Name	ADWF (m³/s)	Contributing Population	Peaking Factor for SPS	Peak Flow for SPS (m ³ /s)
		SHR SPS	· ·	
SHR Site	0.0636	20,349		
SHR Site Extension	0.01314	4,206		
HPR Site	0.07050	22,559		
Subtotal	0.147	47,115	3.04	0.45
Subtotal (with				
15%	0.169	54,182	3.01	0.51
contingency)				
		GRAVITY TO WIS		
Site 1 & 1A	0.027	8,617		
Site 3/4 (West)	0.046	14,719		
Site 5	0.0069	2,226		
		TM54 SPS		
Site 2	0.029	9,417		
Site 3/4 (East)	0.034	10,812		
Site 4A (South)	0.009	3,002		
Site 4A (East)	0.002	538		
Site 4A (West)	0.001	321		
Village	0.029	9,174		

Table 9.1 – Hyd	lraulic Performanc	e of SPSs
-----------------	--------------------	-----------

Site Name	ADWF (m³/s)	Contributing Population	Peaking Factor for SPS	Peak Flow for SPS (m ³ /s)			
	SHR SPS						
Sewerage							
Subtotal	0.104	33,265	3.00	0.31			
Total	0.273	<u>87,446</u>	2.92	<u>0.80</u>			

9.3 Estimated Sewage Flow

- 9.3.1 Local sewerage ranging from 300 mm to 750 mm diameter pipes would be designed to convey the sewage flow generated from the Sites to the SHR SPS for subsequently conveyance to the downstream TM54 SPS.
- 9.3.2 The associated rising mains to convey sewage from the SHR SPS to the downstream TM54 SPS are proposed along Hong Po Road and Tsing Lun Road. With the implementation of the proposed sewerage works and the proposed upgrading of TM54 SPS, no adverse sewerage impact is expected.
- 9.3.3 The proposed gravity sewers and twin rising mains for the sewage pumping station at SHR Site are shown in *Figure No.* **198172**/*B*&*V*/*F***R**/**0901**.
- 9.3.4 The average dry weather flow generated from by each phase of the development is shown in *Table 9.2.*

Site Name	Contributing Population	Unit Flow Factor (l/d)	Average DWF (m ³ /s)
	26,300 residents	190	0.05784
SHR Site	1,447 employees	280	0.00469
	2,304 students	40	0.00107
SHR Site Extension	4,700 residents	190	0.01034
	469 employees	280	0.00152
	2,784 students	40	0.00129
HPR Site	30,000 residents	190	0.06597
	1,319 employees	280	0.00427
	540 students	40	0.00025

Table 9.2 - Average Dry Weather Flow

9.4 Recommendation

- 9.4.1 The network of proposed local sewerage is provided in *Figure No.* **198172/B&V/FR/0901**. The proposed local sewerage network has adequate handling capacity to convey the sewage flow from the SHR Site, SHR Site Extension and HPR Site to the proposed SHR SPS. The sewage flow would then be conveyed via the TM54 SPS to WIS.
- 9.4.2 The design capacity of the pumping station of the WIS, TM54 and SHR Site are shown in *Table 9.3.*

Sewage Pumping Station	Maximum Sewage Flow SPS (m ³ /s)	Design Capacity (m ³ /s)	Upgraded Design Capacity (m³/s)/(m³/day)
SHR SPS	0.51	0.52	0.52/44,928
TM54 SPS	0.80	0.59	0.82/70,848
WIS SPS	1.24	1.52	-

Table 9.3 – Hydraulic Performance of SPSs

- 9.4.3 As shown in **Table 9.3**, the design capacity of TM54 SPS would need to be further upgraded. It is recommended to upgrade TM54 SPS from a design capacity of 0.59m³/s to approximately 0.82m³/s.
- 9.4.4 The design capacity of the proposed SHR SPS is proposed to be 0.52m³/s with 2 duty and 1 standby pump system.

9.5 Summary

9.5.1 Following the implementation of the proposed sewerage works, no insurmountable issue is anticipated for the proposed developments from sewerage viewpoints.
10 PRELIMINARY WATER SUPPLY AND UTILITY IMPACT ASSESSMENT (WSUIA)

10.1 General

- 10.1.1 This chapter aims to assess the potential water supply and utility impacts that may arise from the proposed developments, and recommends the necessary improvements or upgrading works, if any.
- 10.1.2 The capacity of the existing water supply facilities were assessed taken into account the additional water demand generated from the SHR Site, SHR Site Extension and HPR Site in the vicinity.

10.2 Existing Water Supply Systems

Fresh Water Supply

- 10.2.1 Tuen Mun Water Treatment Works (TMWTW) and Au Tau Water Treatment Works (ATWTW) are the source of fresh water supply to Tuen Mun North Fresh Water Service Reservoir (TMNFW S/R) and Tan Kwai Tsuen South Fresh Water Service Reservoir (TKTSFW S/R).
- 10.2.2 The fresh water supply for the existing area within the PDA is presently fed by TMNFW S/R via a network of trunk mains and distribution mains.

Salt Water Supply

10.2.3 Salt water supply for the existing area within the PDA are presently fed by Tuen Mun Salt Water Pumping Station (TMSW P/S) and Tuen Mun North Salt Water Service Reservoir (TMNSW S/R) via a network of trunk mains and distribution mains.

10.3 Water Demand of the Development

10.3.1 The estimated fresh water and salt water demands of SHR Site Extension, SHR Site and HPR Site are summarized in *Table 10.1*. The fresh and salt water demands have been calculated in accordance with DI No. 1309.

Site	Fresh Water Demand	Salt Water Demand
SHR Site Extension	1,328.6 m³/day	412.1 m ³ /day
SHR Site	7,063.4 m³/day	1,934.9 m³/day
HPR Site	7,992.3 m³/day	2,142.3 m ³ /day

Table 10.1 -	Summary	of Estimated	Water Deman	d
	Summary	of Lotinateu	mater Deman	u

10.4 Impact to Proposed Developments on Water Supply

Fresh Water Supply

- 10.4.1 According to the Planning Report No. 3/2010 prepared by WSD, the TMNFW S/R has a storage capacity of 60,564 m³. It is anticipated that the TMNFW S/R will not have sufficient capacity to cater for the total forecasted fresh water demand of the proposed housing developments generated from this Project (SHR Site, SHR Site Extension and HPR Site), TM54, San Hing Tsuen, and along the Tuen Mun/ Yuen Long corridor and the whole north-western Tuen Mun. Therefore, WSD proposed that the water supply to Tuen Mun/ Yuen Long corridor and part of north-western Tuen Mun are to be taken up by the existing TKTSFW S/R.
- 10.4.2 After the implementation of shifting the fresh water supply zones, the TMNFW S/R would have adequate capacity to cope with the increase in fresh water demand arising from the proposed housing development in Tuen Mun Area 29 (Area 29), TM 54 and San Hing Tsuen.
- 10.4.3 According to a recent review with Water Supplies Department (WSD), it is advised that TMNFW S/R and TKTSFW S/R would not have spare capacity for providing fresh water supply for the whole proposed developments. Alternative water supply schemes were considered, and details are discussed in *Section 10.5*.

Salt Water Supply

- 10.4.4 Based on the Planning Report No. 3/2010 prepared by WSD, after the commissioning of LOPSW P/S and other associated water works, the existing TMNSW S/R and TMSW P/S would have adequate capacities to cope with the ultimate salt water demand of Tuen Mun including the increase in salt water demand arising from the proposed housing developments at Area 29, TM54 and San Hing Tsuen.
- 10.4.5 According to a recent review with Water Supplies Department (WSD), it is advised that TMNSW S/R would not have spare capacity for providing salt water supply for the whole proposed developments. Alternative salt water supply schemes were considered, and details are discussed in *Section 10.5*.
- 10.4.6 A capacity assessment of the existing fresh and salt water service reservoirs was conducted to estimate the spare capacity in order to identify the appropriate water supply arrangement for the proposed developments. The summary of the capacity assessment are shown on *Table 10.2* & *Table 10.3* for fresh water and salt water supply.

Site	Source of FW Supply	Formation Level (mPD)	Residual Head at Connection Point (m) ¹	Invert Level (mPD)	Top Water Level (mPD)	Design / Spare Capacity (m ³)	Required Capacity (m ³)
SHR Site Extension	• TMNFW S/R	10.0	31.1	82.3	89.7	3,753	1,062.9
SHR Site	• Existing/ Potentially Relocated TKTSFW S/R in Lam Tei and adjoining areas under WSD's cavern project	6.0	39.6	60.0	67.0	-2	5,650.7
HPR Site	• HPRFW S/R	8.0	22.2	32.0	38.5	6,916	6,793.5

Table 10.2 - Summary of Fresh Water Supply Schemes

Notes:

¹ a minimum residual pressure of 20-metre head should be provided for fresh water supply

² Subject to the latest design of the relocated TKTSFW S/R under WSD's cavern project

Site	Source of SW Supply	Formation Level (mPD)	Residual Head at Connection Point (m) *	Invert Level (mPD)	Top Water Level (mPD)	Design / Spare Capacity (m ³)	Required Capacity (m ³)
SHR Site Extension		10.0	21.9				
SHR Site	• HPRSW S/R	6.0	23.8	32.0	38.5	1,300	1,122.4
HPR Site		8.0	26.1				

Table 10.3 - Summary of Salt Water Supply Schemes

Note:

*a minimum residual pressure of 15-metre head should be provided for salt water supply

10.4.7 The total storage capacity required to meet the additional water demand of the developments is estimated to be 75% of the fresh water mean daily demand (MDD) for service reservoir of interconnected supply zone, 85% of fresh water MDD for service reservoir of isolated supply zone and 25% of the salt water MDD. An 5% of Fresh Water MDD shall be added for supply zone with critical consumers. 6,600m³/d for 8 hours is required fire–fighting water supply for the housing sites.

10.5 Recommendations for Fresh Water Supply

10.5.1 The proposed fresh watermains are shown in *Figure Nos.* **198172/B&V/FR/1001** to **1003** *a*, *b* **&** *c*. The details of the proposed fresh watermains scheme 1 are discussed as below:

SHR Site Extension – Existing TMNFW S/R

10.5.2 The fresh water supply to SHR Site Extension would be sourced from the TMNFW S/R via the proposed fresh water mains along the realigned Hong Po Road. The layout of the fresh water mains to cater for the fresh water demand to SHR Site Extension is shown in *Figure Nos.* 198172/B&V/FR/1001 to 1003 a, b & c.

<u>SHR Site – Existing/Potentially Relocated TKTSFW S/R into Strategic cavern areas in</u> <u>Lam Tei</u>

10.5.3 The fresh water supply to SHR Site is proposed to be sourced from the existing TKTSFW S/R or potentially relocated TKTSFW S/R in strategic cavern areas (SCVA) which is planned to accommodate the existing and proposed service reservoirs in Lam Tei and adjoining area. The layout of the fresh water mains to cater for the fresh water demand to SHR Site is shown in *Figure Nos.* 198172/B&V/FR/1001 to 1003 *a*, *b* & *c*.

<u>HPR Site – HPRFW S/R</u>

- 10.5.4 Originally the shifting of supply zone from TMNFW S/R to TKTSFW S/R, TMNFW S/R would have sufficient spare storage capacity to meet the fresh water demand arisen from HPR Site. However, after reviewing the latest fresh water demand of the new/ planned developments in north-western Tuen Mun, TMNFW S/R would be fully committed to other current development.
- 10.5.5 A new HPR Fresh Water Service Reservoir (HPRFW S/R) with an approximate capacity of 6,916 m³ is proposed to cater for the future fresh water demand at HPR Site. The fresh water mains would be constructed along the access road to DSD's adit shaft building and the realigned Hong Po Road. The HPRFW S/R would be connected to the TMNFW S/R via fresh water mains along Ming Kum Road and proposed Road L7. The layout of the fresh water mains to cater for the fresh water demand to HPR Site is shown in *Figure Nos.* 198172/B&V/FR/1001 to 1003 a, b & c.
- 10.5.6 An alternative arrangement of constructing the proposed HPRFW S/R at the northwestern side of HPR Site with an approximate capacity of 13,955 m³ to cater for the future fresh water demand arising from all three Sites have been explored and would be further developed at later stage if programme mismatch of fresh water supply from future TKTSFW S/R is found.
- 10.5.7 A review of the fresh water supply arrangement would be carried out in later stage.

10.6 Recommendations for Salt Water Supply

10.6.1 The proposed salt watermains are shown in *Figure Nos.* **198172**/*B*&*V*/*FR*/**1004** *to* **1005** *a* & *b*. The details of the proposed salt watermains has been illustrated below:

SHR Site Extension

- 10.6.2 The salt water supply to SHR Site, SHR Site Extension and HPR Site are proposed to be served by the TMSW P/S and TMNSW S/R. However, the results of the capacity assessments showed that the TMNSW S/R would not have enough capacity to cater for the salt water demand of SHR Site Extension, SHR Site and HPR Site. As such, the TMSW P/S is proposed to be the source of salt water supply to SHR Site Extension.
- 10.6.3 A new HPR Salt Water Service Reservoir (HPRSW S/R), with a capacity of approximately 1,300 m³, is proposed at the north-western side of HPR Site to provide salt water and serve as a buffer for maintaining salt water supply. The HPRSW S/R would be connected with the TMSW P/S via the salt water mains on proposed Road L7. Salt water supply to SHR Site Extension would be transported via the salt water mains along the access roads to DSD's adit shaft building and the realigned Hong Po Road. The layout of HPRSW S/R and the proposed salt water mains to cater for the salt water supply to SHR Site Extension are shown in *Figure Nos.* **198172/B&V/FR/1004** to **1005** a **&** b.

SHR Site

10.6.4 Similarly, the salt water supply to SHR Site would be sourced from the TMSW P/S, which would be connected with the HPRSW S/R via the salt water mains on proposed Road L7. Salt water supply to SHR Site would be transported via the salt water mains along the access roads to DSD's adit shaft building and the realigned Hong Po Road. The layout of the salt water mains to SHR Site is shown in *Figure Nos.* **198172/B&V/FR/1004** to **1005** *a* **&** *b*.

HPR Site

10.6.5 Similarly, the salt water supply to HPR Site would be sourced from the TMSW P/S, which would be connected with the HPRSW S/R via the salt water mains on proposed Road L7. Salt water supply to HPR Site would be transported via the salt water mains along the access roads to DSD's adit shaft building and the realigned Hong Po Road. The layout of the salt water mains to HPR Site is shown in *Figure Nos.* **198172/B&V/FR/1004** to **1005** *a* **&** *b*.

10.7 Existing Utilities

10.7.1 Based on the information provided by relevant Government departments and utility undertakers, there are limited existing utilities in the development boundary. It is considered that the existing utility service is not sufficient to support the future population in the Development. A summary of the preliminary review on the existing utilities is provided in *Table 10.4*.

Tuble 10.1 Keview on Existing Othices			
Government Department/ Utility Undertaker	Existing UU in the vicinity of the Sites (Yes/No)	Adequacy of Existing UU to support the Development (Yes/No)	
DSD	Yes	No	
WSD	Yes	No	
TD	Yes	No	
HyD, Lighting	Yes	No	

Table 10.4 – Review on Existing Utilities

Government Department/ Utility Undertaker	Existing UU in the vicinity of the Sites (Yes/No)	Adequacy of Existing UU to support the Development (Yes/No)
EMSD	Yes	No
CLP	Yes	No
Wharf T&T Ltd.	Yes	No
New World Telephone Company Ltd	Yes	No
HKCGC	Yes	No
НКТ	Yes	No
HKBN	Yes	No

10.8 Findings and Recommendations on Utilities

10.8.1 Major utility undertakers, including Hong Kong & China Gas Co. Ltd., CLP Power Hong Kong Ltd., Wharf T&T Ltd., Hong Kong Cable Television Ltd., Hong Kong Telecommunications Ltd. and Hong Kong Broadband Network Ltd., have preliminarily confirmed that they could provide their respective services to support the Development. Detailed provision of utilities would be further considered during the later detailed design stage. Nevertheless, no major constraints for re-provision of existing utilities or laying of new utilities are anticipated. *Table 10.5* summarises the responses from the major utility undertakers.

Utility Undertaker	Provision of Service for the Development (Yes/No)	Remarks
CLP	Yes	The alignment of the facilities could be altered in the future to meet the requirements.
Wharf T&T Ltd.	Yes	The Development would be served by the cable fitted from the Tuen Mun Hub. The proposed duct system inside the Development Area would be constructed in conjunction with new road construction stage.
New World Telephone Company Ltd	TBD	No response was received.
HKCGC	Yes	Medium pressure and low pressure gas pipes along the proposed roads and gas governor kiosks would be installed along the proposed roads at suitable locations within the

Table 10.5 – Review on Proposed Utilities

Utility Undertaker	Provision of Service for the Development (Yes/No)	Remarks
		Development Area.
НКТ	Yes	Underground plant would be constructed along Hong Po Road to serve the Development.
HKBN	Yes	Underground duct route (2x107 mm UPVC ducts) could be installed for the Development.
Hutchison Global Communication Ltd.	TBD	No response was received.

- 10.8.2 With the extension of the existing networks by the service providers (in public areas) and HKHA (within the Site), it is concluded that the utility support to the development is technically feasible.
- 10.8.3 It is noted that the provision of utilities from some utility undertakers, such as telecommunication service providers, could not be confirmed at this stage of the Study. Further liaisons and coordination with relevant Government departments and utility undertakers should be carried out during the later detailed design stage to ensure that sufficient utility services would be provided before the population intakes of the developments.

10.9 Summary

10.9.1 From water supply and utilities viewpoints, no insurmountable issue is anticipated for the proposed development.

11 PRELIMINARY SUSTAINABILITY ASSESSMENT (SA)

11.1 General

- 11.1.1 The objective of the Preliminary SA are listed as follows:
 - a. Use the Computer–Aided Sustainability Evaluation Tool (CASET) as evaluation frameworks to evaluate and assess the infrastructure;
 - b. Devise and modify the set of guiding principles, parameters and evaluation criteria to assess the sustainability implications;
 - c. Conduct preliminary assessment on possible environmental impacts during construction and operational stages;
 - d. Conduct preliminary assessment on the implications to the local community as well as to Hong Kong at large;
 - e. Evaluate and analyse the sustainability implications; and
 - f. Identify the key sustainability issues.
- 11.1.2 A summary of various aspects of the review are presented in the section below:

11.2 Approaches for Sustainability Assessment

- 11.2.1 Computer–Aided Sustainability Evaluation Tool (CASET) was used to evaluate the sustainability implications of the infrastructure proposed under the Project.
- 11.2.2 A set of guiding principles from CASET is listed as follows:
 - Economy
 - Health and Hygiene
 - Natural Resources
 - Society and Social Infrastructure
 - Biodiversity
 - Leisure and Cultural Vibrancy
 - Environmental Quality; and
 - Mobility
 - A list of characterisation includes:
 - Art / Culture / Recreation / Entertainment
 - Conservation, Environment and Agriculture
 - Demographics
 - Economics
 - Education
 - Energy
 - Health and Living Conditions

- Housing
- Industry
- Land and Infrastructure
- Transport
- Waste and Waste Water

11.3 Scenario

- 11.3.1 Considerations are compared between the "with development" and "without development" scenario. For instance, the "without development" scenario represents the baseline condition of the parameter prior to the implementation of the Project while the "with development" scenario would include the Project.
- 11.3.2 The "with development" scenario is defined as the proposed public housing and schools developments and their associated infrastructure works (site formation, drainage and sewerage infrastructural works, road works, environmental mitigation measures, water supply and utilities services works) at HPR Site, SHR Site and SHR Site Extension.
- 11.3.3 Majority of the HPR Site and SHR Site is situated in the areas currently zoned as "Green Belt (GB)" and/or "Residential (Group E)", in which re–zoning is required prior to implementation of the proposed public housing development.
- 11.3.4 The site formation works include slope cutting, earth filling works as well as geotechnical works/structures. The proposed infrastructures, i.e. roadworks, drainage (drains and box culvert), water supply (service reservoirs and water mains), sewerage (gravity sewers, rising mains and pumping station) etc. will be constructed to cater for the increase in future demand due to population intake in 2030, 2031 and 2033 for SHR Site Extension, SHR Site and HPR Site respectively.
- 11.3.5 Environmental mitigation measures, such as noise barrier, would be proposed if necessary, to minimise environmental and public disturbance.
- 11.3.6 The CASET parameters have been assessed qualitatively and quantitatively where possible. The results are summarized in *Table 11.1* below:

Parameters	Qualitative Changes	Expected Range of Change in Conditions
Carbon Dioxide Emitted Per Year	Decrease	Very Small Deterioration
Construction Waste	Decrease	Very Small Deterioration
Cost-benefit	Increase	Small Improvement
Criteria Air Pollutants	Decrease	Small Deterioration
Eco-value Habitats	Decrease	Small Deterioration
Energy Consumption	Decrease	Very Small Deterioration
Excessive Noise	Decrease	Small Deterioration
Fixed Capital	Increase	Small Improvement
Freight Costs	Increase	Small Improvement
Freshwater Supplied and Consumed	Decrease	Small Deterioration
Income Differential	Remain	-
Job Creation/ Loss	Increase	Very Small Improvement
Landfill Capacity	Decrease	Very Small Deterioration
Local Freshwater	Remain	_
Municipal Solid Waste	Decrease	Very Small Deterioration
Open Space Shortfall	Increase	Small Improvement
Protected and Managed Habitats	Decrease	Small Deterioration
River Water Quality	Remain	_
Significant Landscape Features (Area)	Decrease	Small Deterioration
Significant Landscape Features (point)	Decrease	Small Deterioration
Travel Distance	Remain	_
Travel Speed	Remain	-

Table 11.1 – Summary of Anticipated Impacts for each Sustainability Parameter

- 11.3.7 The sustainability assessment at this stage indicates that with the implementation of the Project, the main benefits would be improvement in housing and living conditions, economy, leisure and society and social infrastructure. These benefits come in the form of improved health and well-being of residents, positive economic return, and enhancement to social involvement.
- 11.3.8 The associated transport infrastructure would improve the connectivity between the development sites and the periphery regions to the rest of Hong Kong. On the other hand, negative impacts come mainly in the form of environmental degradation, with deterioration in natural resources.
- 11.3.9 Overall, the benefits produced by the Project outweigh the residual negative impacts especially considering the long term. Therefore, with the implementation of the proposed mitigation measures to minimize the negative impacts, the Project is considered sustainable.

12 PRELIMINARY LAND REQUIREMENT STUDY (LRS)

12.1 General

12.1.1 This chapter aims to summarize the key land requirement issues due to the proposed developments.

12.2 Land Resumption

12.2.1 The Site contains a total area of 29.7 ha and about 12.8 ha belongs to Government land (43%). The area of the affected private land within the Development Area is approximately 16.9 hectare. The detailed breakdown of the area of the affected land is summarized in *Table 12.1* below:

Site	Approx. Area of Land Lots Affected (m ²)
HPR Site	41,510
SHR Site Extension (residential)	17,670
SHR Site Extension (school sites)	14,440
SHR Site (residential)	72,120
SHR Site (school sites)	12,050
Proposed Road L7	570
Realigned Hong Po Road	10,510
Total	168,870

Table 12.1 - Total Area of Private Land Lots to be Affected

12.2.2 There are approximately 249 licenses structure expected to be affected by the proposed developments. The number of affected licenses structure at each site are shown in *Table 12.2* below:

Table 12.2 - Existing Licenses to be Anected		
Site	Number of affected licenses	
SHR Site (residential)	107	
SHR Site (school sites)	14	
SHR Site Extension(residential)	17	
SHR Site Extension (school sites)	45	
HPR Site	60	
Proposed Road L7	3	
Realigned Hong Po Road	3	
Total	249	

Table 12.2 – Existing Licenses to be Affected

12.2.3 There are in total about 384 private land lots needed to be resumed for the proposed developments. The number of affected private land lots are shown in *Table 12.3* below:

Site	Number of affected land lots
SHR Site	161
SHR Site Extension	75
HPR Site	112
Proposed Road L7	4
Realigned Hong Po Road	32
Total	384

Table 12.3 – Existing Private Land Lots to be Affected

12.3 Graves, Kam Taps (Urns) or Shrines

12.3.1 The existing graves and urns that will be potentially affected by the proposed developments are summarized in *Table 12.4* below:

Sito	Number of
Site	Graves/ Urns Affected
SHR Site	0
SHR Extension	2
HPR Site	5
Proposed Pood L7	26 (8 were inaccessible and 7
Proposed Road L7	could not be located)

12.4 Affected Houses

12.4.1 Existing houses within the land resumption boundary have been identified based on examination of aerial photos and observation from site visits in February 2019. The estimated total number of affected houses is 83.

12.5 Affected Agricultural Lands and Orchards

12.5.1 Continuous pieces of agricultural lands and orchards are identified within the land resumption boundary. These agricultural areas penetrate through the village and different type of business areas. The approximate area of the affected agricultural lands and orchards is summarized in *Table 12.5* below:

				Agricultural Lands and Orchards			
Site	Site Area (m²)	Gov. Land Area (m²)	Private Land Area (m²)	Area Affected on Gov. Land (m²) (Approx. % of Land Affected)	Area Affected on Private Land (m²) (Approx. % of Land Affected)		
HPR Site	87,950	46,450	41,510	31,430 (67.60%)	5,330 (12.80%)		
SHR Site Extension (residential)	23,640	5,970	17,670	3,930 (65.70%)	9,000 (51.00%)		
SHR Site Extension (school sites)	24,010	9,570	14,440	9,180 (95.90%)	11,600 (80.00%)		
SHR Site (residential)	91,360	19,230	72,120	5,050 (26.30%)	16,650 (23.10%)		
SHR Site (school sites)	14,170	2,120	12,050	0 (0.00%)	0 (0.00%)		
Proposed Road L7	18,190	17,630	570	13,390 (76.00%)	140 (24.20%)		
Realigned Hong Po Road	23,180	12,670	10,510	3,030 (23.90%)	1,910 (18.20%)		

 Table 12.5 - The Area of Affected Agricultural Lands and Orchards

				Agricultural Lands and Orchards				
Site	Site Area (m²)	Gov. Land Area (m²)	Private Land Area (m²)	Area Affected on Gov. Land (m²) (Approx. % of Land Affected)	Area Affected on Private Land (m²) (Approx. % of Land Affected)			
Total	282,500	113,640	168,870	66,010 (58.00%)	44,630 (26.40%)			

12.5.2 The land requirements are estimated based on the preliminary design and will be subject to review during the detailed design stage. It will be revised and updated, if necessary, in accordance with the latest development of the Project.

12.6 Summary

12.6.1 A preliminary land requirement study has been conducted to identify the land requirements of the proposed site formation and infrastructural works. It is anticipated that there would be no insurmountable issue from the land acquisition viewpoint.

13 ENVIRONMENTAL IMPACT ASSESSMENT (EIA)

13.1 General

- 13.1.1 EIA have been conducted as part of the feasibility study to assess the overall acceptability of any adverse environmental consequences that are likely to arise because of the proposed developments.
- 13.1.2 The EIA report (Development at San Hing Road and Hong Po Road, Tuen Mun (Application No. EIA-263/2020)) has been approved with a condition under Section 8(3) of the EIA Ordinance (Register No. : AEIAR 227/2020).

13.2 Environmental Impact Assessment Ordinance (EIAO)

- 13.2.1 The total study area under the Study is about 29.7ha. Since the total study area exceeds 20 ha,an Environmental Impact Assessment (EIA) is required in accordance with the Item 1 under Schedule 3 of the EIA Ordinance (EIAO) (Cap. 499), i.e. "Engineering feasibility study of urban development projects with a study area covering more than 20 ha or involving a total population of more than 100,000".
- 13.2.2 For any future change(s) after the approval of this EIA report, the following steps will be taken:

(a) Should the change(s) involve a designated project item under Schedule 2 of the EIAO, the requirements under the EIAO will be complied with; and

(b) Should the change(s) not involve any designated project items under Schedule 2 of the EIAO, prevailing planning mechanisms and standards will be followed and relevant EIA findings will be conformed to.

13.2.3 To implement the Project, there is one proposed work classified as DP under Schedule 2 of the EIAO. The details are shown in and *Table 13.1*.

Item Ref. No.	Ref. Category No.	Descriptions of DP under EIAO	Works Component
DP1	F.3 (b) of Part 1 Schedule 2	A Sewage Pumping Station with an installed capacity of more than 2,000 m ³ per day and a boundary of which is less than 150m from an existing or planned residential area	Construction of a sewage pumping station (SPS) with a design capacity of 14,629 m ³ /day at SHR Site and is less than 150m from an existing / planned residential area

Table 13.1 - Summary of Schedule 2 Designated Project

13.3 Air Quality Impact

Key Assessment Scope and Key Criteria

- 13.3.1 The air quality impact assessment was conducted in accordance with the requirements in Annexes 4 and 12 of the Technical Memorandum on EIA Process (EIAO-TM) and the requirements in Section 3.4.3 and Appendix B of the EIA Study Brief. The assessment area for the air quality impact is defined by a distance of 500m from the boundary of the Project Site during construction phase and 500m from the boundary of the PDA and road junction improvement works during operation phase.
- 13.3.2 Quantitative assessments using the relevant air models approved by EPD have been conducted for both construction and operation phases. Cumulative air quality impact has also been determined in the impact assessments.

Construction Phase

13.3.3 Potential construction dust impact would be generated from site clearance, land decontamination, site formation works and roadworks. Quantitative fugitive dust assessments have been conducted. The major concurrent projects include Site 4A (South), Site 4A (East) and Site 5 of the TM54 Development and Area 29 West Development. With the implementation of mitigation measures specified in the Air Pollution Control (Construction Dust) Regulation together with the recommended dust suppression measures including watering once per hour on active works areas, exposed areas and haul roads and other site management measures such as, good site practices, and environmental monitoring and audit (EM&A) programme, the predicted Total Suspended Solid (TSP), Respiratory Suspended Solid (RSP) and Fine Suspended Solid (FSP) at representative air sensitive receivers (ASRs) would comply with the criteria stipulated in the Air Quality Objectives (AQOs) and EIAO-TM. The predicted concentrations for key representative pollutants after implementation of mitigation measures are summarised in *Table 13.2*.

	TSP	R	SP	FS	Compliance	
	1-hr	24-hr (10 th Annual highest)		24-hr (10 th highest)	Annual	compnance
Existing ASRs	202 - 434	83 - 98	36 - 46	62 - 68	25 - 27	Yes
AQOs/ EIAO-TM Criteria	500	100	50	75	35	

Table 13.2 - Summary of Predicted Cumulative Construction Dust Imp	act
(After Implementation of Mitigation Measures)	

Operation Phase

13.3.4 Key existing, planned and committed air pollution sources during operation phase are the vehicular emission from open sections of existing roads, proposed roads and proposed junction improvement works within the assessment area. Cumulative air quality impact at the representative ASRs would also be expected due to the background pollutant concentrations, portal emissions from the noise enclosure at Tsing Lun Road and industrial emissions from the identified chimneys within the assessment area. Key representative air pollutants include Sulphur Dioxide (SO₂), Nitrogen Dioxide (NO₂), RSP and FSP. 13.3.5 Quantitative assessment has been conducted and the assessment results concluded that the predicted cumulative air quality impacts in Year 2029 and Year 2033 on all ASRs would comply with the AQOs. Results in Year 2029 and Year 2033 are summarised in *Table 13.3* and *Table 13.4* respectively. Therefore, no adverse air quality impact during operation phase is anticipated.

	Pollutant Concentration (µg/m ³)								
	N	02	R	SP	FSP		SO ₂		
	1-hr (19 th	Annual	24-hr (10 th	Annual	24-hr (10 th	Annual	10-min (4 th	24-hr (4 th	Compliance
	highest)		highest)		highest)		highest)	highest)	
Existing	104 -	23.7 -	82 - 89	35.6 -	62 - 67	25.3 -	n/2	n/2	Ves
ASRs	139	37.4	02-07	37.6	02 - 07	26.7	11/ a	II/a	105
Planned	104 -	24.3 -	02 06	36.0 -	67 64	25.6 -	122 -	20 21	Voc
ASRs	128	36.9	03 - 00	36.8	02 - 04	26.3	145	20-31	Tes
AQOs/ EIAO-TM	200	40	100	50	75	35	500	125	
Criteria									

Fable 13.3 – Summary of Predicted Concentrations of Representative Air
Pollutants during Operation Phase in Year 2029

Table 13.4 - Summary of Predicted Concentrations of Representative AirPollutants during Operation Phase in Year 2033

	Pollutant Concentration (µg/m ³)								
	N	02	R	SP	FSP		SO 2		
	1-hr (19 th highest)	Annual	24-hr (10 th highest)	Annual	24-hr (10 th highest)	Annual	10-min (4 th highest)	24-hr (4 th highest)	Compliance
Existing ASRs	104 - 128	23.6 - 32.6	82 - 89	35.6 - 37.6	62 - 67	25.3 – 26.7	n/a	n/a	Yes
Planned ASRs	105 - 125	24.6 - 33.2	83 - 85	36.0 - 36.5	62 - 64	25.6 - 26.1	122 - 145	28 - 31	Yes
AQOs/ EIAO-TM Criteria	200	40	100	50	75	35	500	125	

13.3.6 Potential odour impact arising from the operation of the proposed SHR SPS and future refuse collection points (RCPs) has been assessed in this EIA. With the installation of deodourising units with 99.5% odour removal efficiency and incorporation of appropriate design measures at the proposed SHR SPS, and proper ventilation and deodorizing design system installed at the future RCPs together with implementation of good housekeeping, it is anticipated that there would be no adverse odour impact from the proposed SHR SPS and future RCPs at the nearby ASRs.

13.4 Noise Impact

Key Assessment Scope and Key Criteria

13.4.1 The noise impact assessment was conducted in accordance with the requirements set out under Annexes 5 and 13 of the EIAO-TM, and Section 3.4.4 and Appendix C of the EIA Study Brief. The assessment area for noise impact is generally defined by a distance of 300m from the Project Site (i.e. the PDA boundary and the associated infrastructure works).

Construction Phase

Construction Noise

- 13.4.2 Construction noise associated with the use of powered mechanical equipment (PME) for various stages of construction has been assessed. With the implementation of practical mitigation measures including good site management practices, use of Quality Powered Mechanical Equipment (QPME), use of movable noise barrier, noise enclosure and noise insulating fabric and provision of minimum separations from the affected educational institutions or avoidance of any noisy construction activities during the school examination period, the predicted construction noise impact would be 58-75 dB(A) for residential noise sensitive receivers (NSRs), 50-70 dB(A) for education institutions and 50-65 dB(A) for education institutions during school examination period. Hence, no unacceptable impact arising from the construction of the Project would be anticipated.
- 13.4.3 For conducting construction works close to education institution, it is recommended that the Contractor shall liaise with the school representative(s) to obtain the examination schedule so as to avoid noisy construction activities during school examination period.

Operation Phase

Road Traffic Noise

13.4.4 Operation road traffic noise impact on the representative existing and planned noise sensitive uses within and near the PDA and the junction improvement works have been predicted. To mitigate the road traffic noise impact on the existing and planned NSRs within and near the PDA and one of the junctions, a combination of noise mitigation measures has been recommended, including i) application of low noise road surfacing material along some sections of Project roads and other roads; ii) construction of roadside noise barriers along some sections of project roads; iii) provision of acoustic windows for the proposed public housing buildings; iv) construction of boundary wall for the proposed welfare facilities and v) implementation of on-site noise mitigation measures at Site 5 of TM54 by Hong Kong Housing Authority. Restriction on locating the more noise sensitive welfare uses at façade facing the realigned Hong Po Road and access road in HPR Site has also been recommended. With all the proposed mitigation measures in place, the façade noise levels at all the planned NSRs would comply with the respective noise criteria. A summary of the predicted road traffic noise levels with mitigation measures in place is given in *Table 13.5*.

	Summary of Miligated Road Trame Noise Devels					
Uso	Predicted Mitigated Overall Noise Levels,	Criteria, L10(1hr)				
036	L _{10(1hr)} dB(A)	dB(A)				
Residential	41 - 70	70				
Educational Institutions	57 - 65	65				

Table 13.5 - Summary of Mitigated Road Traffic Noise Levels

13.4.5 With the implementation of the proposed mitigation measures, the traffic noise level from Project roads would comply with the respective noise criteria and the Project road contribution to overall noise level of existing residential and educational NSRs outside PDA would be negligible. The details of the proposed noise mitigation

measures is shown in *Figure Nos.* 198172/B&V/FR/1301a and 1301b.

Fixed Noise Source

- 13.4.6 A number of facilities have been recommended to support the operation of the proposed developments. Some of these facilities, which include the proposed SHR SPS and the two PTIs are fixed noise sources that would have potential noise impacts on NSRs. Other than these proposed fixed noise sources, there are existing fixed noise sources including rural workshops and storage sites.
- 13.4.7 Fixed noise source impact assessment has been conducted for all existing and planned NSRs. Noise impact from proposed fixed plant could be effectively mitigated by implementing noise mitigation measure at source. With the adoption of the proposed maximum permissible Sound Power Levels (SWLs) for the ventilation fans of proposed PTIs, the impact noise levels at the representative NSRs would comply with the relevant noise criteria. Therefore, adverse impact on the NSRs due to fixed noise sources is not anticipated.

Rail Noise

13.4.8 Rail noise impact assessment has been conducted for the planned NSRs at the SHR Site which are potentially affected by the WRL and LRT operation. With the implementation of mitigation measure in the form of provision of acoustic windows for the proposed public housing, the impact noise levels at all representative NSRs would comply with the relevant rail noise criteria. Therefore, adverse rail noise impact on the planned NSRs is not anticipated.

13.5 Water Quality Impact

Key Assessment Scope and Key Criteria

13.5.1 The water quality impact assessment was conducted in accordance with the requirements set out under Annexes 6 and 14 of the EIAO-TM, and Section 3.4.5 and Appendix D of the EIA Study Brief. The assessment area of the water quality impact includes the area within 500m from the boundary of the Project Site.

Construction Phase

13.5.2 Water quality impacts from the construction works are associated with the general construction activities, construction site run-off, accidental spillage, and sewage effluent from construction workforce. The site practices as outlined in the ProPECCPN 1/94 "Construction Site Drainage" and the ETWB TC (W) No. 5/2005 "Protection of natural streams/rivers from adverse impacts arising from construction works" are recommended to minimise the potential water quality impacts from the construction activities. Proper site management and good site practices are also recommended to ensure that construction wastes and other construction-related materials would not enter the nearby streams. Temporary sanitary facilities would be provided on construction sites to properly collect the on-site sewage generated from the construction workers. Water quality monitoring will be implemented and regular site inspection will be conducted during construction stage to ensure that the recommended mitigation measures are properly implemented.

13.5.3 With the implementation of the recommended mitigation measures, the construction works for the Project would not result in unacceptable impacts on water quality.

Operation Phase

- 13.5.4 During the operation phase, all the sewage and wastewater generated from the PDA will be properly collected and discharged to the public sewerage system and conveyed to the Pillar Point Sewage Treatment Works (PPSTW) for chemically enhanced primary treatment (CEPT) prior to discharge into Urmston Road. Discharge from the PPSTW after CEPT and UV disinfection would not cause adverse water quality impact in North Western and Western Buffer waters.
- 13.5.5 In view of the potential emergency discharge from the proposed SHR SPS, contingency measures such as a storage tank, twin rising mains, standby pumps and dual electricity supply or backup power supply facilities would be provided at the proposed SHR SPS to minimize the risk of failure of the SPS leading to emergency discharge of untreated sewage. An emergency response plan will be formulated to minimize the impact of emergency discharges and facilitate subsequent management of the emergency. With proper implementation of the recommended mitigation measures, no unacceptable water quality impact would be expected during the operation phase of the Project. No adverse residual water quality impact is anticipated during operation phase of the Project.

13.6 Sewerage and Sewage Treatment Implications

Key Assessment Scope and Key Criteria

13.6.1 Sewerage and sewage treatment implications of the Project were evaluated and assessed in accordance with Section 3.4.6 and Appendix E of the EIA Study Brief and the criteria as stated in Annex 14 of the EIAO-TM.

Operation Phase

- 13.6.2 The proposed development in the PDA will generate large amount of sewage flow. New public sewerage system and the proposed SHR SPS will be constructed to collect sewage generated from the PDA and convey the collected sewage to the downstream sewerage system. According to the assessment, the PPSTW would have adequate treatment capacity to cater for the sewage flow generated from the PDA.
- 13.6.3 Based on the findings of this sewerage and sewage treatment impact assessment, the Project is sustainable from sewage collection, treatment and disposal perspective.

13.7 Ecological Impact

Key Assessment Scope and Key Criteria

13.7.1 The ecological impact assessment was conducted in accordance with the requirements set out under Annexes 8 and 16 of the EIAO-TM, Section 3.4.9 and Appendix H of the EIA Study Brief, EIAO Guidance Notes (6/2010, 7/2010 and 10/2010) and other relevant legislations and guidelines. The assessment area of the ecological impact includes the area within 500m from the boundary of the Project Site.

Potential Impacts and Mitigation Measures

- 13.7.2 Ecological baseline was established by both literature review and 8 months programme of field surveys covering both dry and wet seasons. A total of 11 types of habitat were identified within the ecological assessment area, of which 8 types of habitats, including woodland, agricultural land, natural stream, semi-natural stream, drainage channel, village/orchard, village area and urban area were located within the Project Site where the development area and associated infrastructure works are proposed.
- 13.7.3 Avoidance measures were considered and incorporated during early stages of the Project. Direct encroachment onto the two identified sites of conservation interest within the assessment area, namely a mature secondary woodland (W2) near Tong Hang Road and the Conservation Area (CA) zone near Chung Shan has been avoided by the Project. An upstream section of the semi-natural stream (R1f) at HPR Site is proposed to be retained by adjusting the proposed development boundary for preservation of the localities of two crab species of conservation interest (*Cryptopotamon anacoluthon* and *Somanniathelphusa zanklon*) (See Section 8.7.7 of EIA Report).
- 13.7.4 Most habitats to be lost, such as agricultural land, drainage channel, village area, village/orchard and urban area are of low ecological value. Although the whole of the woodland (W3) is rated as of medium ecological value, the affected portion is a low-quality habitat patch, and regularly disturbed by grave-sweeping activities. Hence, the potential impact due to loss of woodland is minor. The potential impacts due to loss of habitats are all anticipated to be minor (See Section 8.7.39 of EIA Report). A summary of the estimated habitat loss is given in *Table 13.6*.

Habitat	Ecological Value	Permanent Loss (ha)
Woodland (W3) *	Medium	1.2
Agricultural Land	Low	1.1
Natural Stream (of Water Network N2) ^	Low	0.0 (193m ² ; 104m in total length)
Semi-natural Stream	Medium-low to	0.1 (356m in total length)
(of Water Network N1)	Low	
Drainage Channel	Low	0.4 (1,160m in total length)
Village / Orchard	Low	8.0
Village Area	Negligible	17.4
Urban Area	Negligible	0.4

Note:

* Seven woodlands were identified in the ecological baseline, viz, W1-W7. Only a small part of W3 will be in unavoidable conflict with the development. All others W1-2, W4-7 will not be affected.

^ Natural streams of higher ecological value were all located in water network N1 which will not be affected by the proposed development.

13.7.5 Consideration of avoiding the impacts on floral species of conservation interest, viz. *Aquilaria sinensis* and *Pyrenaria spectabilis* has been taken into account during the selection of the alignment and construction method of the proposed Road L7 at Po Tong Ha. With careful designation of the alignment and the adoption of suitable construction method for the retaining structures along the proposed Road L7, all the *Pyrenaria spectabilis* and majority of *Aquilaria sinensis* can be preserved in-situ. Only

three individuals of *Aquilaria sinensis* were found falling within the footprint of the proposed Road L7. To minimize the impact on the three individuals of *Aquilaria sinensis*, two of them will be preserved by transplanting while the remaining one will be felled down due to its poor health condition. With implementation of proposed mitigation measure, the impact on *Aquilaria sinensis* is insubstantial.

- Majority of the recorded localities of the two crab species of conservation interest 13.7.6 (viz. Cryptopotamon anacoluthon and Somanniathelphusa zanklon) are identified at the upper section of the semi-natural stream R1f, marginally located along the boundary of HPR Site. To avoid direct impact on these two crab species, the ecologically sensitive upper section of the semi-natural stream R1f located in HPR Site is proposed to be retained by excluding it from the project site area. After this active avoidance measure, only one locality of Cryptopotamon anacoluthon will be unavoidably affected. Translocation of these crabs at suitable undisturbed stream habitat before construction is recommended. With the implementation of the recommended translocation measure and recommended good site practices (e.g. Use of quiet mechanical plant and regular dust suppression measures, etc.), no significant overall ecological impact on the crab species is anticipated. To further improve the carrying capability of the semi-natural stream for the two crab species of conservation interest, ecological enhancement for the retained section of the seminatural stream by provision of 6m buffer zone along its southern bank, removing existing artificial bank structures and planting of native/self-sustaining vegetation to reinstate its natural riparian habitat is proposed.
- 13.7.7 Though the impact of woodland loss at W3 is anticipated to be minor due to limited size and low-quality of the affected woodland patch, provision of woodland enhancement planting is recommended to enhance ecological performance of the area. According to the current proposal, enhancement planting of not less than 1.2ha will be provided. The Ecological Mitigation and Enhancement Plan is shown in **Figure No. 198172/B&V/FR/1302.**
- 13.7.8 With the implementation of the recommended mitigation measures, no adverse residual ecological impacts arising from the implementation of the proposed Project are anticipated.

13.8 Impact from Electric and Magnetic Fields

Key Assessment Scope and Key Criteria

13.8.1 The assessment for the electric and magnetic fields was conducted in accordance with the requirements set out under Section 3.4.12 of the EIA Study Brief and the guidelines issued by the International Commission on Non-ionizing Radiation Protection (ICNRIP).

Impact Assessment

13.8.2 Some sections of the existing 400kV overhead cables run across the Hong Po Road and San Hing Tsuen areas. On-site measurement of overhead cables has been conducted at representative locations of the proposed PTIs, public housing development areas, school sites and the proposed SPS to investigate the potential health hazard to humans due to exposure to electric field (ELF) and magnetic field (EMF) generated by the overhead lines.

13.8.3 The assessment results confirmed that the strength of the ELF and EMF generated from the 400kV overhead cables are well below the stipulated guideline limits issued by the ICNIRP in 1998. Thus. the ELF and EMF generated by overhead cables will not pose a hazard to human health.

13.9 Landscape and Visual Impact

Key Assessment Scope and Key Criteria

13.9.1 The landscape and visual impact assessment was conducted based on the criteria and guidelines in Annexes 10 and 18 of the EIAO-TM, the EIAO Guidance Note No. 8/2010 and Section 3.4.10 and Appendix I of the EIA Study Brief. According to the EIA Study Brief, the study area for the landscape impact assessment included areas within 500m from the boundary of the Project Site, while the assessment area for the visual impact assessment shall be defined by the visual envelope of the Project.

Landscape Impacts

- 13.9.2 Within the assessment area, 8 Landscape Resources (LRs) and three (3) Landscape Character Areas (LCAs) are identified. Given the rural nature of the Project area, the proposed developments of the Project will inevitably result in some landscape and visual impacts on the LRs and LCAs during construction and operation phases. It is not possible to fully mitigate all landscape impacts in relation to loss of natural/seminatural stream, hillside woodland, agricultural land or orchard during the construction period and early operation stage, mainly as long periods of time are required to sufficiently compensate for the associated impacts. A broad-brush tree survey has been carried out, it is found that four (4) numbers of identified large trees with DBH \geq 1m (including 2 nos. of *Ficus microcarpa*, 1 no. of *Melaleuca cajuputi subsp. cumingiana* and 1 no. of *Litchi chinensis*) are located within the footprint of the Project Site. Hence, the four (4) numbers of trees of large size are suggested to be felled. The other seven (7) numbers of trees of large size are not affected by project footprint and would be retained in-situ. The overall number of trees of large size identified within the assessment area is eleven (11). Three individuals of Aquilaria sinensis were found falling within the footprint of the proposed Road L7. To minimize the impact on the three individuals of Aquilaria sinensis, two of them will be preserved by transplanting while the remaining one will be felled down due to its poor health condition. No Old and Valuable Tree (OVT) was found within the project site boundary or assessment area. Among 9,615 - 11,055 nos. of trees within 500m Assessment Area, approximately 8,350 - 9,800 numbers of trees would be retained. The outline landscape plan is presented in Figure Nos. 198172/B&V/FR/1317a - 1317c.
- 13.9.3 **Table 13.7** shows the significant of landscape impacts before mitigation measures within the Project Site. The main impact for most of the affected LRs and LCAs would be the loss of greenery and existing trees. Approximately 1,300 nos. of existing trees will be affected due to the Project. To compensate the loss of greenery, not less than 1,300 nos. of new trees in different sizes is proposed to be planted as far as practicable. Existing trees will be preserved as far as practicable. If retaining trees are not practicable, transplant the affected trees to other suitable locations within the

Project Site or the adjacent areas will be considered. Tree removal will only be considered when tree preservation or transplanting is found unsuitable or impracticable. Tree Preservation and Removal Proposal will be prepared during detailed design stage, to finalise tree treatment and allocate compensatory planting areas including open space, sitting out areas and streetscape. With the implementation of the proposed mitigation measures, as detailed in *Table 13.8*, the residual impacts to LRs and LCAs can be reduced to an acceptable level.

			A330	ssment Area		
LR/			Cool f	No. of Trees Affected	Impact Significance BEFORE Mitigation (Substantial/ Moderate/ Slight/ Insignificant)	
Code		Name	Scale of Development	(Affected Species of Particular Interest)	Construction	Operation
		LR 1a – Siu Hang Tsuen	N/A	N/A	Insignificant	Insignificant
		LR 1c – Kai Lun Wai Village	N/A	N/A	Insignificant	Insignificant
LR1	Village/ Low-	LR 1e – Tuen Mun San Tsuen	N/A	N/A	Insignificant	Insignificant
	rise	LR 1f – Villa Pinada	3 - 5%	N/A	Slight	Slight
	Residential	LB 1h – Tsz Tin Tsuen	16%	40 - 50	Moderate	Moderate
	Development	LP 1d - San Hing Tsuen	2 _ 30/	20 - 25	Moderate	Moderate
		LP 1g - Chung Shan Area	4.0%	20 25	Substantial	Substantial
		LR 2e – Tong Hang Road Open Storage	N/A	N/A	Insignificant	Insignificant
		LR 2f – Kwong Shan Tsuen Open Storage	N/A	N/A	Insignificant	Insignificant
1.02	Open Storage (LR 2d – San Hing Tsuen Open Storage	N/A	N/A	Slight	Slight
LRZ	Workshop	LR 2b – HPR Workshops (Peripheral greening)	45 - 50%	30 - 40	Moderate	Moderate
		LR 2a – HPR Container Terminal	85 - 90%	190 - 230	Substantial	Substantial
		LR 2c – HPR Workshops with Scattered Greening	90 - 95%	240 - 300 (CEDD-T62)	Substantial	Substantial
LR3	Agricultural Land/	LR 3a – Actively Cultivated Land	15 - 17%	130 – 150 (CEDD-T23 & T503)	Substantial	Substantial
	Orchard	LR 3b – Abandoned Land	95 - 100%	160 - 170	Substantial	Substantial
	Hillside	LR 4a – Mixed Woodland	4 - 5%	90 – 120 (T589 & 3 nos. of Aquilaria sinensis)	Substantial	Substantial
LR4	Woodland	LR 4b – Plantation Woodland	N/A	N/A	Insignificant	Insignificant
		LR 4c – Mature Secondary Woodland	N/A	N/A	Insignificant	Insignificant
	Natural/Semi- Natural Stream	LR 5.1a – 5.1e, 5.1g, 5.1m, 5.2c, 5.2d, 5.4a – 5.4d & 5.8a – 5.8e	N/A		Insignificant	Insignificant
		LR 5.2a, 5.2b, 5.3a & 5.3b	106m (10%) of total length	-	Moderate	Moderate
LR5		LR 5.1i & 5.1f	356m (40%) of total length		Substantial	Substantial
	Channelized Watercourse	LR 5.5 & 5.6	N/A		Insignificant	Insignificant
		LR 5.7	1,160m (31%) of total length	-	Slight	Insignificant
		LR5.1h & 5.1k			Moderate	Moderate
		LR 5.1j & LR 5.1l			Moderate	Moderate
	Urban Area	LR 6a – Residential Blocks	6 - 7%	45 - 60	Slight	Insignificant
I P6		LR 6b – Key Transport Route	N/A	10 - 20	Insignificant	Insignificant
LINU	oi ball Al ed	LR 6c – Public Facilities	2 - 3%	15 - 25	Slight	Insignificant
		LR 6d – Industrial Blocks	N/A	N/A	Insignificant	Insignificant
LR7	(Construction Site	13 - 15%	40 - 45	Insignificant	Insignificant
	Shrubland	LR 8a – West Grassland	N / A	N /Δ	Insignificant	Insignificant
LR8	Grassland Mosaic	Mosaic	11/11	11/1	mərginneane	mərgiillitalit
LIVO		LR 8b – South West Grassland Mosaic	N/A	N/A	Insignificant	Insignificant
LCA1	Tuen Mun North Rural Fringe		23 - 25%	870 – 1030 (CEDD-T23, CEDD-T62 & T503)	Substantial	Substantial
LCA2	Tuen	Mun North Urban Area	5 - 6%	70 - 110	Slight	Insignificant
LCA3	Castle Peak Foothill		2%	90 – 120 (T589 & 3 nos. of Aquilaria sinensis)	Moderate	Moderate

Table 13.7 - Significance of Landscape Impacts before Mitigation within Assessment Area

ID No.	Mitigation Measure	Constuction	Operation
MM1	Tree Protection and Preservation	\checkmark	\checkmark
MM2	Tree Transplanting	\checkmark	\checkmark
MM3	Compensatory Planting and New Tree Planting	\checkmark	\checkmark
MM4 (1) (2)	Roadside Greening		\checkmark
MM5	Screen Planting	\checkmark	\checkmark
MM6	Landscape Treatment on Man Made Slope/ Retaining Structure	\checkmark	\checkmark
MM7	Noise Barrier Treatment		\checkmark
MM8	Minimize Light Pollution and Glare	\checkmark	\checkmark
MM9	Hoarding of Construction Works	\checkmark	
MM10	Enhancement of Semi-natural Stream (6 m buffer zone)	\checkmark	\checkmark
MM11	Landscape Work & Green Roof for Infrastructure		\checkmark
MM12	Woodland Enhancement Planting with Ecological Enhancement		\checkmark

(1) The Contractor will be responsible for landscaping during the agreed establishment and maintenance period. Other designated management agents to take up management of landscaping after end of agreed period.

(2) The Contractor will provide LCSD and HyD with water points, irrigation system where feasible, as well as proper access and safe working conditions for maintenance of the vegetation. LCSD is responsible for the maintenance of the vegetation as well as facilities required for the vegetation maintenance, e.g. the water points and irrigation system, unless another department has agreed to take up the maintenance responsibility of these facilities.

Visual Impacts

Construction Phase

13.9.4 Visual impacts are primarily due to the high-rise buildings development but also due to the construction of proposed road system, proposed SPS, water services reservoirs and natural terrain mitigation measures. In addition, visual impact will also involve the impacts from excavation works, site formation works, as well as the loss of greenery due to the removal of trees and vegetation. All Visual Sensitive Receivers (VSRs) are affected by the Project in different levels. Visual blockage would be gradually built-up by phase as a continuous process. Impact arising from the construction works are limited at lower level due to the existing views are partially screened by adjacent architectures, vegetation and level difference of landforms. With the implementation of proposed mitigation measures such as site hoarding and control of night-time glare, the visual impact during construction phase will be mitigated.

Operation Phase

13.9.5 From a visual perspective, given the nature and scale of the proposed development with high- rise buildings development, the Project will likely alter the visual context of the area. The carpark users in Tuen Tsz Wai and the residents in Po Tong Ha are anticipated to experience substantial visual impacts. On the other hand, a number of VSRs are anticipated to experience moderate visual impacts, including: the passengers on the train heading towards Siu Hong Station, road users at Hong Po Road roundabout, the hikers walking along the trails between Castle Peak and Por Lo Shan, the visitors of Miu Fat monastery and the visitors for tomb-sweeping around Chung Shan hillsides. The remaining VSRs would experience slight visual impacts. The photomontage for all VSRs are presented in *Figure Nos.* 198172/B&V/FR/1303 – 1316 that provide a view on the scale and extent of the proposed development.

- 13.9.6 Based on the impact assessment findings, a number of mitigation measures have been proposed. These include adopting alternative designs to prevent and/or minimise adverse impacts; remedial measures such as colour and textural treatment of building features; and compensatory measures such as the implementation of landscape design elements (e.g. tree planting and creation of new open space, etc.) to compensate for unavoidable adverse impacts and attempt to generate potentially beneficial long-term impacts.
- 13.9.7 For the DP1, which is the proposed SPS at SHR Site, in view of the small scale of development, no significant landscape impacts on LRs and LCAs, as well as visual impacts on VSRs are anticipated.
- 13.9.8 Overall, there will be some adverse effects brought up by the Project. Mitigation measures have been proposed and the project design has endeavoured every effort to minimise potential impacts to practical minimum. Assuming that full and appropriate mitigation measures are to be implemented during construction and operation phases, the residual landscape and visual impacts are perceived to be acceptable, as stated in EIAO-TMs and EIAO Guidance Note No. 8/2010.

13.10 Waste Management Implications

Key Assessment Scope and Key Criteria

13.10.1 The types of waste that would be generated during the construction and operation phases of the Project have been identified. The potential environmental impacts that may result from these waste materials have been assessed in accordance with the criteria and guidelines outlined in Annex 7 and Annex 15 of the EIAO-TM, and Section 3.4.7 and Appendix F of the EIA Study Brief.

Construction Phase

- 13.10.2 The main waste types to be generated during the construction phase of the Project would include construction and demolition (C&D) materials, chemical waste, general refuse and asbestos containing material (ACM). It is estimated that there will be a total of around 99,000 m³ of non-inert C&D materials, about 633,000 m³ of inert C&D materials, a few hundred litres per month of chemical waste, less than 650 kg per day of general refuse and some ACM to be generated during the construction phase of the Project. Reduction measures have been recommended to minimise the amount of materials generated by the Project by reusing C&D materials as far as practicable before off-site disposal.
- 13.10.3 The inert C&D materials generated from the Project will be reused within the Project or other concurrent projects as far as practical. For instance, during site clearance, site formation and infrastructure works, it is estimated that around 58% of the inert C&D materials will be suitable for reuse on-site as backfilling materials under this Project and the rest of inert C&D materials (about 233,000 m³) will be transported to other concurrent projects and/or public fill area (Tuen Mun Area 38 Fill Bank) for reuse. Temporary stockpiling areas are also identified to store the C&D materials for reuse under this Project. Provided that the waste is handled, transported and disposed of using approved methods, adverse environmental impacts would not be expected.

Operation Phase

13.10.4 The main types of waste to be generated during the operation phase of the Project would consist of municipal solid waste (MSW), screenings and chemical waste. It is expected that the Project would generate around 150 tonnes of MSW per day in total, about 3.5 m3 of screenings per week, and a minimal amount of chemical waste, mainly from maintenance activities on the road networks and the SPS within the PDA. The MSW generated would be conveyed to refuse collection points before being transported to the existing West New Territories (WENT) Landfill outside the PDA. Initiatives such as promoting recycling and providing recycling bins would be employed in order to minimise the amount of MSW to be disposed of at landfill. Provided that the waste generated in the operation phase is handled, transported and disposed of properly, no adverse environmental impacts are anticipated.

13.11 Land Contamination Impact

Key Assessment Scope and Key Criteria

- 13.11.1 The land contamination assessment is conducted in accordance with the criteria and guidelines as stated in the requirements given in Section 3.4.8 and Appendix G of the EIA Study Brief, as well as Annex 19 of the EIAO-TM.
- 13.11.2 The land contamination assessment examined the potential contaminative land uses within the Project Site and their potential impacts to future land use. The majority of the potentially contaminated sites could not be accessed for inspection of the site conditions during site walkover and permission could not be obtained from the land site owners/operators to carry out site investigation (SI) works at the time of preparation of the EIA report. As such, the assessment on the potential land contamination was conducted based on the findings from site appraisal comprises site walkover and review of historical aerial photographs and maps, historical spillage and leakage records and previous site investigations undertaken at the Project Site.
- 13.11.3 Based on the available information, 57 potentially contaminated sites have been identified within the Project Site. Based on the site survey and desktop review, the majority of the potentially contaminated sites have been identified as open storage, container yards, workshops and service yards. The source of potential land contaminating activities at the identified sites mainly relates to the spillage and accidents associated with the storage and use of chemicals. As such, it is considered that the potential land contamination at these sites would be localised.
- 13.11.4 The chemicals of concern (COCs) identified with the potential to be present at the potentially contaminated sites include: metals, Volatile Organic Compounds, Semi Volatile Organic Compounds, Petroleum Carbon Ranges and Polychlorinated Biphenyls. These COCs are readily treatable with proven remediation techniques in local remediation experience. By implementing the recommended remediation works, any contaminated site(s) identified within the Project Site could be cleaned up prior to construction/development.
- 13.11.5 The recommended remediation works would not only minimise the health risk to the future occupants arising from the exposure of the contaminated soil and/or

groundwater, it would also provide the opportunity to reuse the treated contaminated materials into useful materials for backfilling, which results in minimising the amount of waste disposing into the depleting landfill in Hong Kong and achieving a more sustainable development.

13.11.6 Since the identified potentially contaminated sites are still in operation, SI is unlikely to be carried out at this stage. There may also be change in land use prior to development within both the potentially contaminated sites and other surveyed sites. In view of this, it is recommended to conduct further works including site reappraisal. Findings from the re- appraisal will be presented in the supplementary Contamination Assessment Plan(s) (CAP(s)). Upon approval of the supplementary CAP and completion of the SI works, a Contamination Assessment Report(s) will be prepared to present the findings of the SI works. If contaminated soil and/or groundwater were identified, remediation should be carried out according to EPD's approved Remediation Action Plan(s) and Remediation Report(s) (RR(s)) should be submitted to EPD for agreement after completion of the remediation works. No development works of the contaminated portions shall be commenced prior to EPD's agreement of the RR(s).

13.12 Impact on Cultural Heritage

Key Assessment Scope and Key Criteria

13.12.1 A cultural heritage impact assessment has been conducted, including a built heritage impact assessment and an archaeological impact assessment, to evaluate the impacts on known or potential cultural heritage resources. The cultural heritage impact assessment followed the requirements of Annexes 10 and 19 of the EIAO-TM as well as those set out in Section 3.4.11 and Appendix J of the EIA Study Brief. The assessment area includes areas within a distance of 50m from the boundary of the Project Site.

<u>Archaeology</u>

- 13.12.2 The proposed developments are partly located within San Hing Tsuen Site of Archaeological Interest (SAI) and Siu Hang Tsuen SAI, and in very close proximity to Kei Lun Wai SAI. The evaluation of the previous findings, proposed works locations and geology, topography at the PDA, identified San Hing Tsuen SAI which may be adversely affected. Archaeological field survey of the identified area of archaeological potential (i.e. northern part of SHR Site) cannot be conducted at this stage as the area is currently in use by light industrial activities and structures. It is recommended that prior to the construction phase, a programme of archaeological field survey should be implemented for the development in northern part upon land resumption and clearance of structures. This will identify if significant deposits or features are present and follow up action can be recommended. The scope and programme of the proposed archaeological work shall be agreed with AMO. Subject to the findings of the archaeological work, appropriate mitigation measures would be proposed by the project proponent in prior agreement with AMO.
- 13.12.3 For the areas with low or no archaeological potential, works should be ceased and AMO should be informed immediately in case of discovery of antiquities or supposed

antiquities in the course of the construction works. Agreement from AMO would be sought on the follow- up actions if required.

<u>Built Heritage</u>

13.12.4 Literature review is conducted to collate relevant information on Declared Monuments and Graded Historical Buildings. Previous investigations within the current study area and surrounding area carried out under other projects (See Section 13.4 of EIA report) indicate little potential for built heritage or heritage as identified in Guidelines for Cultural Heritage Impact Assessment. As such, no further built heritage survey was deemed necessary. There are no proposed or declared monuments, graded or proposed to be graded historic buildings, Government historic sites or new items proposed for grading by the Antiquities Advisory Board within the boundary of the Project Site. In summary, it can be stated that no built heritage will be affected during construction and operation phases of the Project. If there are any buildings/structures both at grade level and underground which were built on or before 1969, AMO should be informed in an early stage or once identified.

13.13 Environmental Monitoring and Audit Requirements

13.13.1 An Environmental Monitoring and Audit (EM&A) programme will be implemented during the construction and operation phases to regularly monitor the environmental impacts on the neighbouring sensitive receivers. Any action required during the construction and/or operation phases are also recommended for implementation. EM&A requirements for air quality, construction noise, road traffic noise, water quality, waste management, land contamination, terrestrial ecology, landscape and visual impacts and cultural heritage have been recommended. Regular site inspection and audits will be conducted during construction phase to ensure that the recommended mitigation measures are properly implemented. The EM&A requirements are specified and detailed in the EM&A Manual. A summary of the EM&A requirements by each of the environmental parameters is presented in **Table 13.9**.

	EM&A Ref.	Construction Phase	Operation Phase
Air Quality Impact	S.5	\checkmark	\checkmark
Noise Impact	S.6	\checkmark	\checkmark
Water Quality	S.7	\checkmark	x
Sewerage and Sewage Treatment	S.8	x	x
Ecological Impact	S.9	\checkmark	x
Impacts from Electric and Magnetic Fields	S.10	x	x
Landscape and Visual Impacts	S.11	\checkmark	\checkmark
Waste Implication	S.12	\checkmark	x
Land Contamination	S.13	\checkmark	x
Cultural Heritage Impact	S.14	\checkmark	x

Table 13.9 - Summary of EM&A Requirements

13.14 Summary

13.14.1 Assessments in the aspects of air quality, noise, water quality, sewerage and sewage treatment implication, ecology, impact from electric and magnetic fields, landscape and visual impacts, waste management implication, land contamination and cultural heritage were conducted under the study. It is anticipated that there would be no insurmountable issue from the environmental point of view.

14 PRELIMINARY LAND CONTAMINATION AND REMEDIATION STUDY (LCRS)

14.1 General

14.1.1 The objective of the Preliminary Land Contamination and Remediation Study (LCRS) is to identify the potential contaminated sites within the Sites and outline the required intrusive site investigation, assessment criteria, result interpretation of the land contamination assessment and requirements of the Remediation Action Plan (RAP) associated with the Developments and the Infrastructure. A summary of various aspects are presented in the section below:

14.2 Site Appraisal Findings

- 14.2.1 A site appraisal, in the form of desktop review and site inspection, was carried out to identify the potential land contamination issues for the Project. Since the majority of the Project Site was not accessible at the time of preparation of the report, SI of the potentially contaminated area was unable to be conducted at this stage. Therefore, the types and extent of contaminants, if any, could not be identified and confirmed at this stage.
- 14.2.2 Based on the findings of the site appraisal, a total of 82 areas, as shown in *Figure No.* **198172/B&V/FR/1401** were identified within the Assessment Area, of which 57 areas were considered potentially contaminated, 5 areas were non-potentially contaminated, and 20 areas were unidentifiable. In the Contamination Assessment Plan (CAP), the extent of the Site Investigation (SI) was determined based on findings from the site appraisal. For each potentially contaminated area, the proposed SI in regular grid pattern for land contamination was determined based on recommendations given in Practice Guide for Investigation and Remediation of Contaminated Land (the Practice Guide).
- 14.2.3 The CAP has determined the approach for SI based on the best available current information and thus the actual sampling and testing strategies could be subject to change after a detailed site walkover is undertaken. A supplementary CAP should be submitted when these identified areas are accessible for re–appraisal.
- 14.2.4 A Contamination Assessment Report (CAR) will be prepared to present the findings of the future SI and laboratory analytical results. The analytical results will be compared against the Risk-Based Remediation Goals (RBRGs) standards.
- 14.2.5 Upon completion of remediation works (if necessary), a Remediation Report (RR) will be prepared and submitted to EPD to demonstrate that the decontamination works are adequate and are carried out in accordance with the approved CAR and RAP prior to commencement of any proposed construction works. No construction works will be carried out at the potentially contaminated areas before the approval of RR by EPD.

14.3 Site Investigation and Sampling Plan

14.3.1 In the CAP, the extent of the SI was determined based on findings from the site appraisal. For each potentially contaminated area, the proposed SI in regular grid pattern for land contamination was determined based on recommendations given in

the Practice Guide. In addition, hot spot locations were also identified for SI. A total of 582 sampling locations (568 from regular grid and 14 from hot spots) were identified.

14.3.2 Further site re-appraisal should be conducted for the Project to assess any changes in land use which may give rise to potential land contamination issues and to ascertain the contaminative sources and hotspots of contamination within the Project Site. A supplementary CAP containing the SI proposal will be submitted to EPD for endorsement and the necessary SI to assess the land and groundwater contamination will be conducted. Based on the laboratory results of the future site investigation, an implementation programme and cost estimate of the remedial works (if required) should be proposed.

14.4 Remediation

- 14.4.1 The actual remediation methods should be confirmed after completion of the site reappraisal and EPD's agreement on the CAR and RAP at the later stage of the Development. The RAP will provide details of the remedial actions for any identified contaminated soil and groundwater.
- 14.4.2 Based on the past and current land use of the potentially contaminated sites, the associated potential Chemical of Concerns (COCs) have been identified. There are several technologies commercially available to tackle these contaminants in soil. Technologies that are commonly used in Hong Kong are biopiling and cement solidification/stabilization. These ex-situ methods were proven to be effective in treating the target COCs (cement solidification/stabilization on metals and biopiling on hydrocarbons). Given the size of the Project Site, there would be enough space available for the handling and treatment of the contaminated soil and the two methods are considered to be appropriate.
- 14.4.3 Contaminants in groundwater exceeding the RBRGs were not commonly found in Hong Kong. This may be due to the fact that groundwater in Hong Kong were not for drinking purposes and the exposure pathway of contaminants in groundwater to human receptors would be associated with inhalation of volatiles rather than direct exposure. Some examples of remediation techniques of contaminated groundwater (e.g. air sparging, recovery trenches / wells, in-ground containment/capping and permeable reactive barriers) are shown in the Practice Guide from EPD.

15 IMPLEMENTATION PROGRAMME

15.1 General

15.1.1 This summarize the key milestones of the implementation programme for proposed developments.

15.2 Implementation Programme

Development Staging

15.2.1 In order to ensure a balanced and programmed development with orderly rehousing/relocation of qualified clearees, the PDA including the associated engineering infrastructure is proposed to be divided into 4 main stages for implementation as illustrated in **Figure Nos. 198172/B&V/FR/1501a** and **1501b**. A summary of implementation programme is shown in *Table 15.1*.

Key Developments in Stage 1

- 15.2.2 The proposed works to be conducted in the first stage includes the following:
 - Site formation and construction of public housing at SHR Site Extension;
 - Construction of SPS at SHR Site and the associated rising mains;
 - Upgrading of existing sewers & modification of the existing TM54 SPS;
 - Construction of the realigned Hong Po Road;
 - Carrying out of junction improvement works;
 - Construction of footbridge to connect SHR Site and Siu Hong Station; and
 - Carrying out of associated drainage works, sewerage works and waterworks etc..

Key Developments in Stage 2

- 15.2.3 The proposed works to be conducted in the second stage includes the following:
 - Construction of proposed Road L7;
 - Carrying out of junction improvement works; and
 - Laying of watermains along the proposed Road L7.

Key Developments in Stage 3

- 15.2.4 The proposed works to be conducted in the third stage includes the following:
 - Site formation and construction of public housing and PTI at SHR Site;
 - Site formation and construction of schools at SHR Site; and
 - Carrying out of associated drainage works, sewerage works and waterworks etc..

Key Developments in Stage 4

- 15.2.5 The proposed works to be conducted in the fourth stage includes the following:
 - Site formation and construction of public housing and PTI at HPR Site;
 - Site formation and construction of schools at SHR Site Extension;
 - Construction of natural terrain mitigation measures at HPR Site;
 - Construction of fresh/ salt water service reservoirs at HPR Site; and

• Carrying out of associated drainage works, sewerage works and waterworks etc.

Stage	Phase	Description of Works	Land Clearance & Construction Period
	1a	 San Hing Road Site Extension (Residential Site) Site clearance, land decontamination and site formation works Internal road, pipe works and landscaping Construction of public housing 	2025-2030
Stage 1	1b	 Sewage Pumping Station at SHR Site Site clearance, land decontamination and site formation works Sewage pumping station construction and the associated rising mains Upgrading works of existing sewers & modification of TM54 SPS 	2026-2030
	1c	 <u>Realigned Hong Po Road</u> Site clearance, land decontamination and site formation works Road and junction Improvement works Pipe works and utilities works 	2026-2029
Stage 2	2	 <u>Proposed Road L7</u> Site clearance and site formation works Road and junction improvement works Pipe works and utilities works 	2026-2029
Stage 3	3a	 San Hing Road Site (Residential Site) Site clearance, land decontamination and site formation works Internal roads and PTI Pipe works and landscaping Construction of public housing 	2025-2031
	3b	 Site clearance, land decontamination and site formation works Internal road, pipe works and landscaping Construction of schools 	2026-2031

Table 15.1 – Summary of Implementation Programme within the PDA

Stage	Phase	Description of Works	Land Clearance & Construction Period
Stage 4	4a	 <u>Hong Po Road Site</u> Site clearance, land decontamination and site formation works Internal road and PTI Pipe works and landscaping Natural terrain hazard mitigation measures Construction of water service reservoirs Construction of public housing 	2026-2033
	4b	 <u>School Site at SHR Site Extension</u> Site clearance, land decontamination and site formation works Internal road, pipe works and landscaping Construction of schools 	2026-2033

15.2.6 The proposed implementation programme as shown in *Table 15.1* above would be reviewed in later investigation and design stage of the Project.

16 CONCLUSIONS AND RECOMMENDATION

16.1 General

- 16.1.1 To meet the demands for housing from the public, the Government has identified the sites at SHR and HPR for potential public housing development and associated GIC facilities. To support this initiative, the Study examined the technical feasibility through preliminary engineering and environmental assessments. With the provision of the recommended improvement works, the Study concluded that there is no insurmountable problem in terms of traffic, environmental, geotechnical, services and utilities, land acquisition, cultural heritage, air ventilation, visual and landscape aspects.
- 16.1.2 Given the various development constraints and infrastructure needs, the technical assessments adopted a holistic approach and the findings were taken forth to formulate site formation and infrastructure proposals including implementation strategy and preliminary programme in support of the proposed amendments to the OZP. A Preliminary Site Layout Plan showing the proposals have been prepared and is shown on *Figure No.* 198172/B&V/FR/0201. Findings from the relevant engineering and environmental assessments conducted is summarized below, and it is revealed that the proposed developments are technically feasible.

16.2 Air Ventilation

- 16.2.1 On basis of the preliminary layout plan, an expert evaluation of wind performance on the Project is carried out.
- 16.2.2 According to the findings of the AVA-EE, the annual prevailing wind comes from NNE, NE, E, SE, SSE and S directions while the summer prevailing wind comes from SSE, S, SSW and SW directions. The building block layout has carefully considered and incorporated good design practices in air ventilation aspect. After taking into account the existing topography, the location of the existing built-up areas, as well as provision of mitigation measures, it is considered that the Project would not cause adverse air ventilation impacts to the surrounding environment. The expert evaluation provided a qualitative assessment on feasibility of the Project, and further quantitative AVA will be required for scheme optimization at detailed design stage.

16.3 Traffic and Transportation

- 16.3.1 To support the Project, new traffic and transport facilities, as well as upgrading of existing facilitates, are recommended. Junction improvement proposals are formulated and listed below. With the proposed junction improvement schemes, the assessment revealed that all key junctions will be operating within capacity in design year 2031 and 2036.
 - Lam Tei Interchange / Castle Peak Road Lingnan (J1);
 - Tsing Lun Road / Hong Po Road (J4);
 - Tsing Lun Road / Tsz Tin Road/Access Road to Siu Hong Court (J6);
 - Hing Kwai Street/ Tsing Wen Road / Ming Kum Road (J12);
 - Proposed Road L7 / Hing Fu Street (J13);

- Proposed Road L7/ Road L54D (J15); and
- Tin King Road / Ming Kum Road (J21).
- 16.3.2 The proposed Road L7 will need to be connected to the realigned Hong Po Road, thus provides a direct link from Lam Tei Interchange to Ming Kum Road forming a direct east-west transport corridor. The concept of the proposed Road L7 is to provide an alternative access route to the proposed public housing developments so as to minimize the traffic impact to the surrounding road network.
- 16.3.3 Realignment and modification of Hong Po Road is proposed such that it matches with the alignment of the proposed Road L7. The realigned section of Hong Po Road will be a 10.3m wide two-way carriageway with footpath on both sides.
- 16.3.4 For SHR Site, the PTI is proposed at the south of the SHR Site access by a one-way access road connecting to Tsing Lun Road. Four bus bays will be provided within the SHR PTI for the provision of future franchised buses. One Green Minibus (GMB) bay and one taxi bay are recommended to provide comprehensive coverage of the public transport services to accommodate the demand from SHR Site. The one-way traffic arrangement would allow bus traffic turn in/out of the terminus without any conflicting traffic movement.
- 16.3.5 Another PTI is proposed at the southern side of HPR Site with access off Hong Po Road. Six bus bays are proposed to be provided within the HPR PTI. One GMB bay and one taxi bay are recommended to provide comprehensive coverage of the public transport services to accommodate the demand from SHR Site Extension and HPR Site. To shorten the walking distance from HPR Site to the HPR PTI, ramp and stair are suggested to connect the HPR access road and the northern footpath of the PTI.
- 16.3.6 With the construction of propsoed Road L7 ,realigned Hong Po Road and implementation of proposed junction improvement works and enhancement on public transport services, it is anticipated that the proposed developments would not induce insurmountable traffic impact onto the adjacent road network. The proposed developments is acceptable in traffic point of view.

16.4 Pedestrian Connectivity

- 16.4.1 The existing pedestrian network in vicinity of Siu Hong Court would be changed upon the population intake of TM54. The information extracted from the TM54 Review and are summarized in below:
 - Junction of Tsing Lun Road / Siu Hong Road would become a signalized junction with crossing facilities at all arms of the junction;
 - A new footpath alongside the LRT tracks and Siu Hong Court has been proposed by Housing Department; and
 - A vertical mechanical transport system (including escalators and elevator) would be constructed and integrated with the existing stairs and footbridge connecting Siu Hong Road and Siu Hong Station PTI (North).
- 16.4.2 To mitigate the impact of the existing signalized crossing at junction of Tsing Lun Road/ Ng Lau Road near Tsing Lun Road roundabout, and to alleviate the pedestrian condition of northern Siu Hong Road footpath, a footbridge system (FB2) with
staircase, lifts and escalators is proposed to connect the open space next to the proposed SHR SPS at SHR Site to the existing footbridge (FB1) near Ng Lau Road. A covered walkway (CW1) and modification of existing walkway, if necessary, are proposed to connect the existing footbridge near Ng Lau Road and the Siu Hong Station. The proposed footbridge system will improve the pedestrian connectivity and its attractiveness would encourage the pedestrians to access to Siu Hong Station.

- 16.4.3 Pedestrian walkways and cycle tracks will be constructed along the proposed Road L7 and the realigned Hong Po Road to promote green commuting. The proposed cycle tracks will be connected to the nearby residential areas (e.g. Tuen Mun New Town) and public transport facilities (e.g. Lam Tei LRT Station and Siu Hong Station) where practicable. Upgrading/modification works to existing footbridge at Ng Lau Road and associated covered walkway are proposed to improve the walking environment and promote green commuting. All these measures are proposed to facilitate walking and cycling across the PDA and in the wider community. This would also in turn help to reduce road-based traffic and hence their associated vehicular noise and pollutants emission.
- 16.4.4 Covered walkway is proposed for Footpaths P3, P4, P5, P8, P9, P10, P11, P13, P14, P16, P17 and CW1 to enhance the walking environment.
- 16.4.5 Proposed accesses will be provided to maintain pedestrian linkage to the existing houses in the vicinity of the proposed public housing developments including:
 - A. Proposed access to Tsz Tin Tsuen from the realigned Hong Po Road;
 - B. Proposed access to Villa Pinada from the realigned Hong Po Road; and
 - C. Proposed footpath from the existing village houses at the north of SHR Site Extension to the proposed access road to the school sites at SHR Site Extension.
- 16.4.6 With the implementation of the proposed pedestrian system and associated improvement works, it is anticipated that all of the footpaths/ footbridge under assessment would be operated within capacity. The proposed developments will not induce insurmountable problem to the pedestrian network from traffic point of view.

16.5 Geotechnical

- 16.5.1 A review of the existing GI records available was conducted. The findings indicate that the proposed sites are covered by 0m to 9.1m of fill layer, as well as 0m to 6.2m of alluvium layer or 0m to 5.1m of colluvium. These superficial material are sandy in nature and should not be highly compressible material, it is considered feasible for supporting light duty retaining structures. No steep rock head depression is founded in existing borehole record, deep foundation for the proposed developments is considered feasible.
- 16.5.2 However, given the borehole record for the Site is very limited, the site should be further characterized by more GI conducted in later stage of the Project. From the current available information, no adverse geological and geotechnical features are observed. The proposed works for the proposed development is considered feasible.
- 16.5.3 A total of 17 registered man-made features are identified to be located within HPR Site, 3 registered man-made features are identified to be located within SHR Site, 2

registered man-made feature is identified to be located within SHR Site Extension and 4 registered man-made features are identified to be located within proposed Road L7. Failure of these slopes may affect the nearby infrastructure and future development. Therefore, further assessment is recommended.

- 16.5.4 There are 8 non-registered geotechnical features within/ in close proximity to the site boundary which will potentially be affected by the proposed developments.
- 16.5.5 Screenings for NTHS were carried out in accordance with "Alert Criteria" stated in GEO Report No. 138 (Second Edition, 2016). Natural terrain hazard mitigation measure is required for Catchment A. Active mitigation measure (Rigid barrier) along the slope toe within the site boundary is proposed to mitigate the potential channelized debris flow and boulder fall hazards of Catchment A. The findings of the NTHS and the proposed mitigation measures should be reviewed in later stage of the project when more geological /geotechnical information is collected from detailed field mapping and ground investigation.

16.6 Site Formation

- 16.6.1 Based on the existing topography, a preliminary site formation scheme is proposed to achieve a balanced cut and fill quantities and to minimize the export or import of fill materials.
- 16.6.2 In view of the gentle topographic nature of the eastern section of SHR Site, this section will be formed at the level between +6.0mPD to +7.6mPD by gentle slope grading.
- 16.6.3 Considering the western part of the SHR Site which is generally flat, the site will be formed at the level between +7.5mPD to +12.0mPD by slope grading.
- 16.6.4 At SHR Site extension, due to the gentle topographic nature, this section will be formed at the level between +9.8mPD to +12.2mPD by slope grading and each school site will be formed with uniform level with slight slope grading. The proposed formation levels of the proposed secondary school, the western primary school and the eastern primary school are +12.2mPD, +12mPD and +10.5mPD respectively.
- 16.6.5 Since the western part of the HPR Site is hilly and the eastern part is generally flat, the site will be formed in several platforms with levels rising diagonally from the southeast to the northwest direction. The uppermost platform will be formed at the level between +21.0mPD to +27.0mPD. The lowest platform will be formed at the level between +8.0mPD to +12.0mPD. An estate road at the level between +16.5mPD to +21.0mPD will be formed by retaining structures separating the two platforms.

16.7 Drainage, Sewerage, Water Supply and Utilities

16.7.1 The technical assessments have identified the needs for new provision of infrastructures and upgrading of existing utilities. Preliminary arrangements have been proposed to support the proposed developments and mitigate impacts arising from the proposed developments. With these provisions in place, adverse impacts with respect to drainage, sewerage, water supply and utilities are not envisaged.

16.8 Environmental, Landscape and Visual, Land contamination and Sustainability

16.8.1 The technical assessments have examined the environmental impacts and planning considerations. Preliminary mitigation measures are proposed to mitigate or minimize adverse impacts as much as practicable. With the recommended measures in place, it is envisaged that there would be no insurmountable issue from the environmental and planning aspects. Further liaison with relevant authorities at the detailed design stage is required to confirm proper implementation of these mitigation measures.

16.9 Land Requirement

16.9.1 Based on the proposed extent of land required for the implementation of the proposed developments, the Study has identified the extent of existing land use and numbers of facilities or properties that may be affected by the proposed developments. Further liaison with LandsD will need to be carried out at a later detailed design stage.

END OF TEXT

FIGURES



	A Ca T	© Cop	right by B	lack & Veatc	h Hong Kon	ng Limited
	The second se	LEGEN				
in the second se			PROF	POSED DEVEL	OPMENT AF	REA BOUNDARY
	130 13 130 130 13 130 13 130 130 13 130 130 13 130 130 130 130 130 130 130 130 130 130					
	200 104 200 200 200 200 200 200 200 200 200 200					
2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	N II N I					
	BABR 14 14 14 14 14 14 14 14 14 14					
TO Dean Storrage III						
	Still Film	Revision	Date	Descrip	tion	Initial
		Loitial	Designed	Checked	Drawn	Checked
		Date	08/20	08/20	08/20	08/20
		Approved				
	12 13 14 15 15 15 15 15 15 15 15 15 15					
		Agreemen	t No. CE	68/201	7 (CE)	
		Project	SITE F	ORMAT		
		INFR/	ASTRU DEVEL (L WOR	KS FOR
	14 14 10 10 10 10 10 10 10 10 10 10	ROAD	AND F	IONG P	O ROA	D, TUEN
		Drawing	title			
			LO	CATION	I PLAN	
	11 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Drawing 1	10.		S	Scale
		198	172/B&	V/FR/01	01	: 2500 (A1) : 5000 (A3)
	The second secon		_, _ ~			
			€±	木工程	拓展	署
	· · · · · · · · · · · · · · · · · · ·	ĆE	DD Civ	/il Engin velopme	eering ent Dep	and partment
		_	., -	R	®	
	ASTLE PEAK ROAD	BLAC	K & VEA 博 威 工	ATCH HON 程 顧 問	G KONG 有限2	LIMITED 公司



	[]
	© Copyright by Black & Veatch Hong Kong Limited
	PROPOSED DEVELOPMENT AREA BOUNDARY
	PROJECT SITE BOUNDARY (SUBJECT TO REVIEW)
	POTENTIAL SCHOOL SITE
	PROPOSED ROADWORKS (INDICATIVE)
	PROPOSED TRAFFIC IMPROVEMENT MEASURES (INDICATIVE)
	ROADWORKS UNDER OTHER PROJECTS
	RESIDENTIAL USE (INDICATIVE)
	RETAIL/SOCIAL FACILITIES / CARPARK (INDICATIVE)
	CYCLE TRACK (INDICATIVE)
	PROPOSED FRESH WATER SERVICE RESERVOIR (INDICATIVE)
	PROPOSED SALT WATER SERVICE RESERVOIR (INDICATIVE)
	PROPOSED COVERED PUBLIC TRANSPORT INTERCHANGE (PTI) (INDICATIVE)
	PROPOSED SEWAGE PUMPING STATION
The second secon	+7 0 PROPOSED FORMATION LEVEL
	PROPOSED RETAINING STRUCTURF
	(INDICATIVE)
	PROPOSED RIGID BARRIER (INDICATIVE)
	+149 mPD APPROX. BUILDING HEIGHT
K 1	
mPD	
	Revision Date Description Initial
	Designed Checked Drawn Checked Initial JLI YKL SZ JW
Y 10 </th <th>Date 08/20 08/20 08/20 08/20 Approved </th>	Date 08/20 08/20 08/20 08/20 Approved
	CE 68/2017 (CE)
	SITE FORMATION AND
	INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING
	ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY
	Drawing title
	PRELIMINARY SITE LAYOUT PLAN
	Drawing No. Scale
	198172/B&V/FR/0201 1 : 2500 (A1) 1 : 5000 (A3)
EREATED HERE I	198172/B&V/FR/0201 1:2500 (A1) 1:5000 (A3)
The ABE MERSEN I as a line of the ABE ABE ABE MERSEN I as a line of the ABE	198172/B&V/FR/0201 1 : 2500 (A1) 1 : 5000 (A3)
ER-AVERAMENTIAL LASA L	198172/B&V/FR/0201 1 :: 2500 (A1) 1 :: 5000 (A3) 1 :: 5000 (A3) Civil Engineering and Development Department
	198172/B&V/FR/0201 1 :: 2500 (A1) 1 :: 5000 (A3)
ER-AND-MEMBER Las Las Las Las Las Las Las Las	198172/B&V/FR/0201 1 :: 2500 (A1) 1 :: 5000 (A3)



	/
	© Copyright by Black & Veatch Hong Kong Limited
	LEGEND:
	PROPOSED DEVELOPMENT AREA BOUNDARY (SUBJECT TO REVIEW)
	TSING SHAN FIRING RANGE BOUNDARY
	RAILWAY PROTECTION ZONE
	70
	150 M
	Revision Date Description Initial
	Designed Checked Drawn Checked
	Initial BF CMW SZ JW Date 01/21 01/21 01/21 01/21
	Approved
	Agreement No.
	CE 68/2017 (CE)
	Project title
	INFRASTRUCTURAL WORKS FOR
	ROAD AND HONG PO ROAD, TUEN
	MUN - FEASIBILITY STUDY
	Drawing title
	PROTECTION ZONE
LAM TEI INTERCHANGE	
	Drawing No. Scale
	198172/B&V/FR/0202 1 : 2500 (A1) 1 : 5000 (A3)
	CEDD Civil Engineering and
	Development Department
	BLACK & VEAICH HONG KONG LIMITED 博威工程顧問有限公司

CAD Filename = C:\Daily Work - Y\20210706D\198172-B&V-FR-0301.dgn

C Copyright by Black	& Veatch Hong Kong Limite	od
LEGEND:	PROPOSED DEVELOPI	MENT
	AREA OF INFLUENCE	Ε (ΑΟΙ)
	CRITICAL JUNCTIO	N
	PLANNED LAYOUT	
Revision Date	Description	Initial
Initial JLI	YKL S	SZ JW
Approved	11/19 11.	/19 11/19
Agreement No.	E 68/2017 (CI	E)
Project title		
INFRASTRU THE DEVEL	JCTURAL WO	ORKS FOR
ROAD AND MUN - F	HONG PO RO	OAD, TUEN STUDY
Drawing title		
RO	AD JUNCTIC	NS
Drawing No.		Scale
198172/RA	&V/FR/∩⊿∩1	1 : 6000 (A1)
		· • 12000 (A3)
	木工程拓 vil Engineerir	展署
	evelopment D	Department
	Ŗ,	
BLACK & VE 博威工	ATCH HONG KO	NG LIMITED 艮 公 可

© Copyright by Black & Veatch Hong Kong Limited	
LEGEND:	
PROPOSED DEVELOPMEN	ΝT
AREA OF INFLUENCE ((AOI)
ROAD LINK	
PLANNED LAYOUT	
Revision Date Description	Initial Checked
Initial JLI YKL SZ	JW
Approved	9 11/19
Agreement No.	
Project title	
SITE FORMATION A	ND RKS FOR
THE DEVELOPMENT AT S	AN HING
MUN - FEASIBILITY S	TUDY
Drawing title	
LOCATION ON KE	ΞŶ
ROAD LINKS	
	See 1-
Uruwing No.	scale
198172/B&V/FR/0402	1 : 6000 (A1) 1 : 12000 (A3)
+ 木 T 程 坊 国	
CEDD Civil Engineering	and
	hainneill
BLACK & VEATCH HONG KONG 博威工程顧問有限	LIMITED 公 可

CAD Filename = Y:\Daily Work\20210713E\11\198172-B&V-FR-0501.dgn

CAD Filename = Y:\Daily Work\20210709c\198172-B&V-FR-0502.dgn

	11/5 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	© Copyright by Black & Veatch Hong Kong Limited
	B16 816	LEGEND:
	AN TEATSTOP	PROPOSED DEVELOPMENT AREA BOUNDARY (SUBJECT TO REVIEW)
	6.5 ₊	POTENTIAL SCHOOL SITE (SUBJECT TO REVIEW)
		FOOTPATH
	蔬菜統營處 藍地菜站	EXISTING FOOTBRIDGE ACROSS
	Lam Tel Vegetable Depot	PROPOSED FOOTBRIDGE CONNECTING
		ACROSS NG LAU ROAD
		PROPOSED STAIRCASE
		PROPOSED LIFT
	5.1	PROPOSED COVERED WALKWAY
1 1		MODIFICATION TO THE EXISTING WALKWAY AS NECESSARY
••••••••••••••••••••••••••••••••••••	7.2 Op	PROPOSED ESCALATOR
Регета № 10 № 10 № 10 № 10 № 10 № 10 № 10 № 1	C 6,8+	\longleftrightarrow footpath accessed
Note: 100 -		
Image: State Deck Decker (F) (M) 201131 Image: State Decker (F) (F) 201131 Image: State Decker (F) (F) 201131 Image: State Decker (F) 201131 Image: St		
Britiston Loss Britiston Loss Britiston Loss Britiston Loss Address Britiston Loss Britiston Loss Britiston Loss Britiston Loss Britiston Loss Britiston Loss<		5.
Exercise		
Printing Control (CE) Printing Co		
Provide The Second Seco		

Image: State Sta	7.0 to the second secon	
Image: State of the state	7	4
Ret Black Date Description Initial No. Description Description No. Description Description No. Description Description No. Description Description Description Description Description No. Description Description Description Description Description Description Description Description Description Description Description Description Description Description <t< th=""><th></th><th></th></t<>		
Berision Berision Description Intition Initial ULL VL S2 Js Initial ULL VL S2 Js Initial ULL VL S2 Js Initial VL S2 Js Js Initial VL S2 VL Is S2 Initial VL VL S2 VL VL S2 VL Initial VL		~^
herister bate trease-tett-on 1-111a designed Drocked Brown Creased itit+ai d.1 rk. sz d. itit+ai d.1 rk. sz d. derined Orocked Brown Creased itit+ai d.1 rk. sz d. derine C2 (20) 01/20 01/20 01/20 derine Sitte FORMATION AND NFRASTRUCTURAL WORKS FOR The Development At SAN HING Solad AND HONG PO ROAD, TUEN WIN - FEASIBILITY STUDY 2/2/2/2/2/2/2/2/2/2/2/2/2/2/2/2/2/2/2/	5.0 小 宮華花園 L	
Bedision 2ata Beach Iphi on Checked Bedision 2ata Drown Checked Lititic JJ Bere 0'/20 01/20 01/20 01/20 O'/20 Approved CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Draving title PEDESTRIAN FLOW PATHS SCIE 198172/B&V/FR/0502 Lititopic title SCIE Draving title SCIE Digiting Yor SCIE Digiting Yor SCIE Lititic SCIE Digititie SCIE Digitititie SCIE		
No. Description Initial Initial Uli YK. S2 JK Date 0'/20 01/20 01/20 01/20 01/20 Approve CE 68/2017 (CE) Project title State FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING STATE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING NUN - FEASIBILITY STUDY Drowing title Drowing title Drowing title Drowing title Image: neg base Surre Surre Drowing title Image: neg base Surre Surre Drowing title Image: neg base Surre Surre Drowing base Surre Surre Surre Image: neg base Surre Surre Surre Drowing base Surre Surre Surre Drowing title Image: neg base Surre Surre Drowing title Image: neg base Surre Surre Drowing title Image: neg base Surre Surre <th></th> <th></th>		
Periation Data Description 1nitia Initial JL Y(L S2 JV Jote 01/20 01/20 01/20 01/20 Approved CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Dravid CE 101/20 01/20 01/20 Dravid CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Dravid CSHEET 2 OF 2) Dravid Scole 198172/B&V/FR/0502 1000 (AI) 198172/B&V/FR/0502 1000 (AI) 198172/B&V/FR/0502 1000 (AI) Exception Civil Engineering and Development Department Image Mid A Mid A Mid A Mid A Mid A Mid Image Mid A Mid A Mid A Mid		
Iritial JLI YKL S2 JN Bare 3'/20 01/20 01/20 01/20 01/20 Approvec CE 68/2017 (CE) Project Hille SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drowing Hille PEDESTRIAN FLOW PATHS Creating HOL Scale 1 1000 (41) 1 2000 (43) Image: Scale Scale 1 1000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41) 1 2000 (41)		Revision Date Description Initial Designed Checked Drawn Checked
Date B1/2C D1/2B C1/2B C1/2B C1/2B Approved CE 68/2017 (CE) Project fittle SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing fittle PEDESTRIAN FLOW PATHS (SHEET 2 OF 2) Drawing No. Saction 198172/B&V/FR/0502 1 ± 1000 (A1) 1 ± 2000 (A3) Exercise Civil Engineering and Development Department Exercise BLACK & VEATCH HONG KONG LIMITED 博威工程額問有限公司 ELACK & VEATCH HONG KONG LIMITED	°E 69	Initial JLI YKL SZ JW
CE 68/2017 (CE) Project fille SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Ordeing fille Droeing fille Droeing Mo. 198172/B&V/FR/0502 State 在 好 展 着 展 着 Civil Engineering and Development Department	+5.3	Date 01/20 01/20 01/20 Approved
CE 68/2017 (CE) Project 1110 SITE FORMATION AND INFRASTRUCTURAL WORKS FOR INFR		
CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Draving title Certain flow PATHS SHEET 2 OF 2) Draving No. Scale 1 2000 (A1) 1 2000 (A2) Civil Engineering and Development Department ELACK & VEATCH HONG KONG LIMITED M M I E M M A R 2 J		
Project #### Froject #### Project #### SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drewing #### PEDESTRIAN FLOW PATHS (SHEET 2 OF 2) Drewing No. 198172/B&V/FR/0502 1 : 1000 (A1) 1 : 2000 (A3) Civil Engineering and Development Department EXACK & VEATCH HONG KONG LIMITED 博 威 工 程 顔 間 有 限 公 司		CE 68/2017 (CE)
SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing title Drawing title BEDESTRIAN FLOW PATHS (SHEET 2 OF 2) Drawing No. 198172/B&V/FR/0502 1: 1000 (A1) 1: 2000 (A3) Drawing No. Scole 1: 2000 (A3) Drawing No. Scole 1: 2000 (A3) Drawing No. Scole 1: 2000 (A3) Drawing No. Drawing No. Scole Distribution	Cooling 1.200 + 5.5	Project title
INFRASTRUCTORAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drowing title Drowing title Improve the second state of the second	9.5	SITE FORMATION AND
ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing title PEDESTRIAN FLOW PATHS (SHEET 2 OF 2) Drawing No. 1 : 1000 (A1) 1 : 2000 (A3) 1 : 2000 (A3)	+15.2	THE DEVELOPMENT AT SAN HING
Drawing title PEDESTRIAN FLOW PATHS (SHEET 2 OF 2) Drawing No. 1 : 1000 (A1) 1 : 2000 (A3) Citil Engineering and Development Department ELACK & VEATCH HONG KONG LIMITED 博威工程颜間有限公司		ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY
Bit Strate Current of the strat Current of the strate	16.1	Drawing title
Based of the set of the	6,2 CUL II IIII IIII IIIII IIIIIIIIIIIIIIII	
PATHS (SHEET 2 OF 2) Drowing No. 198172/B&V/FR/0502 1 : 1000 (A1) 1 : 2000 (A3) 1 : 2000 (A3) 1 : 2000 (A3) 1 : 2000 (A3) 1 : 2000 (A3) Civil Engineering and Development Department BLACK & VEATCH HONG KONG LIMITED 博威工程顧問有限公司	H H	PEDESTRIAN FLOW
(SHEET 2 OF 2) Drawing No. 198172/B&V/FR/0502		PATHS
Best and an an an and an		(SHEET 2 OF 2)
198172/B&V/FR/0502 1 : 1000 (A1) 1 : 2000 (A3) 1 : 2000 (A3)		Drawing No. Scale
		108172/B&V//EP/0502
A 座 A E A E		$\begin{bmatrix} 130172/DXV/FN/0302 \\ 1 : 2000 (A3) \end{bmatrix}$
	H A座 Block A 6.6	
BLACK & VEATCH HONG KONG LIMITED 博威工程顧問有限公司		エホエ 柱 拓 炭 者 CFDD Civil Engineering and
5.8 b b b b b b c		Development Department
COL BLACK & VEATCH HONG KONG LIMITED 博威工程顧問有限公司		
BLACK & VEATCH HONG KONG LIMITED 博威工程顧問有限公司		
	CUL A	BLACK & VEATCH HONG KONG LIMITED 博威工程顧問有限公司

CAD Filename = C:\Daily Work - Y\20210706D\198172-B&V-FR-0601.dgn

	© Copyright	by Black	& Veatch	Hong Kon	g Limited
	LEGEND:				
		PROPOSED (SUBJECT	DEVELO TO REV	PMENT ARI 'IEW)	EA BOUNDARY
		EXISTING	DRILLH	OLE	
	-	EXISTING	TRIAL	PIT	
	4	BOREHOLE UNDER TH	CONDUC IS STUD	TED	
To Open Storege To Carden And Car					
	Revision Date Desig	jned Che)escript ecked	ion Drawn	Initial Checked
	Initial BF	. (CMW	SZ	JW
	Date 07/ Approved	18 07	/18	07/18	07/18
	Agreement No.				
5868/BH2 05502/D-71 35262/0D210/DH/032		CE 68	/2017	(CE)	
03408/13806/1170DT/ + 03408/73806/1109D	Project title	E FOR	ΜΑΤΙ	ON AN	ID
↓35261/DD210/D4/033	INFRAST	RUCT	URAL ⁄IENT	. WORI AT SA	KS FOR
135309/DD220/DH/023	ROAD AN MUN	ID HON	NG PC) ROAI	D, TUEN
	Drawing title				
	GRC		VES	TIGAT	ION
		KE(JUKL	5	
DO20 "	Drawing No.			c	cale
					• 2500 (11)
	198172/	/B&V/F	R/060)1	• 2500 (AT) • 5000 (A3)
		1			
	CEDD	土 木 Civil E	上 桯 ingine	拓展 ering a	者 and
		Develo	opme	nt Dep	artment
		ſ			
	BLACK &	VEATCH	HONG	® KONG	LIMITED
	博 展	或工程	顧問	有限公	一 司

<u>GEOLOGICAL PROFILE SECTION 1-1</u> SCALE: VERTICAL A1 1 : 400 HORIZONTAL A1 1 : 1000

<u>GEOLOGICAL PROFILE SECTION 2-2</u> SCALE: VERTICAL A1 1 : 400 HORIZONTAL A1 1 : 1000

TYPE OF ROCK: CDWPLETELY DECOMPOSED HOH HIGHLY DECOMPOSED ALL ALLIVIUM A ANDESTTE R RINDLITE FILL FILL MC METACONGLOWERATE WS METASLISTONE MA META-ANDESITE MT META-TUFF S SAUSTONE S SAUSTONE S SAUSTONE S SAUSTONE S STUFFACEOUS SILTSTONE T TUFF TUFFACEDUS SILTSTONE T TUFFACEDUS SILTSTONE T TUFFACEDUS SAUSTONE T TUFFITE T TUFFACEDUS SAUSTONE T TUFFACEDUS Dote	TYPE OF ROCK: CD CUMPLETELY DECOMPOSED HD HIGHLY DECOMPOSED AL ALLUTIM A ADDESTIE R RHTCARGOREATE NG METACARGOREATE SS SILISTONE SS SILISTONE TUFFACEOUS SALISTONE TUFFACEOUS SALISTONE TE TUFFACEOUS SANDETONE TS TUFFACEOUS SANDETONE ADPOONED SCE ADPOONED SCE METACARGOREAT SCE ADPOONED	TYPE OF ROCK: C2 COMPLETELY DECOMPOSED M3 MODENTELY DECOMPOSED M4 MODENTELY DECOMPOSED M4 MODENTELY DECOMPOSED M4 MODENTELY DECOMPOSED M4 MODENTELY M5 MODENTE M4 MADESITE M5 META-ANDESITE M5 META-ANDESITE M5 SAMOSTOLE S SAMOSTOLE S SAMOSTOLE S SAMOSTOLE S SULTSTOLE TUFF BECCIA TOFF ACEUS SALISTOVE TS TUFFACEUS SALISTOVE TE TOFACEUS ALISTO	© c	opyright by]	Black & Veatc	h Hong Kong	Limited
Revision Date Description Initial Designed Checked Drawn Checked Initial TW SC SZ NS Date 07/18 07/18 07/18 07/18 Approved E 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing title	Revision Date Description Initial Designed Checked Drawn Checked Initial TW SC SZ NS Date 07/18 07/18 07/18 07/18 Approved CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing title GEOLOGICAL PROFILE (SECTIONS 1-1 TO 2-2) Drawing No. Scole	Revision Date Description Initial Designed Crecked Drown Checked Initial TW SC SZ NS Date 07/18 07/18 07/18 07/18 Approved CE 68/2017 (CE) Project 1110 OFF	TYPE CD HD MD SD ALL A R FILL MC MS MA MT QV S SS T TB TSS TS TE	OF ROC COMPLE HIGHLY MODERA SLIGH ALLUV ANDES FILL METACO METAS META-A META- QUARTZ SANDS SILTS TUFF TUFF E TUFF AC TUFF AC	CK: ETELY DECOM DECOMPOSE ATELY DECOMPOSE TLY DECOMPOSE IUM ITE ITE DNGLOMERATE ILTSTONE ANDESITE FUFF Z VEIN FONE BRECCIA CEOUS SILTS CEOUS SANDS FE	MPOSED ED POSED SED	
Revision Date Description Initial Designed Checked Drawn Checked Initial TW SC SZ NS Date 07/18 07/18 07/18 07/18 Approved CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing title Drawing title	Revision Date Description Initial Designed Checked Drawn Checked Initial TW SC SZ NS Date 07/18 07/18 07/18 07/18 Approved CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing title GEOLOGICAL PROFILE (SECTIONS 1-1 TO 2-2) Drawing No. Scale	Revision Date Description Initial Designed Checked Drown Checked Initial TW SC SZ NS Date 07/18 07/18 07/18 07/18 Approved CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drowing title GEOLOGICAL PROFILE Drowing title GEOLOGICAL PROFILE Scale 198172/B&V/FR/0601a As SHOWN Image: Scale Civil Engineering and Development Department Scale					
Revision Date Description Initial Designed Checked Drawn Checked Initial TW SC SZ NS Date 07/18 07/18 07/18 07/18 Approved CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing title Drawing title Drawing title	Revision Date Description Initial Designed Checked Drawn Checked Initial TW SC SZ NS Date 07/18 07/18 07/18 07/18 Approved CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing title GEOLOGICAL PROFILE (SECTIONS 1-1 TO 2-2) Drawing No. Scale	Revision Date Description Initial Designed Checked Drawn Checked Initial TW SC SZ NS Date 07/18 07/18 07/18 07/18 Approved Agreement No. CE 68/2017 (CE) CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing title Drawing title GEOLOGICAL PROFILE Drawing No. Scale 198172/B&V/FR/0601a As SHOWN Civil Engineering and Development Department					
Agreement No. CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing title	Agreement No. CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing title GEOLOGICAL PROFILE (SECTIONS 1-1 TO 2-2) Drawing No. Scale	Agreement No. CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing title GEOLOGICAL PROFILE Drawing No. Scale 198172/B&V/FR/0601a Scale AS SHOWN Scale	Revision Initial Date Approv	Date Designed TW 07/18 ed	Descrip Checked SC 07/18	tion Drawn SZ 07/18	Initial Checked NS 07/18
	GEOLOGICAL PROFILE (SECTIONS 1-1 TO 2-2)	GEOLOGICAL PROFILE (SECTIONS 1-1 TO 2-2) Drawing No. Scale 198172/B&V/FR/0601a AS SHOWN Image: State of the state	Agreem Projec INFI THE ROA	ent No. Cl t title SITE F RASTRU DEVEL DEVEL DAND F MUN - F	E 68/201 FORMAT JCTURAI OPMENT HONG PO EASIBIL	7 (CE) ION AN L WORK T AT SA O ROAE ITY STU	D (S FOR N HING D, TUEN JDY

<u>GEOLOGICAL PROFILE SECTION 3-3</u>

SCALE: VERTICAL A1 1 : 400 HORIZONTAL A1 1 : 1000

<u>GEOLOGICAL PROFILE SECTION 3-3</u> SCALE: VERTICAL A1 1 : 400 HORIZONTAL A1 1 : 1000

TYPE CD HD SD ALL A R FILL MC MS MA MT QV S SS T TB TSS TS TE	OF RO	ICK: LETELY RATELY I HTLY DEC VIUM SITE LITE CONGLOMI SILTSTOI ANDESI TUFF TZ VEIN STONE BRECCI ACEOUS ITE	DECOMP DECOMPOSE DECOMPOSE ERATE NE TE	POSED DSED D DNE DNE	
Revision	Date Designed	De I Chee	escript cked	ion Drawn	Init Check
Initial Date	TW 07/18	S	C /18	SZ 07/18	NS 07/
Agreeme Project INFF THE ROA	nt No. C SITE SITE RASTR DEVEL D AND	E 68/ FORI UCTL LOPM	2017 VIATIO JRAL IENT G PC	(CE) ON AN WOR AT SA D ROA	ND KS FC AN HIN D, TUI
Drawing	GEOI			PROFI 3-3)	ILE
Drawing 109	172/R <i>8</i>	,\//FD	/0601	s b	Scale ∆S SHU™
CE		:木口 ivil Er evelo	L 程 ngine pmer	拓展 ering a nt Dep	署 and partme

NO

<u>Geological profile section 4-4</u> SCALE: VERTICAL A1 1 : 400 HORIZONTAL A1 1 : 1000

© Copyright by Black & Veatch Hong Kong Limited	1
TYPE OF ROCK:CDCOMPLETELY DECOMPOSEDHDHIGHLY DECOMPOSEDMDMODERATELY DECOMPOSEDSDSLIGHTLY DECOMPOSEDALLALLUVIUMAANDESITERRHYOLITEFILLFILLMCMETACONGLOMERATEMSMETASILTSTONEMAMETA-TUFFQVQUARTZ VEINSSANDSTONESSSILTSTONETTUFFTBTUFF BRECCIATSSTUFFACEOUS SILTSTONETETUFFITE	
Revision Date Description Init Designed Checked Drawn Checked Initial TW SC SZ Nt Date 07/18 07/18 07/ Approved	ial ked S '18
Agreement No. CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FO THE DEVELOPMENT AT SAN HIN ROAD AND HONG PO ROAD, TU MUN - FEASIBILITY STUDY Drawing title	DR NG EN
GEOLOGICAL PROFILE (SECTION 4-4)Drawing No.Scale198172/B&V/FR/0601cAS SHOW	WN

CAD Filename = Y:\Daily Work\20210602\198172-B&V-FR-0602.dgn

	© Copyright by Black & Veatch Hong Kong Limited
	LEGEND:
	PROPOSED DEVELOPMENT AREA BOUNDARY (SUBJECT TO REVIEW)
	OF HONG PO ROAD SITE
	OF SAN HING ROAD SITE EXTENSION
	BOUNDARY OF HOUSING DEVELOPMENT OF SAN HING ROAD SITE
	2 - .
SITE IN STREET	
	Revision Date Description Initial
	Designed Checked Drawn Checked
	Date 12/19 12/19 12/19 12/19
	Approved
	Agreement No. CE 68/2017 (CE)
	Project title
	SITE FORMATION AND
	THE DEVELOPMENT AT SAN HING
	ROAD AND HONG PO ROAD, TUEN
	MUN - FEASIBILITY STUDY
	Drawing title
	ASSESSMENTS
	Drawing No. Scale
	198172/B&V/FR/0603
	CEDD Civil Engineering and
	Development Department
	BLACK & VEATCH HONG KONG LIMITED

66.6 + · · · · · · · · · · · · · · · · · ·	© Copyright by Black & Veatch Hong Kong Limited
60 49	LEGEND.
G	PROPOSED DEVELOPMENT
	AREA BOUNDARY
	SITE AREA OF HONG PO ROAD SITE
	CATCHMENT
	ROCK OUTCROP
Grave	CHANNELLIZED DEBRIS FLOW
	OPEN HILLSLOPE LANDSLIDE
	PROPOSED FRESH WATER SERVICE RESERVOIR (INDICATIVE)
	PROPOSED SALT WATER SERVICE RESERVOIR (INDICATIVE)
	+7.6 PROPOSED FORMATION LEVEL
50 50 50 50 50 50 50 50 50 50 50 50 50 5	PROPOSED RETAINING STRUCTURE
	(INDICATIVE)
ONG PO ROAD SITE	PROPOSED RIGID BARRIER
	(INDICATIVE ONLY)
	PROPOSED MAINTENANCE ACCESS OF RIGID BARRIER (INDICATIVE ONLY)
23.7	
25.1, + - - - - - - - - - - - - - - - - - -	
R	
39.1 *	
	Revision Date Description Initial Designed Checked Drawn Checked
TS	Initial JLI YKL SZ JW
TS	Date 11/19 11/19 11/19
* ^{31,3} + ^{34,0}	Agreement No.
	CE 66/2017 (CE)
▼ 332 # 26.	SITE FORMATION AND
	INFRASTRUCTURAL WORKS FOR
	THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD. TUEN
	MUN - FEASIBILITY STUDY
基地 Graves	Drawing title
E TS	
Burne Ts	PROPOSED HAZARD
70 225.8 225.8 4	MITIGATION STRATEGY
23.3	Drawing No. Scale
	198172/B&V/FR/0604a
22.9	土木工程拓展署 CEDD Civil Engineering and
	Development Department

•

PROPOSED HAZARD MITIGATION MEASURES CROSS-SECTION

	opyright	by Blac	ck & Vea	tch Hong	Kong	Limited
Revision	Date		Descr	iption		Initial
	Desigr	ned	Checkec	l Dro	IWN	Checked
Initial Date	AN 10/19	9	YLC 10/19	10/	5Z 19	YLC 10/19
Approve	ed and					
Agreeme	ent No.					
<u> </u>		CE 6	68/20 ⁻	17 (CE	E)	
Project	t title			TION	A N 15	<u> </u>
						J
INFF	RAST	RUC	IUR		JRK	S FOR
INFF THE ROA	RAST DEVI D AN	RUC ELOI D H(PMEN DNG		JRK SAI DAD	S FOR N HING
INFF THE ROA	RAST DEVI D AN MUN	RUC ELOI D H(- FE/	PMEN DNG ASIBI	NT AT PO RO LITY S	DRK SAI DAD STU	S FOR N HING , TUEN DY
INFF THE ROA M	SITI RAST DEVI D AN MUN	RUC ELO D HO - FE	PMEN DNG ASIBI	NT AT PO RC LITY S	DRK SAN DAD STU	S FOR N HING , TUEN DY
INFF THE ROA I	SITI RAST DEVI D AN MUN	RUC ELO D HC - FE	PMEN DNG ASIBI	NT AT PO RO LITY S		S FOR N HING , TUEN DY
INFF THE ROA I	SITI RAST DEVI D AN MUN TITI	RUC ELO D H(- FE	DNG ASIBI	NT AT PO RC LITY S	SAI SAI DAD STU ARD	S FOR N HING , TUEN DY ES
INFF THE ROA I	SITI RAST DEVI D AN MUN T title PF MITI	RUC ELO D H(- FE/ GAT CRO	DNG ASIBI	D HAZ	ARD URE DAD	S FOR N HING , TUEN DY
INFF THE ROA I	SIT RAST DEVI D AN MUN T title PF MITI		DNG ASIBI	D HAZ		S FOR N HING , TUEN DY
INFF THE ROA I	SIT RAST DEVI D AN MUN T title PF MITI (RUC ELO D H(- FE, GAT CRO	DSEE	D HAZ		S FOR N HING , TUEN DY
INFF THE ROA I Drawing	SITI RAST DEVI D AN MUN T title PF MITI (T T72/E	RUC ELO D H(- FE, GAT CRO	PMEN DNG ASIBI DSED DSED SS-S	D HAZ		S FOR N HING N TUEN DY
INFF THE ROA I Drawing	SITI RAST DEVI D AN MUN T title PF MITI (7 No.		PMEN DNG ASIBI DSED TON I SS-S	D HAZ		S FOR N HING , TUEN DY S S
INFF THE ROA I Drawing	SITE RAST DEVI D AN MUN T title PF MITE (T T2/E	RUC ELO D H(- FE, GAT CRO	FR/06	NT AT PO RC LITY : D HAZ MEAS ECTIC		S FOR N HING N, TUEN DY
INFF THE ROA I Drawing Drawing 198	SITI RAST DEVI D AN MUN T title PF MITI (T72/E	RUC ELO D H(- FE, GAT CRO 3&V/I 3&V/I ± *		NT AT PO RO LITY S D HAZ MEAS ECTIO		S FOR N HING , TUEN DY S S S S S S S S S S S S S S S S S S
INFF THE ROA Drawing Drawing 198	AST DEVI D AN MUN T title PF MITI (No. 172/E	RUC ELO D H(- FE, GAT CRO 3&V/I 3&V/I ± オ Civil Dev	FR/06	NT AT PO R(LITY S D HAZ) MEAS ECTIO		S FOR N HING , TUEN DY
INFF THE ROA I Drawing 198	AST DEVI D AN MUN T title PF MITI (7 No. 172/E	RUC ELO D H(- FE, GAT CRO 3&V/I 3&V/I ± オ Civil Dev		NT AT PO RO LITY S D HAZ MEAS ECTIO		S FOR N HING N TUEN DY
INFF ROA Drawing 198	SITI RAST DEVI D AN MUN g title PF MITI (No. 172/E			NT AT PO RO LITY S D HAZ MEAS ECTIO		S FOR N HING N TUEN DY S S S S S S S S S S S S S S S S S S

CAD Filename = c:\daily work - y\20210706d\198172-B&V-FR-0701.dgn

Plot Date = 1/29/2021

	C Copyright by Black & Veatch Hong Kong Limited
	NOTE: 1. THIS DRAWING SHOULD BE READ IN CONJUNCTION WITH DRAWING NO. 198172/B&V/SFA/008. LEGEND: EXISTING GROUND LEVEL PROPOSED SITE FORMATION LEVEL
m 56 58 60 62 64	RevisionDateDescriptionInitialDesignedCheckedDrawnCheckedInitialJLIYKLSZJWDate07/2007/2007/2007/20Approved
	Agreement No. CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN
	Drawing No. Scale
m 6	198172/B&V/FR/0701a 1 :: 200 (A1) 1 :: 400 (A3) 1 :: 400 (A3)

Plot Date = 1/29/202

		© Copyright by Black & Veatch Hong Kong Limited
		NOTE: 1. THIS DRAWING SHOULD BE READ IN CONJUNCTION WITH DRAWING NO. 198172/B&V/SFA/008. LEGEND:
		EXISTING GROUND LEVEL
mPD Z		PROPOSED SITE FORMATION LEVEL
m 45 50		
PROPOSED ROAD		
• • • • • • • • • • • • • • • • • • •		
	— m	
65 70 75 8	80	
		Revision Date Description Initial
		DesignedCheckedDrawnCheckedInitialJLICMWSZJWDate07/2007/2007/2007/20
		Approved
		Agreement No. CE 68/2017 (CE)
		Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR
		THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY
		Drawing title
		CROSS-SECTION OF SITE FORMATION PLAN
		(SHEET 2 OF 4) Drawing No. Scale
		198172/B&V/FR/0701b AS SHOWN
		土木工程拓展署 Civil Engineering and Development Department
		EEE。 BLACK & VEATCH HONG KONG LIMITED 博威工程顧問有限公司

•

Plot Date = 1/29/2021

CAD Filename = Y:\Daily Work\20210118C\198172-B&V-FR-0701c.dgn

W I	HIS DRAWING ITH DRAWING	SHOULD BE NO: 198172	READ IN COM /B&V/SFA/OC	IJUNCTION)8.
LEG	END: fxis	STING GROUN	DIEVEI	
	PROF	POSED SITE	FORMATION L	EVEL
Revision	Date	Descrip	tion	Initia
Initial	JLI	YLK	SZ	JW
Date Approve	07/20 ed	07/20	07/20	07/20
	ent No.			
Agreem		60/004-	7 (CE)	
Agreem	CE	: 68/2017	(
Agreem Projec	CE + +i+Ie SITE F)
Agreem Project	CE + +i+Ie SITE F RASTRU DEVELO	FORMAT	ΊΟΝ ΑΝΙ L WORK) S FOR N HINC
Agreem Project INFE THE ROA	CE SITE F RASTRU DEVELO D AND F MUN - F	FORMAT ICTURA OPMENT HONG P EASIBIL	ΊΟΝ ΑΝΙ L WORK Γ ΑΤ SAI Ο ROAD ITY STU	D S FOR N HINC , TUEI DY
Agreem Project INFF THE ROA	CE SITE F RASTRU DEVELO DAND F MUN - F	FORMAT ICTURA OPMEN HONG P EASIBIL	ΊΟΝ ΑΝΙ L WORK Γ ΑΤ SAI Ο ROAD ITY STU) S FOR N HINC , TUEI DY
Agreem Project INFE THE ROA	CE SITE F RASTRU DEVELO D AND F MUN - F	-SECTIC	ION ANI L WORK T AT SAI O ROAD ITY STU) S FOR N HINC , TUEI DY
Agreem Projec INFF THE ROA	CE SITE F RASTRU DEVELO DAND F MUN - F g title CROSS FOF	-SECTIC	ION ANI L WORK T AT SAN O ROAD ITY STU) S FOR N HINC , TUEI DY
Agreem Projec INFF THE ROA	CE SITE F RASTRU DEVELO D AND F MUN - F g title CROSS FOF	-SECTIC	ION ANI L WORK T AT SAI O ROAD ITY STU	D S FOR N HINC , TUEI DY
Agreem Projec INFF THE ROA Drawing	CE SITE F RASTRU DEVELO DAND F MUN - F g title CROSS FOF	-SECTIC	ION ANI L WORK T AT SAI O ROAD ITY STU	D S FOR N HINC D TUE DY
Agreem Projec INFF THE ROA Drawing	CE SITE F RASTRU DEVELO DAND F MUN - F g title CROSS FOF	-SECTIC RMATIOI	ION ANI L WORK T AT SAI O ROAD ITY STU	D S FOR N HINC , TUEN DY
Agreem Project INFF THE ROA Drawing Drawing	CE SITE F RASTRU DEVELO D AND F MUN - F g title CROSS FOF	-SECTIC RMATIOI	ION ANI L WORK T AT SAI O ROAD ITY STU	D S FOR N HINC , TUEN DY
Agreem Projec INFI THE ROA Drawing Drawing 198	CE SITE F RASTRU DEVELO DAND F MUN - F g title CROSS FOF	· ORMAT ICTURA OPMEN HONG P EASIBIL -SECTIC RMATION //FR/070	ION ANI L WORK T AT SAI O ROAD ITY STU ON OF SI N PLAN (SHE Soci 1C	D S FOR N HINC DY ITE ITE
Agreem Projec INFI THE ROA Drawing Drawing 198	CE SITE F RASTRU DEVELO DAND F MUN - F g title CROSS FOF g No. 3172/B&\	· ORMAT CTURA OPMEN HONG P EASIBIL -SECTIC RMATION //FR/070 木工程 vil Engin	ION ANI L WORK T AT SAI O ROAD ITY STU ON OF SI N PLAN (SHE (SHE) (SHE) Soci 1c	S FOR N HINC , TUEN DY ITE ET 3 OF ITe AS SHOWN

CAD Filename = Y:\Daily Work\20210118C\198172-B&V-FR-0701d.dgn

	© Copyright by Black & Veatch Hong Kong Limited
	NOTE: 1. THIS DRAWING SHOULD BE READ IN CONJUNCTION WITH DRAWING NO. 198172/B&V/SFA/008.
	EXISTING GROUND LEVEL PROPOSED SITE FORMATION LEVEL
SAN HING TSUEN +10.30mPD	
	Revision Date Description Initial Designed Checked Drawn Checked Initial JLI YKL SZ JW
OL SITE	Date 07/20 07/20 07/20 07/20 Approved
	Agreement No. CE 68/2017 (CE)
+8.10mPD	Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY
	CROSS-SECTION OF SITE FORMATION PLAN
	(SHEET 4 OF 4)
	Drawing No. Scale 198172/B&V/FR/0701d AS SHOWN
	EEE。 BLACK & VEATCH HONG KONG LIMITED 博威工程顧問有限公司

	(©) Copyright by Black & Veatch Hong Kong Limited LEGEND: PROPOSED DEVELOPMENT AREA BOUNDARY (SUBJECT TO REVIEW) POTENTIAL SCHOOL SITE (SUBJECT TO REVIEW) PROPOSED HOUSING SITE BOUNDARY (SUBJECT TO REVIEW) PROPOSED U-CHANNEL AND PROPOSED CATCH PIT (INDICATIVE) (TO BE MAINTAINED BY ARCHSD) PROPOSED U-CHANNEL AND PROPOSED CATCH PIT (INDICATIVE) (TO BE MAINTAINED BY ARCHSD) PROPOSED U-CHANNEL AND PROPOSED CATCH PIT (INDICATIVE) (TO BE MAINTAINED BY HD) EXISTING STREAM EXISTING STREAM PROPOSED DRAINAGE RESERVE AREA (INDICATIVE ONLY) PROPOSED BOX CULVERT (INDICATIVE) (TO BE MAINTAINED BY DSD) PROPOSED DRAINAGE PIPE AND MANHOLE (INDICATIVE) PROPOSED CATCH PIT (INDICATIVE) PROPOSED CATCH PIT (INDICATIVE) PROPOSED CATCH PIT (INDICATIVE) PROPOSED CATCH PIT (INDICATIVE) PROPOSED SLOPE (INDICATIVE) PROPOSED SLOPE (INDICATIVE) PROPOSED INTAKE FOR EXISTING STREAM
PROPOSED 1050 DIA: DRAINAGE / PIPE WIJH GRADJENT 1:100 (TO BE MAINTAINED BY DSD)	RevisionDateDescriptionInitialDesignedCheckedDrawnCheckedInitialJLIYKLSZJWDate03/2003/2003/2003/20Approved
	CE 68/2017 (CE) Project title SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY Drawing title PROPOSED DRAINAGE LAYOUT PLAN
	Drawing No. Scale 198172/B&V/FR/0802 1 :: 2500 (A1 1 :: 5000 (A3) ごごご 上木工程拓展署 Civil Engineering and Development Department
	BLACK & VEATCH HONG KONG LIMITED 博威工程顧問有限公司

	© Copyright by Black & Veatch Hong Kong Limited
	LEGEND:
	PROPOSED DEVELOPMENT AREA BOUNDARY
	(SUBJECT TO REVIEW)
	POTENTIAL SCHOOL SITE (SUBJECT TO REVIEW)
	CATCHMENT AREA (M2)
	PROPOSED DRAINAGE PIPE AND MANHOLES
	PROPOSED CATCH PIT (INDICATIVE)
	PROPOSED INTAKE FOR EXISTING STREAM
In the second se	
	Revision Date Description Initial
	Initial JLI YKL SZ JW
	Date 03/20 03/20 03/20 03/20
	Approved
	Agreement No.
	CE 68/2017 (CE)
	Project title
	INFRASTRUCTURAL WORKS FOR
	THE DEVELOPMENT AT SAN HING
	MUN - FEASIBILITY STUDY
	Drawing title
	PROPOSED U-CHANNEL
	CATCHMENT PLAN
Patrice (Cartering) Patrice (
	Drawing No. Scale
	1981/2/B&V/FR/0803
	<u></u> 土木工程拓展署
	CEDD Civil Engineering and Development Department
	BLACK & VEATCH HONG KONG LIMITED
HUBBRING HUBBRI	博威工程顧問有限公司

CAD Filename = c:\daily work - y\20210706d\198172-B&V-FR-0901.dgn

	© Copyright by Black & Veatch Hong Kong Limited
	DOTENTIAL SOUDOL STTE
	(SUBJECT TO REVIEW)
	PROPOSED DEVELOPMENT AREA (PDA) BOUNDARY (SUBJECT TO REVIEW)
	+7.0 (INDICATIVE)
	PROPOSED RETAINING STRUCTURE
	FROFOSED SLOFE (INDICATIVE)
	-RM-RM- FOR SEWAGE PUMPING STATION
	PROPOSED GRAVITY SEWER (INDICATIVE)
	INDICATIVE POINT OF DISCHARGE
	EXISTING SEWERS TO BE UPGRADED
	(INDICATIVE)
	Revision Date Description Initial
	Initial III YKI SZ IW
	Date 01/20 01/20 01/20 01/20
	Approved
	Agreement No.
	CE DO/ZUI/ (CE)
	SITE FORMATION AND
	INFRASTRUCTURAL WORKS FOR
	THE DEVELOPMENT AT SAN HING
	MUN - FEASIBILITY STUDY
	Drawing title
	PROPOSED SEWERAGE
	LAYOUT PLAN
Pedan (% Tel luest oser)	
	Drawing No. Scale
	198172/B&\//EP/0001 1:2500 (A1)
CRCATERHEAD	
	< 土 木 工 程 拓 展 署
	CEDD Civil Engineering and
	Development Department
	BLACK & VEATCH HONG KONG LIMITED
	四 一 一 一 一 一 一 一 一 一 一 一 一 一 一 一 一 一 一 一

			© Co	pyright by H	Black & Veatc	h Hong Kong	Limited
				<u>\</u> D•			
					POSED DEVEL	NPMENT AREA	BOUNDARY
			L	(SUE	BJECT TO RE	VIEW)	
				PROF SER'	POSED FRESH VICE RESERV	WATER DIR (INDICA	TIVE)
			-0	o— EXIS	STING FRESH	WATER MAIN	
			0	o— Prof	POSED FRESH	WATER MAIN	
				(INI)ICATIVE)		
ONNECTED TO	THE						
RELOCATED TKT	SFW S/R						
	- 2 e						
	i i La						
			Revision	Date	Descrip	tion	Initial
				Designed	Checked	Drawn	Checked
			Initial	JLI	YKL	SZ	JW
			Date Approve	11/19 ed	11/19	11/19	11/19
				CE	68/2017	7 (CE)	
			Project	SITE F	FORMAT		D
		2		RASTRU	JCTURA	WORK	S FOR
			THE ROA	DEVEL D AND I	OPMEN HONG P	I AT SAF O ROAD	N HING . TUEN
			ſ	MUN - F	EASIBIL	ITY STU	DY
			Drawing	; title			
			F	ROPOS		SH WAT	TER
				SUPPL	I ARRA	NGEMEI	
			Drawing	No.		Sco	ı l e
			100				8000 (A1)
			198	172/B&\	//FK/100		16000 (A3)
				< ±	木工程	拓展署	
					vil Engin	eering ar	nd rtment
ANNU L'ARCONNERTAINN ANNU A' A' ANNU ANNU A' A' ANNU ANNU						ni Deha	
MENSION (m)	SERVICE F	RESERVOIR					
19	vuluml (r		RI A	CK & VF	ATCH HOM	RUNC I	
19	6	916		博威工	程 顧 問	有限公	司

CAD Filename = C:\Daily Work - Y\20210706D\198172-B&V-FR-1003a.dgn

	${\mathbb C}$ Copyright by Black & Veatch Hong Kong Limited
	LEGEND:
	PROPOSED DEVELOPMENT AREA BOUNDARY
	DOTENTIAL SCHOOL SITE
Image Image <th< th=""><th>(SUBJECT TO REVIEW)</th></th<>	(SUBJECT TO REVIEW)
	FW
	(INDICATIVE) PROPOSED FRESH WATER MAINS FROM
	TKT SOUTH S/R TO SHR SITE (INDICATIVE)
YRC (YERDER) a become yer become the formation of the	PROPOSED FRESH WATER
C AND	PROPOSED SALT WATER
	SERVICE RESERVOIR (INDICATIVE)
	+7.6 (INDICATIVE)
	PRUPUSED RETAINING STRUCTURE (INDICATIVE)
	PROPOSED SLOPE (INDICATIVE)
No. 10 THE	
A A BB Two A C A A A A A A A A A A A A A A A A A	
Refit Filter Bad Refit Filter Bad	
Store Read	
	Revision Date Description Initial Designed Checked Drawn Checked
	Initial JLI YKL SZ JW
	Date 07/20 07/20 07/20 Approved
THP Hore 2017	
	Agreement No.
WP Her 2017	Project title
	SITE FORMATION AND
Grandeer Graden	INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING
1. (1980) 1. (1990) 1. (1990)	ROAD AND HONG PO ROAD, TUEN
Real Hard Control Land Control	MUN - FEASIBILITY STUDY
	Urawing title
Ball-store Cur Pool	PROPOSED FRESH WATER
	MAINS LAYOUT PLAN
	(SHEET 2 OF 3)
Ringston Terros Kingston Terros	198172/B&V/FR/1003b
[Pedim Cerl Phi asser]	
stal stal	<u></u> 土木工程拓展署
	CEDD Civil Engineering and Development Department
	R,
	BLACK & VEATCH HONG KONG LIMITED
	博威工程顧問有限公司


CAD Filename = Y:\Daily Work\20210310C\198172-B&V-FR-1003c.dgn

	C Copyright by	Black & Veatch	Hong Kon	g Limited
	LEGEND:			
		EXISTING FRE	SH WATER	MAIN
60				
80				
100				
$ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\$				
× * </th <th></th> <th></th> <th></th> <th></th>				
Track				
105.0 ₊				
+103.2				
+101.3				
	Revision Date Designe	Descrip d Checked	Drawn	Initial Checked
97.5+	Initial .II I	YKI		
CUL		07/20	07/20	07/20
	Approved			
	Approved			
	Approved			
CUL CUL + CUL + 44.8 CUL + CUL	Approved			
	Approved	E 68/2017	(CE)	
	Approved C	E 68/2017	′ (CE)	
R CUL CUL CUL CUL CUL CUL CUL CUL	Approved C Project title SITE	E 68/2017 FORMAT	(CE)	١D
	Approved C Project title SITE INFRASTR	E 68/2017 FORMAT	(CE) ION AN WORI	ID KS FOR
	Approved Project title SITE INFRASTR THE DEVE	E 68/2017 FORMAT UCTURAL	(CE) ION AN WORI	ID KS FOR AN HING
	Approved C Project title SITE INFRASTR THE DEVE ROAD AND MUN -	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO	(CE) ION AN WORI AT SA D ROAI	ID KS FOR AN HING D, TUEN
	Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN -	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI	(CE) ION AN WORI AT SA D ROAI TY STU	ID KS FOR AN HING D, TUEN JDY
	Approved C Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI	(CE) ION AN WORI AT SA D ROAI TY STI	ID KS FOR AN HING D, TUEN JDY
	Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI	(CE) ION AN WORI AT SA D ROAI TY STU	ID KS FOR AN HING D, TUEN JDY
	Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title PROPC	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI	(CE) ION AN WORI AT SA D ROAI TY STI	ID KS FOR AN HING D, TUEN JDY
	Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title PROPO MAI	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI	(CE) ION AN WORI AT SA D ROAI TY STU	ID KS FOR AN HING D, TUEN JDY
	Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title PROPC MAI	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI	(CE) ION AN WORI AT SA D ROAI TY STU	ID KS FOR AN HING D, TUEN JDY
	Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title PROPC MAI	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI	(CE) ION AN WORI AT SA D ROAI TY STU	ID KS FOR AN HING D, TUEN JDY ATER ATER N
	Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title PROPC MAI	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI	(CE) ION AN WORI AT SA D ROAI TY STU	ID KS FOR AN HING D, TUEN JDY ATER N EET 3 OF 3)
	Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title PROPC MAI Drawing No.	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI	(CE) ION AN WORI AT SA D ROAI TY STU	ID KS FOR AN HING D, TUEN JDY ATER N EET 3 OF 3) cale
	Approved Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title PROPC MAI Drawing No. 198172/B8	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI	C (CE)	ID KS FOR AN HING D, TUEN JDY ATER N EET 3 OF 3) cale : 1000 (A1) : 2000 (A3)
	Approved Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title PROPC MAI Drawing No. 198172/B8	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI	(CE) ION AN WORI AT SA D ROAI TY STU SH WA JT PLA	ID KS FOR AN HING D, TUEN JDY ATER N EET 3 OF 3) cale : 1000 (A1) : 2000 (A3)
	Approved Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title PROPC MAI Drawing No. 198172/B8	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI	C (CE)	ID KS FOR AN HING D, TUEN JDY ATER N EET 3 OF 3) cale : 1000 (A1) : 2000 (A3)
	Approved Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title Drawing No. 198172/B8	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI DSED FRE NS LAYOU	Y (CE) ION AN WORI AT SA D ROAI TY STU SH WA JT PLA (SHI (SHI (SHI 3c 1	ID KS FOR AN HING D, TUEN JDY ATER N EET 3 OF 3) cale : 1000 (A1) : 2000 (A3)
	Approved Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title Drawing No. 198172/B8	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI DSED FRE NS LAYOU &V/FR/100	Y (CE) ION AN WORI AT SA D ROAI TY STU SH WA JT PLA (SHI (SHI (SHI (SHI 3c 1 3c 1	JD KS FOR AN HING D, TUEN JDY ATER N EET 3 OF 3) cale : 1000 (A1) : 2000 (A3)
	Approved Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title Drawing No. 198172/B8	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI OSED FRE NS LAYOU &V/FR/100	Y (CE) ION AN WORI AT SA D ROAI TY STU SH WA JT PLA (SHI (SHI (SHI (SHI 3C 1 3C 1 3C 1	JD KS FOR AN HING D, TUEN JDY ATER ATER N EET 3 OF 3) cale : 1000 (A1) : 2000 (A3)
	Approved Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title PROPC MAI Drawing No. 198172/B8	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI DSED FRE NS LAYOU &V/FR/100	ION AN ION AN WORI AT SA D ROAI TY STU SH WA JT PLA (SHI) (SHI) 3c 1 AT R eering a int Depart	JD KS FOR AN HING D, TUEN JDY ATER N EET 3 OF 3) cale : 1000 (A1) : 2000 (A3) Cale
	Approved Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title Drawing No. 198172/B8	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI DSED FRE NS LAYOU &V/FR/100	ION AN ION AN WORI AT SA O ROAI TY STI SH WA JT PLA (SHI (SHI 3c 新 展 eering a Int Dep	JD KS FOR AN HING D, TUEN JDY ATER N EET 3 OF 3) cale : 1000 (A1) : 2000 (A3) # # and artment
	Approved Approved Project title SITE INFRASTR THE DEVE ROAD AND MUN - Drawing title PROPC MAI Drawing No. 198172/B8 BLACK & V	E 68/2017 FORMAT UCTURAL LOPMENT HONG PO FEASIBILI DSED FRE NS LAYOU EXV/FR/100	ION AN ION AN WORI AT SA NOAI TY STI SH WA TY STI SH WA SH WA	JD KS FOR AN HING D, TUEN JDY ATER M EET 3 OF 3) cale : 1000 (A1) : 2000 (A3) Cale : 2000 (A3) Cale



1000 B			HIGHWAY	© Co	pvright	by Bla	ack & Veatch	n Hong I	Kong Li	mited
000	a tere				, , , , , , , , , , , , , , , , , , , ,					
820				LEGE	ND:					
				[PROPO (SUBJ	SED DEVELO ECT TO REV	DPMENT /IEW)	AREA E	BOUNDARY
						PROPO	ISED SALT V	VATER		
0000						SERVI	CERESERVO	DIR (IN	DICATI	VE)
_				-0	-0	EXIST	ING SALT W	VATER M	AIN	
				(Ø45	50)	PROPO	ISED SALT V	VATER M	AIN DI	AMETER
				.,						
25										
AR										
			Al Cha							
55	式 一									
			All of Colo							
			ALL BX							
	17 25									
SA		Seller -	COPE							
eur	VG KA	TSUEN								
	241637									
\geq										
N.										
F	VAS &		SATUS							
	JI A	AL AS	XING							
SA	SV - V									
\$ (()	KALL.	EDJK	ALL A							
NA										
	ACT									
Z II S	NPA 2									
	ARS		SAG							
		2-555								
524										
UK.		22775								
35			100555	Revision	Date		Descrip	tion		Initial
57	162	3 AS			Desig	ned	Checked	Draw	/n	Checked
		AN SI	PAV	Initial	JLI	Ι	YKL	SZ		JW
	57115	200	45AL	Date	07/2	20	07/20	07/2	:0	07/20
15				Approve	ed					
554										
\mathcal{P}	S MK									
X S				Agreeme	ent No.					
3A		205	20			CE	68/2017	' (CE))	
		ST.		Projec ⁻	t title					
	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	1972	S TAI LAM CT		SIT	E FO		ION A	1ND	
RY		MIL		INF TUC						
XX	22 25	V D		ROA						TUEN
5		) Ku	225		MUN	- FE	ASIBILI	TY S	TUD	Y
R	2 Fre	$\langle \rangle$	LYNN	Drawing	1 + 1 + 1 0					
$\sum$	26 ~ 5	0	55	UWLIQ	,					
Yo	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	680								
$\langle \rangle$	5				PRO	POS	SED SA		ATE	R   T
2					SUP	'PLY	ARRAI	NGEI	VIEN	1
$\int$		Ter Liesconfeedonal		Drawing	] No.				Scale	e
S7	THE ARE DESCRIPTION			100	4 7 0 / Г	ע מר		4	1 • 1	5000 (41)
			Co Vinta	198	172/6	3&V/	/FR/100	4	1:30	0000 (A3)
		0.50 ¹⁰ 大概源村								
		A State Chung				۔ ۔ بر		⊥ <del>,</del> ⊑		
	3 00 3 0 Ma					土 7 〇 ^二	ҡ⊥樦 ӹ╘∽∽ ^៲ ┈	11月 月	€ 者	4
日本 日 日 日 日 日 日 日 日 日 日 日 日 日	Revenue		37))/C		:UU		ıı ⊏rigine /elonme	et De	y an( ynart	nent
						561	Sightine		γαιι	
NO?	#1 ~ () ()除(()勝安)	1 ( ) ) ( ) ( ) ( ) ( ) ( ) ( ) ( ) ( )								
		SEBNICE	RECERVATE				Ŗ			
MEINS	SION (m)	VOLUME (	$M^{3}$ )	BLA	.CK &	VEA	TCH HONG	® G Kon(	G LIM	MITED
, 5			300		博威	ÌΙ	程顧問	有 限	公司	J
										1



		© Copyrigh	t by Black & Veatcl	h Hong Kong Limited
IENSION (m)	SERVICE RESERVOIR VOLUME (m³)	LEGEND:		
5	1300		PROPOSED DEVELO (SUBJECT TO RE)	OPMENT AREA BOUNDARY VIEW)
		; <u>-</u>	POTENTIAL SCHOU (SUBJECT TO RE)	DL SITE VIEW)
			PROPOSED SALT	WATER MAINS FROM
		5	EXTENSION AND	SHR SITE, SHR SITE SHR SITE (INDICATIVE)
		—SW —	PROPOSED SALT FROM TUEN MUN ( (INDICATIVE)	WATER MAINS S.W. P/S TO HPRSW S/R
			PROPOSED FRESH	WATER DIR (INDICATIVE)
			PROPOSED SALT	WATER
		+7.6	PROPOSED FORMA	TION LEVEL
Markey Poles			(INDICATIVE) PROPOSED_RETAIL	NING STRUCTURE
101. 74 00 90% 73		<b>*</b>	PROPOSED SLOPE	(INDICATIVE)
	ь. г. Ту	·		
	a da kana da ka			
10 10 10 10 10 10 10 10 10 10 10 10 10 1				
	n. n. have a second sec			
	13 Optient Van 14 Optient Van 15 Optient Van 15 Optient Van 16 Optient Van 17 Optient Van 18 Optient Van 19 Optient Van			
		Revision Date	B Descrip	tion Initial
		Initial J	_ I YKL	SZ JW
6. Bar		Date 07, Approved	/20 07/20	07/20 07/20
14 - 14 - 14 - 14 - 14 - 14 - 14 - 14 -				
		Agreement No	CE 68/2017	7 (CE)
		Project titl		
	·····································	SI ⁻ INFRAS	TE FORMAT TRUCTURAI	ION AND L WORKS FOR
		THE DE	/ELOPMENT	F AT SAN HING
		MUN	I - FEASIBIL	ITY STUDY
		Drawing titl	е	
		PRO		I T WATER
		M	AINS LAYO	UT PLAN
	COL 0 Peden (* TellSelle) Peden (* TellSelle)			
		Drawing No.		(SHEET 1 OF 2)
La	High         High <td< th=""><th>100170</th><th></th><th><b>1</b> : 2500 (A1)</th></td<>	100170		<b>1</b> : 2500 (A1)
		198172/	B&V/FR/100	
UDE-CARDARDARDA				
		CFDD	エホエ程 Civil Engine	· 珀
			Developme	ent Department
10 10 10 10 10 10 10 10 10 10				® KONC I IMITED
HIGH REAL CONTROL OF THE CONTROL OF			× v laiten ⊓unt 或 工 程 顧 問	有限公司



	C Copyright by Black & Veatch Hong Kong Limited
	LEGEND:
Top Pr Rede Computy Graden	
	(INDICATIVE)
	4
A HI	
	Š
A B C C C C C C C C C C C C C C C C C C	-
1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1	
V C + 4	
	1
Benefit and a second and a se	
	Revision Date Description Initial
	Designed Checked Drawn Checked
	Initial JLI YKL SZ JW
A Martin and a martine and a mar	Date 07/20 07/20 07/20 07/20
1000 000	Approved
	9
Benjara	
	Agreement No.
Pedum         0.3         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4         0.4 </th <th>CE 68/2017 (CE)</th>	CE 68/2017 (CE)
	Project title
A B Construction of the second	
Real for the second sec	MUN - FEASIBILITY STUDY
	MON - TEASIBLETTI STODT
	Drawing title
Base         Patient         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1	
Here are there is a set of the se	PROPOSED SALT WATER
	MAINS LAYOUT PLAN
The second secon	
「 「 「 」 「 」 「 」 「 」 「 」 「 」 」 」 「 」 」 」 「 」 」 」 「 」 「 」 」 」 」 」 」 」 」 」 」 」 」 」	
	Scale Scale
and the second sec	109170/D9\//ED/40066 1:2500 (A1)
With the part of t	
	📄 🧹 🗲 土 木 工 程 拓 展 署
Image: State of the s	<b>CEDD</b> Civil Engineering and
	Development Department
VILLAGE VILLAGE TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL	
and	」 DLAUK & VEATUR HUNG KUNG LIMITED 博威工程顧問有限公司





		×	© Co	pyright by Black	< & Veatch Hon	q Kong Limited	
	· _	ATT					
	+ 19.2 N	) (e ^k ^y )	LEGEI	ND:			
] BIOCK ]	74		- C	PROPOSEI	D DEVELOPMEN	NT AREA BOU	NDARY
скон			<u> </u>	PROPOSEI	D JUNCTION IN	PROVEMENT	WORKS
	\ _E_5	Block		PROPOSEI	D JUNCTION IN	PROVEMENT	
				TUEN MU	N AREA 54 R	DAD ALIGNMEI	NT
		$\mathbb{Z}$		2.5m VEF	TICAL BARRIE	R	
				5.5m VEB	TICAL BARRIE	R WITH 2.5m	CANTILEVER
	///	17.8		AT 45'			
	12			5.5m VER AT 45' (A	TICAL BARRIE	R WITH 2.5m	CANTILEVER
	/			5.5m VER	TICAL BARRIE	R WITH 3.5m	CANTILEVER
$\overline{}$		/		AT 45 (A	BSORPTIVE)		
E L				6.5m VER AT 45' (A	TICAL BARRIE	R WITH 3.0m	CANTILEVER
$\square$				2.0m CON	CRETE WALL		
TS S				LOW NOIS	E ROAD SURF	ACING	
$\searrow$				ACOUSTIC	WINDOWS		
/	11.5			PROHIBITE	D FOR CLINIC	S, CONVALES	CENCES AND
				WARDS U	OR THE AGED SES WHICH RE FOR VENTILAT	DIAGNOSTIC LY ON OPEN TION	ROOMS AND ABLE
ĺ.				PROHIBITE			CENCES
				HOMES FO	OR THE AGED	DIAGNOSTIC	ROOMS AND
'V		LANTER WALL		INCLUDING RELY ON	; KINDERGART OPENABLE WI	EN AND NURS NDOWS FOR V	ENTILATION
	W			FULLY EN	CLOSED WALK	WAY	
LA				3.5m (H)	x 4.0 (W) OF	PENING	
	-						
$\sim$	2	A					
	$\square$						
TA							
			NOTE:	YOUT IS FOR		AND SUBJEC	1 10
			REVIEW	AND CHANGE		1110 000000	
			Revision	Designed	Checked	Drawn	Checked
/			Initial			KL	кү
$\mathbb{Z}$ / /			Date			07/20	07/20
) / _	+		Approve	1			
	1						
$\top$							
			Agreeme		20/2047		
· /				UE	00/2017	(UE)	
			Project	title	0.000		
				SILEF			
			THE	DEVEL	OPMEN	L AT SA	N HING
			ROA	D AND H	IONG P	O ROAD	, TUEN
			l	MUN - F	EASIBIL	ITY STU	DY
			Drawing	title			
			L		NS OF C	ONSID	ERED
						EASUR	E ES
						_,	
						(SHEE	T 2 OF 2)
			Drawing	No.		Sco	le
			100	170/00	//ED/400	16	
			198	0172/B&	v/FR/130		
				<del>~</del> +	木工程	拓展爭	2
					/il Enaine	eerina a	nd
				De	velopme	nt Depa	rtment
						-	
n 25m	50	m			Ð		
12.2					V'		
+		L I	BLA	CK & VEA 博感丁	TCH HONO 程 顧 問	GKONGLI	MITED ㅋ
	1			1			-



CAD Filename = Y:\Daily Work\20200915J\FIGURE 8.4.dgn

POTENTIAL RECEPTOR SITE FOR TRANSPLANTED	© Copy	right by Black	k & Veatch Hong Ko	ng Limited	
AQUILARIA SINENSIS POTENTIAL RECEPTOR SITES FOR TRANSLOCATING	LEGEN	↓ [] ● ●			
SOMANNIATHELPUSA ZANKLON		PR	OPOSED DEVEL	OPMENT ARE	BOUNDARY
ONSITE SECTION OF STREAM TO BE RETAINED BY AVOIDANCE MEASURE		500	)m ASSESSMEN1	F AREA	
AND ENHANCEMENT		PR	DJECT SITE BO	DUNDARY	
PROPOSED WOODLAND ENHANCEMENT PLANTING		BOI	JNDARY OF WOO	DDLAND GROU	Ρ
	W1-W7	NUM	MBERING OF WO	DODLANDS	
	R1a-R	5 NUM	MBERING OF WA	ATERCOURSE	
		WOO	DDL AND		
		SHF	RUB-GRASSLANE	)	
		NA	FURAL STREAM		
		SEM	MI-NATURAL SI	(ream	
		DR/	AINAGE CHANNE	EL	
		LAN	NDSCAPE PLANI	FATION	
		AGF	RICULTURAL LA	AND	
		VIL	LAGE/ORCHARE	)	
		VIL	LAGE AREA/OF	PEN AERA	
NO		PO	١D		
		URE	BAN AERA		
		САМ	Mellia crapne	ELL I ANA	
		DIC	DSPYROS VACCI	INIOIDES	
		GEC	DORUM DENSIF	LORUM	
		GLE	EDITSIA AUSTR	ALIS	
		RH	DOLFIA CHAME	PIONII	
R7-	×	AQI	JILARIA SINEM	NSIS	
	$\times$	NEF	PENTHES MIRAE	BILIS	
	$\rightarrow$	L I T	ITLE EGRET St asian por(	CUPINE	
	$\bigstar$	INS	SECTIVOROUS E	3AT	
		CR`		NACOLUTHON	
		NO	OF INDIVIDU	JALS IN CON	GESTED
	(3)		CALITY OF FLO NSERVATION IN	DRAL SPECIE NTEREST	S OF
	₹   	- PR(	DPOSED SLOPE DPOSED RETAIN	NING STRUCT	URE
	Revision	Date	Descrip	tion	Initial
		Designed	Checked	Drawn	Checked
		JLI	YKL	SZ	JW
	Approved	10719	10/19	10/19	10/19
	Agreemer	nt No.			
	Project		E 68/2017	(CE)	
LLW TEI BITERCUAREE		SITE	FORMAT	ION ANI	o l
	INFR	ASTR	UCTURAL		SFOR
	ROAL	) AND	HONG P	D ROAD	, TUEN
	Ν	IUN - F	FEASIBILI	ITY STU	DY
	Drawing	title			
	E		OGICAL M		ON
	A	IND EF	NHANCEN	IENT PI	
EXAMPLES AND	Drawing	No.		Sco	ıle
				_ 1.	3000 (11)
	1981	72/B&	V/FR/130	2	6000 (AT)
		<±	木工程	拓展署	Ë
	CE		ivil Engine	eering a	nd rtment
				ni Dehq	
All And All All All All All All All All All Al					
				®	1,417-0
	BLAU	/r & VE 博威	- AICH HUNU E 程 顧 問	, _{NUNG} L 有限公	ɪ WI I E U 司



- Keyplan Legend
- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ---- Project Site Boundary
- Zone of Visual Influence





VSR1 Photomontage - With Mitigation at Day 1



VSR1 Photomontage - With Mitigation in Year 10

Relation	Dote	Description		Wild		
04	Designed	Checked	Dram	Checke		
Hild	Gordon Li		Gordon Li			
Date	01/09/2020		28/05/2020			
Agreem	ent No. CE	68/2017 (	CE)			
TH	E DEVEL AD AND MUN - F	OPMENT HONG PC	AT SAN I O ROAD, 1 TY STUD	HING TUEN Y		
	PHOT CAR PA	OMONTA IRK, TUEI	GE VSR1 N TSZ WA	AI.		
Drawing Ho. 19	98172/B&	V/FR/130	3			
土木工程拓展暑 CEDD Civil Engineering and Development Department						
C	De De	velopme	ent Depa	rtme		

C Capyright by Black & Wealch Hong Kong Limited

CAD Filename = \$FILE\$



- Keyplan Legend
- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ----- Project Site Boundary
- Zone of Visual Influence



VSR2 Public Open Area between Tuen Mun San Tsuen and Botania Villa



VSR2 Photomontage - No Mitigation at Day 1



VSR2 Photomontage - With Mitigation at Day 1



VSR2 Photomontage - With Mitigation in Year 10

Relation	Dobs	Description		trilid.
04	Deligned	Checked	Dan	Checker
Hild	Gordon Li		Gordon Li	
Dale	01/09/2020		28/05/2020	
Agreen	tent No.			
	CE	68/2017	(CE)	
Project 10				
	SITE F	ORMAT	ION AND	
IN	FRASTRU	ICTURAL	WORKS	FOR
TH	E DEVELO	OPMENT	TAT SAN F	IING
RC			D ROAD, T	UEN
	WON - FI	ENGIDILI	11 31001	
Douby II	le .			
	PHOTO		GE VSR2	
	FROM P	UBLIC O	PEN ARE/	A
	BETWEE	EN TUEN	I MUN SAN	1
	TSUEN A	ND BOT	ANIA VILL	A
Drawing No.			Socie	
1	98172/B&\	V/FR/130	04	
-			-	
		_ <b>_</b> _ z		
<	<u> </u>	자 ㅗ 찬	至阳展	
C	EDD Ci	vil Engli	neering a	nd
	🥪 De	velopm	ient Depa	rtme
		7	1	
		1. S.	<b>7</b> 8	
D	10K A 1/5	R.		

C Copyright by Black & Wealth Hong Kong Limited



- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ---- Project Site Boundary
- Zone of Visual Influence





VSR3 Photomontage – With Mitigation at Day 1



VSR3 Photomontage - With Mitigation in Year 10

Relation	Date	Description		Intic			
04	Designed	Owded	Drawn	Checker			
Hild	Gordon Li		Gordon L	i			
Date	01/09/2020		28/05/202	0			
Agreem	ent No. CE	68/2017	(CE)				
Project 104	,						
TH RO	E DEVEL AD AND MUN - F	OPMENT HONG PO EASIBILI	AT SAN D ROAD, TY STUE	I HING TUEN DY			
Drawing till							
	PHOT FROM P ł	OMONTA UI LIN GA HANG TSI	GE VSR ARDEN, 3 JEN	3 SIU			
Drawing No.			3	de			
19	98172/B8	V/FR/130	5				
上木工程拓展書 Civil Engineering and Development Department							
-							

BLACK & VEATCH HONG KONG LIMITED 博賞工程展開有限公司

C Capyright by Black & Wealch Hong Kong Limited





- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ---- Project Site Boundary
- Zone of Visual Influence



VSR4 Photomontage - With Mitigation in Year 10

15

BLACK & VEATCH HONG KONG LIMITED 博成工程展開有限公司



- Keyplan Legend
- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ----- Project Site Boundary
- Zone of Visual Influence



VSR5 Existing View from Kei Lun Wai Children's Playground



VSR5 Photomontage - No Mitigation at Day 1



VSR5 Photomontage - With Mitigation at Day 1



VSR5 Photomontage - With Mitigation in Year 10



C Copyright by Black & Vealch Hong Kong Limited



VSR6 Photomontage - With Mitigation in Year 10





- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ---- Project Site Boundary
- Zone of Visual Influence





- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- Project Site Boundary
- Zone of Visual Influence



VSR8 Existing View from Tsing Chung Koon Road



VSR8 Photomontage - No Mitigation at Day 1



VSR8 Photomontage - With Mitigation at Day 1



VSR8 Photomontage - With Mitigation in Year 10

Relation	1049	Decision		Mild			
04	Designed	Credied	Dan	Guided			
hilid	Gordon Li		Gordon Li				
Date	01/09/2020	Y	28/05/2020				
Agreem	ent No. CE	68/2017 (	CE)				
Rojet 66 INI TH RO	SITE FORMATION AND INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY						
Draing 05	PHOTOMONTAGE VSR8 FROM TSING CHUNG KOON ROAD						
Drailing Ho.	98172/B&\	//FR/1310	)				
<u>c</u>	土木工程拓展暑 CEDD Civil Engineering and Development Department						
BL	ACK & VE		GKDNGL 有限公	IMITED			

C Copyright by Back & Wealch Hong Kong Limited





- Keyplan Legend
- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ----- Project Site Boundary
- Zone of Visual Influence





VSR9 Photomontage - No Mitigation at Day 1





- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ----- Project Site Boundary

Keyplan Legend

Zone of Visual Influence



VSR9 Photomontage - With Mitigation at Day 1



VSR9 Photomontage - With Mitigation in Year 10





- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ----- Project Site Boundary
- Zone of Visual Influence



VSR10 Photomontage - No Mitigation at Day 1



VSR10 Photomontage - With Mitigation at Day 1



VSR10 Photomontage – With Mitigation in Year 10





KEYPLAN

- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ---- Project Site Boundary
- Zone of Visual Influence



VSR11 Existing View from between Castle Peak Trail and Por Lo Shan



VSR11 Photomontage – No Mitigation at Day 1



VSR11 Photomontage - With Mitigation at Day 1



VSR11 Photomontage - With Mitigation in Year 10



KEYPLAN

- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ---- Project Site Boundary
- Zone of Visual Influence



VSR12 Existing View From Chung Shan Hillside with Graves



VSR12 Photomontage - No Mitigation at Day 1



Credied Drawn Owied

Gordon Li

C Capyright by Black & Wealch Hong Kong Limited



Keyplan Legend

- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ---- Project Site Boundary
- Zone of Visual Influence



VSR12 Photomontage - With Mitigation at Day 1



Owied

R, BLACK & VEATCH HONG KONG LIMITED 博成工程展開有限公司

VSR12 Photomontage - With Mitigation in Year 10





- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ----- Project Site Boundary
- Zone of Visual Influence



VSR13 Existing View From West Railway heading South-west



VSR13 Photomontage - No Mitigation at Day 1



VSR13 Photomontage - With Mitigation at Day 1



VSR13 Photomontage - With Mitigation in Year 10

Rabian 04	Date Designed	Decription Oracled	Dan	initid Chadad
Roldon 04 Hild	Date Datigned Gardon Li	Decription Oracled	Dam Gordon Li	initid Checked
Realizo 04 Initia Date Approved	Date Designed Gordon Li 01/09/2020	Oscopilion Greated	<b>Dan</b> Gordon Li 28/05/2020	jalid Gwdad
Relation 04 billiol Dolle Approved Agreemen	Date Datigned Gordon Li 01/09/2020 ent No. CE	0eetplien 9eeted	Dan Gozdon El 28/05/2020 CE)	initia Chuchead
Relation 0.4 Hillsi Date Agrowed Agreemin Roject title THI ROJ	Date Designat Goridon Li 01/09/2020 ent No. CE SITE F RASTRU E DEVEL AD AND I MUN - F	Condition Condition 68/2017 ( FORMATI JCTURAL OPMENT HONG PC EASIBILI	CE) ON AND WORKS AT SAN I D ROAD, TY STUD	FOR HING FUEN
Roldon 0.4 billio Dollin Aground Rojut Illio THI ROJ	Data Data Gordon Li 01/09/2020 01/09/2020 CE SITE F RASTRL E DEVEL AD AND MUN - F	Decision Ordent 68/2017 ( FORMATI JCTURAL OPMENT HONG PC EASIBILI	Dan Gesden Ef 38/05/2020 (CE) ON AND WORKS AT SAN I D ROAD, T Y STUD	FOR HING Y
Relation 0.4 India Agreem Regist Bio Regist Bio THI RO, Douby We	Date Date Gordon Li Gordon Li Giles/2020 ant No. CE SITE F FRASTRL E DEVEL AD AND I MUN - F HOTC FROM HEADI	68/2017 ( GRMATI JCTURAL OPMENT HONG PC EASIBILI DMONTAC 1 WEST R NG SOUT	Dem Gerden LE 28/09/2020 (CE) ON AND WORKS AT SAN I D ROAD, 1 TY STUD' GE VSR13 (AILWAY TH-WEST	FOR HING FUEN Y
Reden 04 Hills Date Agreem INF THI RO, Douby We Douby He 15	be     Indyat     Gradam EI     indyat     Gradam EI     indyat     Gradam EI     indyat     i	Certific General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General General Gener	CE) ON AND WORKS AT SAN I PRODUCT SE VSR13 AILWAY FH-WEST	FOR HING Y
Relation 0.4 Helidi Date Agreement Project Sile Cooling His 15 Dooling His 15	Be      Daged     Gradian EI     Digged     Gradian EI     Creater     Sitter     Creater     Sitter     Sitter     Photoco     Photoco     FROM     HEADI     HEADI     Creater	Decide Decide Decide Decide Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Constant Cons	Dem Gosten EF 29/09/2020 CCE) ON AND WORKS AT SAN I P ROAD, 1 TY STUD SE VSR13 AILWAY FH-WEST	FOR HING UUEN Y

C Copyright by Back & Watch Hong Kong Limited



Keyplan Legend

- Visually Sensitive Receivers (VSRs)
- - Proposed Development Area Boundary
- ----- Project Site Boundary
- Zone of Visual Influence



VSR14 Existing View From Hong Po Road Roundabout



VSR14 Photomontage - No Mitigation at Day 1



VSR14 Photomontage - With Mitigation at Day 1



VSR14 Photomontage - With Mitigation in Year 10

Relation	Dates	Description	Daw	bilid (hulud
HN	Gordon Li	- Villes	Gordon Li	Cincles
Dole	01/09/2020		28/05/2020	
Agreeme	ent No.			
Rojet the INF THE RO/	SITE F RASTRU E DEVEL AD AND I MUN - F	ORMATIO ICTURAL OPMENT HONG PC EASIBILIT	ON AND WORKS AT SAN I ROAD, 1 IY STUD'	FOR HING FUEN Y
Drailing Sile	PHOTC FROM R	MONTAG HONG P OUNDABG	GE VSR14 O ROAD OUT	
Drawing Hou 19	8172/B&	V/FR/1316	5	2
<u>C</u>	DD Civ	木工程 /il Engine /velopme	拓展 # eering a ent Depa	nd Intment
BLA	ICK & VE		GKDNGL 有限公	IMITED

C Copyright by Black & Vealch Hong Kong Limited





MM4 & MM5: Roadside Greening & Screen Planting

> MM7: Noise Barrier Treatment

> > MM7: Noise Barrier Treatment

MM7: Noise Barrier Treatment

Opening of 3.5m (H) x 4m (W) at the barrier for crossing. A fully enclosed pedestrian walkway (=15m long) will be provided behind the crossing

and the set which is a few which

Station of the second

RANGING AND A DIS OFFICE

Laying of fresh water main and salt water main via Ming Kum Road would be reinstated

						-		
Laying of salt water main and rising mains and would be reinstated	Laying of t mains along Po Road an Tsing Lun F convey sew the SHR SP the downst and would	he rising g Hong d Road to rage from S to ream TM54 S be reinstate	Landso Nos. of be ref Nos. of be trans	Cape Impact Assessment Trees to tained Trees to moved Trees to splanted	vithin 500m Area - 8,350 - 9,800 ~ 1,300 2	CyddibyBut & Webh H Legend Propose Area Bo Project Interfoc Under T Retainin Slope Retain Noise Water Compa Roadsi Screen Roadsi Screen Project 1 Project Slope Retainin Noise Water Compa Roadsi Screen Project Screen Roadsi Screen Project Screen Roadsi Screen Roadsi Screen Roadsi Screen Roadsi Screen Roadsi Screen Roadsi Screen Roadsi Screen Roadsi Screen Roadsi Screen Roadsi Screen Roadsi Screen Roadsi Screen Roadsi Screen Roadsi Screen Roadsi Screen Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi Roadsi	g Kog Linked d Development sundary Site Boundary ing Housing Sites uen Mun Area 5 g Structure ed Tree of Larg Barrier Treatme Channel Legen insatory Planting - isape Work & Roof for Infras de Greening & Planting & ind Enhanceme ig With ical Enhanceme ig With ical Enhanceme ial School Site ive Open Space	s 4 4 e Size ents d copy ng and structure int wnt
Id be reinstated	「一世」「「	1	Nos. of be newly	Trees to y planted	~ 1,300			
Contra Contra	Overall	Greenery Area SHR Site	Extension	SHR Site	HPR Site			
	Site Area Greenen	a (m ² ) Around	15,000 A	Around 75,000 5,000 (min. 20%	Around 76,000			
	(m ² ) Greener	y Area Provided 3.0	a) 0	of Site Area)	of Site Àrea)			
		Dpen Space		1 m²/person				
						Remarks: T for concept purpose. T shown on f	⁻ he layout is ual layout p he provisior his plan are	s Ian Is
						subject to ristages Note: Date: Dat	eview at late	9r Owled
			- ALANA			Bady The Dedy The Dedy The Dedy The Dedy The Dedy The	RMATION AN TURAL WORK OPMENT AT D AND HONG MUN - FEASI STUDY	D (S FOR SAN PO BILITY
						Outline L Douby Ha 198172/B&V/F	(SHEE (SHEE R/1317b	an [2 OF 3) : 3000 n A2)
	AT NOR		ALL	1		CEDD Civil Deve	工程拓展 Engineering a lopment Depa	F nd artment
	ALL MARKER	日本の日本の	A CAL	4111	A STATISTICS AND	BLACK & VEATO		DETED



	Lands Nos. o	Scape Impact w Assessment	vithin 500m Area	Cryffit by Back & Math Hang Kang United         Legend         Proposed Development Area Boundary         Project Site Boundary         Interfacing Housing Sites Under Tuen Mun Area 54         Retaining Structure         Noise Barrier Treatments         Water Channel Legend copy         Compansatory Planting and New Tree Planting - Planting Area         Landscape Work & Green Roof for Infrastructure         Roadside Greening & Screen Planting	
	Nos. o be re Nos. o be trai	f Trees to emoved f Trees to nsplanted f Trees to	~ 1,300 2	<ul> <li>Woodland Enhancement planting With Ecological Enhancemwnt</li> <li>Proposed PTI Area</li> <li>Potential School Site</li> <li>Tentative Open Space</li> </ul>	
Overall Greenery Area Site Area (m ² ) Greenery Area Required (m ² ) Greenery Area Provided (m ² )	SHR Site Extension Around 15,000 3,000 (min. 20% of Site Area) 3,000	SHR Site           Around 75,000           15,000 (min. 20% of Site Area)           15,000	HPR Site Around 76,000 15,200 (min. 20% of Site Area) 15,200		
				Remarks: The layout is for conceptual layout plan purpose. The provisions shown on this plan are subject to review at later	
				stages       Notify The Description       Date     Description     Infil       C3     Dedy Odd     Down     Oxaded       Paul OHN     Paul OHN     Paul OHN     Oxaded       Date     Paul OHN     Paul OHN     Paul OHN       Date     Paul OHN     Paul OHN     Paul OHN       Date     Paul OHN     Paul OHN     Paul OHN       CE 68/2017 (CE)       Paperal     Paul OHN       CE 68/2017 (CE)       Paperal     CE CE 68/2017 (CE)       Paperal     CE CE 0PMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN - FEASIBILITY STUDY       Destription     Outline Landscape Plan       (SHEET 3 OF 3       Destription     State       State       State       Distription     State       State <th colspan<="" td=""></th>	
				土木工程拓展書 Civil Engineering and Development Department	



	© Coj	pyright	by Bla	ck & Veatc	h Hong Kong	g Limited
	LEGEI	ND				
			PROPO AREA	SED DEVEL BOUNDARY	OPMENT	
	-		POTEN IDENT	TIALLY CO IFIED HOT	NTAMINATEE SPOTS	) AREAS
	1 ac					
	1787					
	***** 4 ~ ~ A					
	1					
	/					
The Contract of the Contract o						
	01-					
	Revision	Design	ned	Checked	Drawn	Checked
	Date	JL I 07/	20	CMW 07/20	SZ 07/20	JW 07/20
	Approve	d				
	lii H					
	Agreeme	nt No.	CE 6	68/2017	7 (CE)	
	Project	title CIT				
		RAST				KS FOR
				ONG P	O ROAI	D, TUEN
	Drawing	title				
	- 					
			NTA	MINAT	ED SIT	ES
est						
	Drawing	No.			Sc	cale
	198 <i>′</i>	172/E	3&V/	FR/140	<b>1</b>	: 2500 (A1) : 5000 (A3)
			土才	トエ 程 I Fogio	拓展 eering	署
			Dev	elopme	ent Depa	artment
	[					
	BLA	CK &	VEAT	CH HON	® G KONG L	_IMITED
	>	博威	工材	程顧問	有限公	司





ot Date = 8/3/20

Possible Activities Centry	© Copyright by Black & Veatch Hong Kong Limited
	LEGEND:
Pag (Figerea) sa Page (Select) Page (Select) Pa	PROPOSED DEVELOPMENT AREA BOUNDARY
O USA DEC DE	POTENTIAL SCHOOL SITE
	PROJECT SITE BOUNDARY
	PROPOSED TRAFFIC IMPROVEMENT MEASURES
	PROPOSED ROADWORKS
	STAGE 1 (2025-2030)
	STAGE 2 (2026-2029)
	STAGE 3 (2025-2031)
	STAGE 4 (2026-2033)
Link and a state of the state o	
43         13         13         13         13         14         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         15         1	
PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PREAM PR	
All	
「 「 「 「 」 」 「 」 」 「 」 」 」 「 」 」 」 」	
UP Nov 2017	
WP Her 2017 11 Tombe Cent - Tea	
P0     P0       (78888)     P0       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)       (24)     (24)	RevisionDateDescriptionInitialDesignedCheckedDrawnChecked
лого на 2 на 1 на 4 на 1 на 1 н на 1 н н на 1 н н н н н н н н н н н н н	Initial JLI YKL SZ JW
世紀2回 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本日本 日本	Date         07/20         07/20         07/20           Approved
Address Ballersberg Ger Perk U.S.	Agreement No.
	CE 68/2017 (CE)
	Project title
	INFRASTRUCTURAL WORKS FOR
	ROAD AND HONG PO ROAD, TUEN
Strate Reference Reference Reference Packing Ger Franceson Packing Franceson Packing Ger Franceson Packing Fra	MUN - FEASIBILITY STUDY
	Drawing title
	STAGES OF DEVELOPMENT
	Drawing No. Scale
	<b>198172/B&amp;V/FR/1501b</b>
	← 十 木 T 程 拓 屈 墨
	CEDD Civil Engineering and
	Development Department
	BLACK & VEATCH HONG KONG LIMITED
	博威工程顧問有限公司

# **APPENDIX** A

# SUMMARY OF PREVAILING WINDS, MAJOR BREEZWAYS AND LAYOUT PLANS



Appendix A-1 - Summary of Prevailing Wind Directions in the vicinity of Study Area



Appendix A-2 – Building Morphology Effect on Wind Environment of the Site



Appendix A-3 - Layout Plan for Hong Po Road Site



# Appendix A-4 - Layout Plan for San Hing Road Site Extension



Appendix A-5 - Layout Plan for San Hing Road Site

# **APPENDIX B**

# ACCESS ARRANGEMENT AND PUBLIC TRANSPORT FACILITIES


Ā



	$\langle$	IK		.//				
					C TO C			+ 10.0
	Å			TS [ ]	TS Fry J	$\rightarrow$		停車場 Car Park
						$\mathcal{A}$	$\times$	
	TS			•	KVIII.			
		3121	TS A HILL HILL	/			10.7 +	
	$\sum$	1		4	TS			+ [] 隱廬
	3,15	1						ł /
		×E	7.0 TS /					
	$\searrow$						L L	
	X	W-I						
				TS	$-\gamma$ $\langle j$			
		$\times$	K AN I THE T	- <u>TA</u> '7/	$\gamma = \gamma =$	2		
	$\searrow$		1 1 1		$\left( \begin{array}{c} \\ \\ \\ \\ \end{array} \right) $	<u></u>	+ ^{9.0}	
	$\langle \rangle$	<u>)</u>	LIS LIS		7.7			
	$\langle \rangle$					1		
	$\sum$		ISK ALL A 323 TS		TS			/ E
	X > 1	' AF				474		
	s S	$\sum   _{31}$	15 F 6.2		TS TS	K. Z		
	$\langle \gamma \rangle$			, Lt	6.6 + Le	$\lambda$		TS G
$\mathcal{Y} = (\mathcal{Y}, \mathcal{Y})$					E E	TS >		
	$\langle \rangle$			3 326		4		
	$\overline{}$			-1-1-1-				
			218 TS THE KING IS A		E E			
		1 A						
	,			V BY	$\searrow$			
		/  L.		_})) /!			۲ ل	
,5 ' K	$\sim$	1					Γ — Ι	
	$\square$	$\nearrow$					TS	
	X	$\mathcal{A}$		$\chi$				
	Ľ,	X						
	1	S		$\checkmark$				
	`		CT FD					
	``							and I
	Л,		Romp	1				
		· Y		$(\mathcal{A})$	- 五柳路			
		'		$\langle I \langle I \rangle $		- 🔶 -	+ - NG LAU ROAD	
	/ '				Cycling Track 8.	1 · · · ·		
8.2	,	)	7.9	TSING		+	# 58 55	
	· (	∧			LON ROAD	8.0		
	+8.7			· •	0	+		7.5+
	$\frown$	<i>"</i>	+9.3				7.6	
		н	9.7					TH 1 1 1 1 1 1
	~/~					ycling Track	<u>單庫徑</u> L 7.6 +	<i>₩</i> - <i>1</i> + 7.2
	* / · · ·	`~~_,						HT- H, T-
E MINOR AMENDMENT	LTH	09JUL21	Project Title		Drawing Title			
D MINOR AMENDMENT	LTH	10JAN20	AGREEMENT NO. CE 68/2017 (CE) SITE FORM	ATION AND	PROPOS	ED VEHIC	CULAR ACCESS	TO THE SITES (3
C MINOR AMENDMENT	LTH	230CT19	AT SAN HING ROAD AND HONG PO ROAD,	TUEN MUN				
B MINOR AMENDMENT	LTH Checked	26JUN19			Designed WTH Checked	LTH Scale 1	:1000(A3)	
							0012021	



## **APPENDIX C**

## LAYOUT OF PROPOSED JUNCTION IMPROVEMENT



PHK50323410/TIA3/F61-F.DGN/DAL/02JUL21

)		//	$\mathbb{X}$	TS,		M $M$ $323$ $TS$			× *	TS	
$\mathbb{N}$							T S				Ž ľ
	e L Solo Contraction	//		TS	TS ==	E	6.2		T	s A	
~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~							+		6.6 + Le A	~/ >>	
~				\sim			TS X	+		TS >	
								11-326			
	7.4	, <i>R</i> J	• L							-74	
	Ro Hole R - C	() (TS Tuen Liogo		TS TS		E		JUNI JUNI
			Stop/GW AF		Kowloop	K-11			e •	r —	ESS
		SP /		-M	WZTMZLamtei (DS4)	-20		322		1	
		Ź			Son Hing Tsue		, <u> </u>			1	
\geq		EG .	' }	Hong Po Road 康寶路)) L	- 4 1	T _{IS}	$\geq -$	i	Shenzhen Bay
		R	' <i>\</i>	NTW/TM/HonPo-01				San Hing 新慶村	Tsuen		「深圳灣」 Yuen Long TS
		Ä	\`X	E CONTRACTOR	12 - 3		/	Tsing Chu	en Wai	いた Kowloan 九龍	元朗 Fu Tei
		K'		5.4				- NTW/TM/LamT	eiV-23	/TM/Lamteil-2	4 虎地
\backslash			Contraction of the second seco					(DS6)			Tuen Mun Town 屯門市
	8.9 + J / J / J)				N-TSB3			A i		//////Kowloon
\backslash			1	TL/Sharp Devi			-1-10-1-				/ NTW/TM/LamTeiI-07B
		H H	'	· /	A60		LI]		The RELEASE		
7			1		GW O VIIII	A A A A A A A A A A A A A A A A A A A	X	8240	I.M.		Rd Ah
2			TL	and Dev							6.0 NG LAU ROAD
患			Ishorp D	, TL/Sh	E-					Tr. Salation	
	Podium L & B A		8.2	TUSTOP DEV	The Level Cit		TS	SING THE	FP	8.3	
5		1	ţ			And the second s		LUN ROAD			
>		A			2.0	AL AND			R/a Ah	-	
				+ 8,57 L	T- qota			300	· · · · · · · · · · · · · · · · · · ·	1	
		-			TODUTI DUS		Cyt Dism Ped	Rd A			ALCO LIN CONTRACTOR CONTRACT
					TO TOTAL A	L. ISTONTL	Cyt Dism Ped				
			Kei Lun Wgi 麒麟圍				Cyr Diam Peg				
	9.4 Landon Landon		Kei Lun Wai 麒麟圍 Tsz Tin Tsu 紫田村	CARRIAGEWAY TO INCREASE CAPACITY	TO BE WID JUNCTION	LSTOOTL L ENED	Cyst Dillion Peg				
			Kei Lun Wai 麒麟圍 Siz Tin Tau 紫田村 Ziu Hong Cou	CARRIAGEWAY TO INCREASE CAPACITY		ENED	Crit Diam Pag				
			Kei Lun Wat 離離 圍 了sz Tin Ta 紫田村 7 Siu Hong Cou 光康苑	CARRIAGEWAY TO INCREASE CAPACITY			ELEVATED R		「 「		
			Kei Lun Wai 離幽 Isz Tin T 紫田村 Siu Hong Cou 米康苑	CARRIAGEWAY TO INCREASE CAPACITY / 籃球場 / Basketball	TO BE WID JUNCTION N-T583		ELEVATED RO				
			Kei Lun Wat 麟麟園 Sz Tin I 紫田村 7 Siu Hong Cou 米康苑	CARRIAGEWAY TO INCREASE CAPACITY - / / Strikk un-040 / Basketball / Court /	TO BE VID Ne TO BE VID UNCTION N-TS的3 / / Basketball / Court /	NED I I I I I I I I I I I I I I I I I I I	ELEVATED RC			Construction	And Male and A
			Kei Lun Wai 蟹麟圖 Tsz Tin TA 紫田村 7 Siu Hong Cou 光慶范	CARRIAGEWAY TO INCREASE CAPÁCITY I Basketball Court I I	TO BE WID Ne TO BE WID JUNCTION N-TSB3 / / / 籃球場 / Basketball / / Court /	ENED	ELEVATED RC				ала на Ц
			Kei Lun Wal 戲麟圖 Sz Tin Ifu 紫田村 7 Siu Hong Cou 米康苑	CARRIAGEWAY TO INCREASE CAPACITY INCREASE CAPACITY INCREASE CAPACITY INCREASE CAPACITY INCREASE CAPACITY INCREASE CAPACITY INCREASE	TO BE VID Ne TO BE VID UNCTION N-TISb3 /// // 篮球場 // Basketball / / Court /	STOUTL L ENED L L L L L L L L L L L L L L L L L L L	ELEVATED RC				лин на
			Kei Lun Wai 離幽 Isz Iin K S Siu Hong Cou 光康范	CARRIAGEWAY TO INCREASE CAPACITY INCREASE CAPACITY / 監球場 / Basketball / Court / J	TO BE VID Nd TO BE VID JUNCTION N-TS83 / / / Easketball / Court /	NED I I I I I I I I I I I I I I I I I I I	ELEVATED RC	100 A 100 A 10	「 「 「 」 」 」 」 」 」 」 」 」 」 」 」 」		
	0.4. 0.4.		Kei Lun Wals 離幽 Star Tin Tau 紫田村 Stu Hong Cou 米康苑 TM/TsiL	CARRIAGEWAY TO INCREASE CAPACITY / Un-04T / Basketball / Court / / Basketball	TO BE WID JUNCTION N-TSB3 / / / Statkstball / Basketball / Court / /	STORTL L NED L L L L L L L L L L L L L L L L L L L	ELEVATED RC	и	「 「 「 」 」 」 」 」 」 」 」 」 」 」 」 」	The second seco	ПАРАМЬ
			Kei Lun Wol 解離國 Tsz Tin Ta Su Hong Cou 光康范 TM/TsiL	CARRIAGEWAY TO INCREASE CAPACITY I Gasketball / Gasketball / Court / Gasketball	TO BE VID Ne TO BE VID UNCTION N-TS83 / / / Basketball / / Court /	NTW/TM/LamTeil-17A	ELEVATED RC				альны Ц
	0.4. U. A. U. A.		Kei Lun Wai 離 鮮 Tsz Tin Tau 紫田村 Tsu Hong Cou 米康苑 TM/TsiL	CARRIAGEWAY TO INCREASE CAPACITY I Basketball Court I Basketball Court I Gwah Group of Hospitals Tze Tin Memorial College	TO BE WID Ne TO BE WID JUNCTION N-TSB3 / / / SE球場 / / Basketball / / Court /	STOUTIL NED NED NED NED NED NED NED NED					
			Kei Lun Wol 解離國 Siz Tin Ta Siu Hong Cou 光康范 TM/TSIL	CARRIAGEWAY TO INCREASE CAPACITY , Setting ,	TO BE VID Ne TO BE VID UUNCTION Y-TS83 / / / Basketball / Court / /	NTW/TM/LamTeiI-17A				Image: Source of the second secon	альны Ц
	0.4. 0.4.		Kei Lun Wai 離 點 Tsz Tin Tau Siu Hong Cou 米康苑 TM/TsiL	CARRIAGEWAY TO INCREASE CAPACITY I ERRIAGEWAY TO INCREASE CAPACITY I ERRIAGEWAY TO INCREASE CAPACITY I ERRIAGEWAY I ERRIAGEWAY I ERRIAGEWAY	TO BE WID Ne TO BE WID JUNCTION N-TSB3 / / / SE球場 / / / Sex場 / / / Sex場 / / / Sex場 / / / Sex場 / / / Sex場 / / / Sex / / / Sex / / / Sex /	STOUTL NED NED NED NED NED NED NED NED	CFT Distor Peg FD ELEVATED RC 「 」 」 「 」 」 」 「 」 」 」 」 」 」 」 」 」 」 」 」 」	Хтор 1 1 1 1 1 1 1 1 1 1 1 1 1	「 「 「 」 」 」 」 」 」 」 」 」 」 」 」 」		ПАРАНЦ
L. L.			Kei Lun Wat 戲麟圖 Sz Tin Ku Sz Tin Ku Ku Hong Cou 光度苑	CARRIAGEWAY TO INCREASE CAPACITY - Int / Basketball / Basketball / Basketball / Court / / Stationary / Court / / Court / / Stationary / Court / / Court / Court / / Court / Court / Court / Court / / Court / Cour	TO BE VID Ne TO BE VID UUNCTION Y-TS\$3 / / / Basketball / Court / / / J / Basketball / Court	NTW/TM/LamTeil-17A	Cri Diam Peg Cri Diam Peg Fo L L	L Stand L Stand L Stand			С. С
			Kei Lun Wai 離麟 國 Siz Tin Tau 紫田村 Ziu Hong Cou 米康范 TM/TsiL	CARRIAGEWAY TO INCREASE CAPACITY TO INCREASE CAPACITY Basketball / Basketball / Basketball / Court / Basketball / Court / J Basketball / Court / J Basketball	TO BE VID Ne TO BE VID JUNCTION N-TSB3 / / / 籃球場 / Court / /	STOUTL ENED	Cri Dism Peg FD L ELEVATED RC L L L L L L L L L L L L L L L L L L L	и Клач 1 1 1 1 1 1 1 1 1 1 1 1 1	「 「 「 」 」 」 」 」 」 」 」 」 」 」 」 」		
	9.4 9.4 9.4 9.4 9.4 9.4 9.4 9.4 9.4 9.4		Kei Lun Wat 戲劇 圖 Tsz Tin If 紫田村 7 Siu Hong Cou 光康范	CARRIAGEWAY TO INCREASE CAPACITY I	TO BE VID Ne TO BE VID UNCTION N-TS83 / / Basketball / Basketball / Court / / / Basketball / Court / / / Basketball / Court	NED I I I I I I I I I I I I I I I I I I I	Cri Diam Peg Cri Diam Peg Fo L L	L Stand L L Stand L L L L L L L L L L L L L L L L L L L	「		С. С
			Kei Lun Wai 離幽 Stsz Tin Téu 紫田村 Zstu Hong Cou 米康范 TM/TsiL	CARRIAGEWAY TO INCREASE CAPACITY INCREASE CAPACITY Basketball Court Basketball Court	TO BE VID Ne TO BE VID JUNCTION N-TS83 / / / Basketball / Court / / / / / / / / / / / / / / / / / / /	NTW/TM/LamTeiI-17A	Cri Dism Peg Cri Dism Peg ELEVATED RC L L L L L L L L L L L L L	240 1 1 1 1 1 1 1 1 1 1 1 1 1	「		Are Ma ^L a a a b a a a a a a a a <
L. L. L	- UPDATE BASE MAP		Kei Lun Wal 蘇麒 「Sz Tin Ifu 紫田村 「Siu Hong Cou 大康苑 「TM/TSIL	CARRIAGEWAY TO INCREASE CAPACITY - 	TO BE VID Ne TO BE VID UNCTION N-TSB3 / / Basketball / Court / / Basketball / / Basketball / / / / / / / / / / / / /	NED NED NED NED NED NED NED NED	Cri Diam Pag Cri Diam Pag ELEVATED RC I I I I I I I ELEVATED RC I	PROPOSED	「 「 「 」 」 」 」 」 」 」 」 」 」 」 」 」	The intervence of the interve	VEMENT SCHEME
 L. L. L. B.	9.4 9.4 9.4 9.4 9.6 9.6 9.6 9.6 9.6 9.6 9.6 9.6 9.6 9.6		Kei Lun Wai 離離 Siz Tin Téu 紫田村 Tsz Tin Téu 紫鹿花 (小田/TsiL 25JUN21 230CT19	CARRIAGEWAY TO INCREASE CAPACITY	TO BE VID UNCTION V-TS83 / / / Basketball / / Court / / Start Court / / Court / Court / / Court / / Court / / Court / Court / / Court / / Court / Court / Court / / Court / Court / Court / Court / / Court /	NED NED NED NED NED NED NED NED	Critican Peg Critican Peg ELEVATED Peg L L L L L L L L L L L L L	PROPOSED TSING	「 「 」 」 」 」 」 」 」 」 」 」 」 」 」	Demonstrate in the intervention of the int	MALE TO THE SCHEME SPOROAD (J4)
- C B A	- UPDATE BASE MAP MINOR AMENDMENT MINOR AMENDMENT		Kei Lun Wal 離幽 圖 Tsz Tin Tau Kei Lun Kal Kei Kei Lun Kal Kei Kei Lun Kal Kei Kei Kei Kei Kei Kei Kei Kei Kei Kei	CARRIAGEWAN CARRIAGEWAN TO INCREASE CAPACITY Basketball Court J Basketball Colege Basketball Colege Colege State Colege J Colege State Colege State Colege State Colege State Colege State Colege State Colege State Colege State Colege State Colege State Colege State Colege State Colege State	TO BE VID Ne TO BE VID UNCTION N-TSB3 / / Basketball / / Basketball / / Court / / / / / / / / / / / / / / / / / / /	NED I I I I I I I I I I I I I I I I I I I	Cri Diam Pag Cri Diam Pag ELEVATED RC Cri Diam Pag ELEVATED RC Cri Diam Pag ELEVATED RC Cri Diam Pag Cri Dia	PROPOSED TSING	「 「 「 」 」 」 」 」 」 」 」 」 」 」 」 」	 International and the second s	Comment Schemes





+ /	CUL CUL	4	•_		\ \	/ `,		/	/		/ /				ÆШ		1
			<u> </u>	°°°))	1	L	5			7 LHBaskett ノ	水场 all Court +16.6	ÍÆ			
								L + NON		77	$\langle \langle$			A Villin	A Lietoff		-
)/	3-5- A Block	J座 ck 7		\land	$\sum $		L				座 🔪				Minny J.		
		57			AL		<u> </u>		\backslash	- BIOC	~L					Ţ	
				\bigcirc	PH	~				nL	7					1	
///		L	Corpor		Н				C.W		/ L	20		15.4		V.	
~			RO				第八座		L	R		K Y					
	× //) (let a let	17.5 +	L				Block 8	л /		°°°	γ	000					
	LI MC TREES M	K				L /	/		L L		2	1	IS S	\$	IT T		
		$\left\{ \right\}$					5		LH		~		Control Martin			<u> </u>	-111111111
1./		} \/	1				51			$ \begin{array}{c} 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 $			S		1		
Š/			• L	第六座 Block 6			en an			15 4	4-TS896	8	+		1-1-1		7 - 7 - 7 - 7
2000							N M			+ L				// ¥			
		$\langle $ (HM	A LE LE	In the second se				1.0/	14.9				
1					Puno L							NUT AL	日本 日本 日本 日本 日本 日本 日本 日本 日本 日本				
/		/						Toxi					н-тзаз ,		1		
								15.4 +			HAA	A Ten	195/EGH, Contraining Place-Parties			20.2	
				7	End			and the second			ŦĦ	Y (12. Sty and the detected to construent Construction of the anti- Construction of the anti- Structure of the structure of the anti- per person of the structure of the structure 単重 注意車 透一种会须 经 1. 国际家庭的重要化。	ester de la constante de La constante de la constante de		· · · ·	/
	25.5	\bigcap		ίζ		T I I I I I I I I I I I I I I I I I I I					HH.		H 4-75895 H	-1.4.1111111111111111111111111111111111			
, '		L			LA DU	and a second sec	The set				H		Τō.				K_
	15.	.1	~~~		H						Ð	THE	ISUN WEN				
<i>'</i> /	CUL	•		Block 5		寶田城			۹. 1		()			TOAD			/ ` _ ` _ ` _ ` _ ` _ ` _ ` _ ` _ ` _ `
		\sim		52		FIN A STAT	TE,		₩ Å 3 8 8 8 8 8 8 8 8 8 8 8 8 8			H				BE TO TO	
					J /				.8		H						
				Hull -	H /		J	12	/1								\mathcal{I},\mathcal{K}
			ζ				平台 odium					Po Tong Ho 寶塘下		12.9 +	€7195.ECH		
		25										Siu Hang 小坑村			13.0		
	, ^{14.7}	へっ 第四回				可場 Tin						NTW/TM/M	inKum-09			_//	
\geq		Bloc	k 4	Z	Shopping	Centre	/					=			\times		
			77	Cw 7	/		< /r	ALI H R R R R R R R R R R R R R R R R R R R					je I		震寰路變電站		
		/				平台 Podium	24.8	Cyt Dia	N			X	<u>A I I I I I I I I I I I I I I I I I I I</u>				
\backslash					(下層寶田街市)				צ א				* * *	LEUN	G TAK	STREET	
			[(Pc	Tin Market ur	nder)	× H	5 蓰	MIN		ק /						
			31.		dium	/								/ //		METHOD	OF CONTRO
L			\mathcal{Y}				Podium L		, I			Ī				1	
1		E.		/			• • • •	┐ │					第一座	` \`	HT29.3		`` ×
	L Playground)'/ ,	\square		L +13.8						+		Tower 1	\Box \backslash \Box	Hr	<i>,</i>	
LE	GEND:		4	$\int_{C} O \left \frac{1}{2} \right $		Г				I		<u>і</u> н		\•	П.		,t P
	Proposed Works									• I	•	Ī			H		
D	UPDATE BASE MAP	LTH	25JUN21	Project Title					Dra	awing Title							
C		LTH	230CT19	AGREEMEN INFRASTF	NO. CE 6 RUCTURAL V	8/2017 (CE) VORKS FOR) SITE FOF R THE DEV	RMATION A	ND T	I	PRO	POSE					
A	MINOR AMENDMENT		13JUN19 15MAY19	AT SAN H	NG ROAD A - FEA	AND HONG AS <mark>I</mark> BILITY S	PO ROAD STUDY	, TUEN MU									(012)
Rev.	Description	Checked	Date						Des	signed WT⊢	Che	cked LTH	^{Scale} 1:1000(A	.3) Date Jul 2	2021 Dr		

PHK50323410/TIA3/F64-D.DGN/DAL/25JUN21





PHK50323410/TIA3/F65-D.DGN/DAL/25JUN21



PHK50323410/TIA3/F66A-C.DGN/FAL/25JUN21

)						
		Ι TS Γ						
								_
31.0 +								
				-1-10,)	7			150
B					W II	P Nov 2020	L	
R							11	
				®i //				22.6
				BLI		5	E	
				×1			H TS	X / X
				-		23.8		JESS L
				,t		+		
		*	Track	`\}			V / IK m	5m 1
		/		11		Ţ	Y/III	1 1 1 1 1 1 1 1 1 1
	/ /			1/1				
		/				[] [z]		Silcon Garden
				7/1		I I I III		9º 10 11
	/					22.9		
				anna 1	H		\sim	
					2		L,	
		1	J But But But	4			/ // 1	s 7
				i. F		F TS R	20.7	
40				TS107/		T I TS K	+ e	
	2¢	5.9 I		TS414		TS /		19.5
	_		diference definition					
+28.7						19.0		
						Fung Sh		9.1 +
						-	Temple /	18.6 +
2:				\$				
								1 1 1 8 80°54
					801S			
	<i>·</i>	- 40						
		+ 40.2		18.9 L		∇ - 7	L	
					802		\sim	KETI B TS
							> '∠	7 16.6
						(寶華花園)	٢	
						第二期		
LEGEND:								
Droppood Modifie	otion M	Vorke					1	A H TS
	auon v	VUIKS				PA CI I	16 Y	1+ E==
		-				- 11ª24 - C & 1		TON THE
				1 ·		I H I I	T I	
D UPDATE BASE MAP	LTH	28JUN21	Project Title [Drawing Title				
C MINOR AMENDMENT	LTH	13JUL20	AGREEMENT NO. CE 68/2017 (CE) SITE FORMATION AND	I	PROPOSED) JUNCTIOI	NIMPRO	VEMENT SCHEME
B MINOR AMENDMENT	LTH	250CT19	INFRASTRUCTURAL WORKS FOR THE DEVELOPMENT AT SAN HING ROAD AND HONG PO ROAD, TUEN MUN		ROAD) L7 / ROAD	L54D (J	15) (OPTION 2)
A MINOR AMENDMENT	LTH	15MAY19	- FEASIBILITY STUDY	Designed	Checked	Scale	Date	Drawing No
Rev. Description	Checked	Date		WTH	LTH	1:1000(A3)	Jul 2021	

PHK50323410/TIA3/F66B-D.DGN/DAL/28JUN21





Provision of GIC Facilities and Open Space in the Lam Tei and Yick Yuen Outline Zoning Plan

		HKPSG	Prov	Surplus/		
Type of Facilities	Hong Kong Planning Standards and Guidelines (HKPSG)	Requirement (based on planned population)	Existing Provision	Planned Provision (including Existing Provision)	Shortfall (against planned provision)	
District Open Space	10 ha per 100,000 persons [#]	8.84 ha	0.71 ha	0.71 ha	-8.13 ha	
Local Open Space	10 ha per 100,000 persons [#]	8.84 ha	2.16 ha	10.30 ha	+1.46 ha	
Secondary School	1 whole-day classroom for 40 persons aged 12- 17	99 classrooms	25 classrooms	55 classrooms	-44 classrooms	
Primary School	1 whole-day classroom for 25.5 persons aged 6- 11	121 classrooms	0 classroom	174 classrooms	+53 classrooms	
Kindergarten/ Nursery	34 classrooms for 1,000 children aged 3 to 6	50 classrooms	8 classrooms	38 classrooms	-12 classrooms	
District Police Station	1 per 200,000 to 500,000 persons	0	0	0	0	
Divisional Police Station	1 per 100,000 to 200,000 persons	0	0	0	0	
Hospital	5.5 beds per 1,000 persons	490 beds	0	0	-490 beds	
Clinic/Health Centre	1 per 100,000 persons	0	0	0	0	
Magistracy (with 8 courtrooms)	1 per 660,000 persons	0	0	0	0	
Child Care Centre	100 aided places per 25,000 persons ^{#@}	353 places	0 place	160 places	-193 places	
Integrated Children and Youth Services Centre	1 for 12,000 persons aged 6-24 [#]	1	0	1	0	
Integrated Family Services Centre	1 for 100,000 to 150,000 persons [#]	0	0	0	0	

		HKPSG	Prov	Surplus/	
Type of Facilities	Hong Kong Planning Standards and Guidelines (HKPSG)	Requirement (based on planned population)	Existing Provision	Planned Provision (including Existing Provision)	Shortfall (against planned provision)
District Elderly Community Centres	One in each new development area with a population of around 170,000 or above [#]	n.a.	0	0	n.a.
Neighbourhood Elderly Centres	One in a cluster of new and redeveloped housing areas with a population of 15,000 to 20,000 persons, including both public and private housing [#]	n.a.	0	2	n.a.
Community Care Services (CCS) Facilities	17.2 subsidised places per 1,000 elderly persons aged 65 or above ^{#*@}	378 places	18 places	48 places	-330 places
Residential Care Homes for the Elderly	21.3 subsidised beds per 1,000 elderly persons aged 65 or above #@	468 beds	63 beds	1618 beds	+1150 beds
Library	1 district library for every 200,000 persons π	0	0	0	0
Sports Centre	1 per 50,000 to 65,000 persons [#]	1	0	0	-1
Sports Ground/ Sport Complex	1 per 200,000 to 250,000 persons [#]	0	0	0	0
Swimming Pool Complex - standard	1 complex per 287,000 persons [#]	0	0	0	0

Note:

The planned resident population in the Lam Tei and Yick Yuen is about 88,400. If including transients, the overall planned population is about 89,100.

The requirements exclude planned population of transients.

^ The provision of hospital beds is to be assessed by the Hospital Authority on a regional basis.

* Consisting of 40% centre-based CCS and 60% home-based CCS.

(a) This is a long-term goal and the actual provision would be subject to the consideration of the Social Welfare Department in the planning and development process as appropriate.

 π Small libraries are counted towards meeting the HKPSG requirement.